UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K/A

(Amendment No. 1)

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) **OF THE SECURITIES EXCHANGE ACT OF 1934**

Date of Report (Date of earliest event reported): March 22, 2024

Summit Midstream Partners, LP

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation)

001-35666 (Commission File Number)

45-5200503 (IRS Employer Identification No.)

910 Louisiana Street, Suite 4200

Houston, TX 77002

(Address of principal executive office) (Zip Code)

(Registrant's telephone number, including area code): (832) 413-4770

Not applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

X Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Securities Act:

| Title of each class | Trading Symbol(s) | Name of each exchange on which registered | |
|---------------------|-------------------|---|--|
| Common Units | SMLP | New York Stock Exchange | |

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter)

Emerging growth company D

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Explanatory Note

Summit Midstream Partners, LP, a Delaware limited partnership ("Summit"), filed a Current Report on Form 8-K, with the Securities and Exchange Commission on March 22, 2024 (the "Original Filing") to report the completion of the sale of Summit Midstream Utica, LLC, a Delaware limited liability company, from Summit Midstream Opco, LP, a Delaware limited partnership and an indirectly owned subsidiary of Summit, to a subsidiary of MPLX LP (NYSE: MPLX), a Delaware limited partnership. This Form 8-K/A is being filed to (i) add inline XBRL tagging to Item 1.01 and (ii) amend Item 9.01 of the Original Filing to provide the required pro forma financial information described under Item 9.01 below. No other changes or amendments are being made to the Original Filing, which is restated herein.

Item 1.01 Entry into a Material Definitive Agreement.

Disposition Agreement

On March 22, 2024, Summit Midstream Partners, LP (NYSE: SMLP), a Delaware limited partnership ("Summit" or the "Partnership") and Summit Midstream Opco, LP, a Delaware limited partnership ("OpCo") and an indirectly owned subsidiary of Summit, completed the sale of Summit Midstream Utica, LLC, a Delaware limited liability company ("Utica"), to a subsidiary of MPLX LP (NYSE: MPLX), a Delaware limited partnership ("MPLX"), for cash consideration of \$625.0 million, subject to customary post-closing adjustments (the "Transaction"), pursuant to a Purchase and Sale Agreement, dated March 22, 2024 (the "Purchase Agreement"), by and among OpCo, as Seller, MPLX, as Buyer, and solely for purposes of Section 12.18 thereto, Summit, as Seller Parent (each as defined therein). Utica is the owner of (i) approximately 36% of the issued and outstanding equity interests in Ohio Gathering Company, L.L.C., a Delaware limited liability company ("OCC" and, together with OGC, "Ohio Gathering") and (iii) midstream assets located in the Utica Shale. Ohio Gathering and, prior to the closing of the Transaction, was OpCo's joint venture partner.

The Partnership intends to use the proceeds from the Transaction to reduce outstanding borrowings under its credit facility, for general partnership purposes or future acquisitions. In addition, a subsidiary of the Partnership has agreed to provide customary transition services to MPLX for an initial period of 90 days, unless terminated or otherwise extended.

The foregoing summary of the Purchase Agreement is not complete and is qualified in its entirety by reference to the full and complete Purchase Agreement, a copy of which is filed as Exhibit 2.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Amendment to ABL Facility

In connection with the closing of the Transaction, Summit Midstream Holdings, LLC, a Delaware limited liability company ("Holdings"), entered into that certain Fourth Amendment to Loan and Security Agreement (the "ABL Amendment") by and among Summit, Holdings, the subsidiaries of Holdings party thereto, as subsidiary guarantors, and Bank of America, N.A., as administrative agent ("Agent"), pursuant to which Holdings amended its revolving credit facility governed by that certain Loan and Security Agreement dated as of November 2, 2021 (as amended, restated, supplemented or otherwise modified prior to the date hereof, the "Credit Agreement" and the Credit Agreement, as amended by the ABL Amendment, the "Amended Credit Agreement") by and among Summit, as parent guarantor, Holdings, as borrower, the subsidiaries of Holdings party thereto, as subsidiary guarantors, the lenders party thereto from time to time and Agent. The ABL Amendment includes amendments to the Credit Agreement that, among other things, (i) permit the Transaction, (ii) amend the change of control provision to permit certain structural changes in connection with a conversion to a C-corporation, (iii) permit certain distributions using the proceeds from the Transaction subject to conditions in the Amended Credit Agreement, and (iv) amend the Interest Coverage Ratio (as defined therein) covenant such that the Interest Coverage Ratio as of the last day of any fiscal quarter must be less than (a) for any fiscal quarter ending on or before December 31, 2024, (x) if no loans under the Credit Agreement are outstanding and unrestricted cash sceeds \$200,000,000, 1.50:1.00 and (y) if any loans under the Credit Agreement are outstanding or unrestricted cash is less than such threshold, 1.75:1.00 and (b) thereafter, 1.90:1.00.

The foregoing descriptions of the ABL Amendment do not purport to be complete and are qualified in their entirety by reference to the ABL Amendment, a copy of which is filed as Exhibit 10.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Item 2.01 Completion of Acquisition or Disposition of Assets.

The information set forth in Item 1.01 of this Current Report on Form 8-K is incorporated herein by reference.

Item 8.01 Other Events.

On March 22, 2024, the Partnership issued a press release announcing the completion of the Transaction and the plan to seek approval from the Partnership's unitholders to convert to a C-corporation at a Special Meeting later this year.

A copy of the press release is filed as Exhibit 99.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Forward Looking Statements

This Current Report on Form 8-K includes certain statements concerning expectations for the future that are forward-looking within the meaning of the federal securities laws. Forward-looking statements include, without limitation, any statement that may project, indicate or imply future results, events, performance or achievements and may contain the words "expect," "intend," "plan," "anticipate," "estimate," "believe," "will be," "will continue," "will likely result," and similar expressions, or future conditional verbs such as "may," "will," "should," "would," and "could," including statements about the divestiture, and the potential C-Corp conversion, uses of proceeds from the divestiture, the benefits of divestiture and/or the C-Corp conversion, and any related opportunities and the plans and objectives of management for future operations. In addition, any statement concerning future financial performance (including future revenues, earnings or growth rates), ongoing business strategies and possible actions taken by the Partnership or its subsidiaries are also forward-looking statements. Forward-looking statements also contain known and unknown risks and uncertainties (many of which are difficult to predict and beyond management's control) that may cause the Partnership's actual results in future periods to differ materially from anticipated or projected results. An extensive list of specific material risks and uncertainties affecting the Partnership is contained in its Annual Report on Form 10-K for the year ended December 31, 2023, which the Partnership filed with the Securities and Exchange Commission (the "SEC") on March 15, 2024 (the "Annual Report"), as amended and updated from time to time. Any forward-looking statements to reflect new information or events.

Additional Information and Where to Find I

This Current Report on Form 8-K relates to the proposed corporate reorganization of the Partnership. This Current Report on Form 8-K may be deemed to be solicitation material in respect of the proposed C-corporation conversion. The proposed conversion is expected to be submitted to the Partnership's unitholders for their consideration. In connection with the proposed conversion, the newly formed corporation is expected to file with the SEC a Form S-4 containing a proxy statement/prospectus (the "Proxy Statement/Prospectus") to be distributed to the Partnership's unitholders in connection with the Partnership's solicitation of proxies for the vote of the Partnership's unitholders in connection with the proposed conversion. The Partnership's unitholders in connection with the proposed conversion. The Partnership's unitholders in connection with the proposed conversion. The Partnership's unitholders in connection may file other relevant documents with the SEC regarding the proposed conversion. The Partnership's unitholders when available. BEFORE MAKING ANY VOTING OR INVESTMENT DECISION WITH RESPECT TO THE PROPOSED CONVERSION, INVESTORS AND UNITHOLDERS AND OTHER INTERESTED PERSONS ARE URGED TO READ THE DEFINITIVE PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED CONVERSION (INCLUDING ANY AMENDMENTS OR SUPPLEMENTS THERETO) AND OTHER RELEVANT MATERIALS CAREFULLY AND IN THEIR ENTIRETY WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT THE PROPOSED CONVERSION.

The Proxy Statement/Prospectus, any amendments or supplements thereto and other relevant materials, and any other documents filed by the Partnership or the newly formed corporation with the SEC, may be obtained once such documents are filed with the SEC free of charge at the SEC's website at *WWW.sec.gov* or by directing a written request to the Partnership at 910 Louisiana Street, Suite 4200, Houston, Texas 77002.

No Offer or Solicitation

This Current Report on Form 8-K does not constitute an offer to sell or the solicitation of an offer to buy any securities, or a solicitation of any vote or approval, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offering of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

Participants in the Solicitation

The Partnership, Summit Midstream GP, LLC, the general partner of the Partnership (the "General Partner"), and certain of the General Partner's executive officers, directors, other members of management and employees may, under the rules of the SEC, be deemed to be "participants" in the solicitation of proxies in connection with the proposed conversion. Information regarding the General Partner's directors and executive officers is available in the Annual Report. To the extent that holdings of the Partnership's securities have changed from the amounts reported in the Annual Report, such changes have been or will be reflected on Statements of Changes in Beneficial Ownership on Form 4 filed with the SEC. These documents may be obtained free of charge from the sources indicated above. Information regarding the participants in the proposed conversion of their direct and indirect interests, by security holdings or otherwise, will be contained in the Form S-4, the Proxy Statement/Prospectus and other relevant materials relating to the proposed conversion to be filed with the SEC when they become available. Unitholders and other investors should read the Proxy Statement/Prospectus carefully when it becomes available before making any voting or investment decisions.

Item 9.01 Financial Statements and Exhibits.

(b) Pro Forma Financial Information

In accordance with Item 9.01(b), the following unaudited pro forma condensed consolidated financial information giving effect to the Transaction is filed hereto as Exhibit 99.2:

- the unaudited pro forma condensed consolidated balance sheet of the Partnership as of December 31, 2023, giving effect to the Transaction as if it had occurred on December 31, 2023; and
- the unaudited pro forma condensed consolidated statement of operations of the Partnership for the year ended December 31, 2023, giving effect to the Transaction as if it had occurred on January 1, 2023.

(d) Exhibits

| Exhibit Number | Description |
|----------------|--|
| 2.1* | Purchase and Sale Agreement, dated as of March 22, 2024, by and among Summit Midstream Opco, LP, as Seller, MPLX LP, as Buyer, and, solely for purposes of Section 12.18 thereto, Summit Midstream Partners, LP, as Seller Parent. |
| 10.1 | Fourth Amendment to Loan and Security Agreement, dated as of March 22, 2024, by and among Summit Midstream Partners, LP, Summit Midstream Holdings, LLC, the subsidiaries of Holdings party thereto, the lenders party thereto and Bank of America, N.A., as agent for such lenders. |
| 99.1 | Press Release, dated as of March 22, 2024. |
| 99.2 | Unaudited pro forma condensed consolidated balance sheet of the Partnership as of December 31, 2023, giving effect to the Transaction as if it had occurred on December 31, 2023, and the unaudited pro forma condensed consolidated statement of operations of the Partnership for the year ended December 31, 2023, giving effect to the Transaction as if it had occurred on January 1, 2023. |
| 104 | Cover Page Interactive Data File – the cover page XBRL tags are embedded within the Inline XBRL document |

* Certain of the schedules and exhibits to the agreement have been omitted pursuant to Item 601(a)(5) of Regulation S-K. A copy of any omitted schedule or exhibit will be furnished to the SEC upon request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Summit Midstream Partners, LP (Registrant)

By: Summit Midstream GP, LLC (its general partner)

Dated: March 28, 2024

/s/ William J. Mault

William J. Mault, Executive Vice President and Chief Financial Officer

PURCHASE AND SALE AGREEMENT

BY AND AMONG

SUMMIT MIDSTREAM OPCO, LP,

as Seller,

MARKWEST UTICA OPERATING COMPANY, L.L.C.,

as Buyer,

AND SOLELY FOR PURPOSES OF SECTION 12.18 HEREOF,

SUMMIT MIDSTREAM PARTNERS, LP,

as Seller Parent

MARCH 22, 2024

TABLE OF CONTENTS

| ARTICL | ΕI | DEFINITIONS AND INTERPRETATIONS | 1 |
|--|---|--|----------------------------------|
| 1.1. 1.2. | | ons etations | |
| ARTICL | EII | PURCHASE AND SALE OF THE ACQUIRED INTERESTS; CLOSING | 14 |
| 2.1. 2.2. 2.3. 2.4. 2.5. 2.6. | Purchas Purchas Intentio Withho | se and Sale of the Acquired Interests se Price mally Deleted Iding se Exclusion of Operations Services from the Purchase and Sale | 14 14 17 17 |
| ARTICL | EIII | REPRESENTATIONS AND WARRANTIES OF SELLER | 18 |
| 3.1. 3.2. 3.3. 3.4. 3.5. 3.6. 3.7. 3.8. 3.9. | Authori Non-Co Govern Legal P Owners Brokers Bankru | zation; Qualification ity; Enforceability ontravention mental Approvals roceedings chip of Acquired Interests of Fees ptcy er Representations and Warranties | 18 19 19 20 20 20 |
| ARTICL | EIV | REPRESENTATIONS AND WARRANTIES REGARDING THE ACQUIRED COMPANY | 20 |
| 4.1. 4.2. 4.3. 4.4. 4.5. 4.6. | Non-Co Govern Capitali Compli Real Pr | zation; Qualification ontravention mental Approvals ization; Subsidiaries ance with Law operty | 21 21 21 22 22 |
| 4.7. 4.8. 4.9. | Custom | Properties and Assets; Condition of Assets ers and Suppliers al Statements | 25 |
| 4.10. 4.11. | Debt Absenc | e of Certain Changes | 26 26 |
| 4.12. 4.13. 4.14. 4.15. 4.16. | Materia Legal P Permits | nmental Matters I Contracts roceedings | 27 29 30 |
| 4.17. 4.18. 4.19. | Employ Intellec | vee Benefits; Employment and Labor Matters tual Property; IT ce | 31 32 |

| 4.20. | | |
|----------------|---|----|
| 4.21. 4.22. | | |
| 4.22. | · · · · · · · · · · · · · · · · · · · | |
| 4.23. | | |
| | | |
| ARTICL | | |
| | AND OCC | |
| 5.1. | Organization; Qualification | |
| 5.2. | Non-Contravention; Approvals | |
| 5.3. | Capitalization | |
| 5.4. | Absence of Changes | |
| 5.5. | Legal Proceedings | |
| 5.6. | Taxes | |
| ARTICL | LE VI REPRESENTATIONS AND WARRANTIES OF BUYER | 36 |
| 6.1. | Organization | |
| 6.2. | Authority; Enforceability | |
| 6.3. | Non-Contravention | |
| 6.4. | Governmental Approvals | 37 |
| 6.5. | HSR Act Approval | 37 |
| 6.6. | Legal Proceedings | |
| 6.7. | Matters Relating to Acquisition of the Acquired Interests | |
| 6.8. | Bankruptcy | |
| 6.9. | Financing | |
| 6.10. | | |
| 6.11. | s 옷과 방법방법 🗸 다양 것 - 가방 전화 방법, 동일 전 것 : : : : : : : : : : : : : : : : : : | |
| 6.12. 6.13. | / | |
| | | |
| ARTICL | LE VII COVENANTS OF THE PARTIES | 39 |
| 7.1. | Intentionally Deleted | |
| 7.2. | Intentionally Deleted | |
| 7.3. | Consents and Approvals | 39 |
| 7.4. | Continuing Indemnification | |
| 7.5. | Retention of Books and Records | |
| 7.6. | Expenses | |
| 7.7. | Further Assurances | |
| 7.8. | Public Statements | |
| 7.9. 7.10. | Transfer Taxes Tax Matters | |
| 7.10. | R&W Policy | |
| 7.11. | 같은 동 1987 · · · · · · · · · · · · · · · · · · · | |
| 7.12. | | |
| 7.14. | | |
| 7.15. | | |
| 7.16. | | |
| | | |

| 7.17. 7.18. 7.19. 7.20. 7.21. | Insuran Intentio Indeper | ce Matters nally Deleted ident Investigation Releases | .47 .47 .47 |
|---|---|--|--------------------------|
| ARTICL | E VIII | INTENTIONALLY DELETED | 49 |
| ARTICL | EIX | CLOSING | .49 |
| 9.1. 9.2. | | nd Place of Closing ies and Actions at Closing | |
| ARTICL | ΕX | INTENTIONALLY DELETED | 51 |
| ARTICL | EXI | INDEMNIFICATION | 51 |
| 11.1. 11.2. 11.3. 11.4. 11.5. 11.6. 11.7. | Obligat Obligat Limitat Indemn Damage | I ions of Seller ions of Buyer ions on Liability ification Procedure es Waiver ve Remedy | .51 .52 .53 .54 |
| ARTICL | EXII | MISCELLANEOUS | 55 |
| 12.1. 12.2. 12.3. 12.4. 12.5. | Consen Waiver Amend | ing Law t to Jurisdiction of Jury Trial ment and Modification of Compliance; Consents | .55 .56 .56 |
| 12.6. 12.7. 12.8. | Notices Assignr | nent arty Beneficiaries | .56 .57 |
| | Severab | Agreement ility entation by Counsel | .58 |
| 12.12. 12.13. | Seller D Facsimi | Disclosure Schedule les; Counterparts | .58 .59 |
| 12.15. | Affiliat | ed Communications e Liability c Performance | .60 |
| 12.17. | Time is | of the Essence arent Guaranty | 61 |

Exhibits

| Exhibit A | _ | Form of Settlement Statement |
|-----------|---|--|
| Exhibit B | | R&W Policy |
| Exhibit C | | Form of Assignment of Interests (Acquired Interests) |
| Exhibit D | | Form of Transition Services Agreement |
| Exhibit E | | Letter Agreement Amendment |
| Exhibit F | | Excluded Liability |
| | | |
| Annexes | | |

- Illustrative Working Capital Calculation Accounting Principles Annex 1
- Annex 2 ____

Schedules

| Schedule 1.1(a)-PLOpermitted LiensSchedule 2.6Included SystemsSchedule 3.4Governmental Approvals (Seller)Schedule 3.4Governmental Approvals (Seller)Schedule 3.4Brokers' Fees (Seller)Schedule 4.2Non-Contravention (Acquired Companies)Schedule 4.3Governmental Approvals (Acquired Companies)Schedule 4.4(a)Ownership of Equity InterestsSchedule 4.5- Compliance with LawSchedule 4.6(b)- Leased Real PropertySchedule 4.6(c)- EasementsSchedule 4.6(d)- Real Property LiensSchedule 4.6(d)- Condition of AssetsSchedule 4.8(a)- Material Customers & Material SuppliersSchedule 4.8(a)- Material Customers & Material SuppliersSchedule 4.8(b)- Communications from Material Customers & Material SuppliersSchedule 4.9(c)- Undisclosed LiabilitiesSchedule 4.9(c)- Undisclosed Certain Changes (Acquired Companies)Schedule 4.9(b)- GAAPSchedule 4.11- Absences of Certain Changes (Acquired Companies)Schedule 4.12- Environmental MattersSchedule 4.13- Legal Proceedings (Acquired Companies)Schedule 4.14- DebtSchedule 4.15- Devironmental MattersSchedule 4.16- TaxesSchedule 4.18(a)- Intellectual PropertySchedule 4.18(b)- Intellectual PropertySchedule 4.18- PermitsSchedule 4.19- InsuranceSchedule 4.19- InsuranceSchedule 4.19- Bank Acc | Schedule 1.1(a) | _ | OCG and OCC Percentages |
|---|---|---|--|
| Schedule 2.6-Included SystemsSchedule 3.3-Non-Contravention (Seller)Schedule 3.4-Governmental Approvals (Seller)Schedule 3.6(a)-Ownership of Acquired InterestsSchedule 4.3-Brokers' Fees (Seller)Schedule 4.3-Governmental Approvals (Acquired Companies)Schedule 4.3-Governmental Approvals (Acquired Companies)Schedule 4.4(a)-Ownership of Equity InterestsSchedule 4.5-Compliance with LawSchedule 4.6(c)-EasementsSchedule 4.6(d)-Real PropertySchedule 4.6(d)-Real Property LiensSchedule 4.7(a)-AssetsSchedule 4.7(c)-Condition of AssetsSchedule 4.8(b)-Communications from Material SuppliersSchedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.13-Legal Proceetings (Acquired Companies)Schedule 4.16-TaxesSchedule 4.17-Environmental MattersSchedule 4.18-Information TechnologySchedule 4.10-DebtSchedule 4.11-Absence of Certain Changes (Acquired Companies)Schedule 4.13-Itegal Proceetings (Acquired Companies) <td></td> <td>_</td> <td></td> | | _ | |
| Schedule 3.3-Non-Contravention (Seller)Schedule 3.4-Governmental Approvals (Seller)Schedule 3.7-Brokers' Fees (Seller)Schedule 3.7-Brokers' Fees (Seller)Schedule 4.2-Non-Contravention (Acquired Companies)Schedule 4.3-Governmental Approvals (Acquired Companies)Schedule 4.4(a)-Ownership of Equity InterestsSchedule 4.4(a)-Ownership of Equity InterestsSchedule 4.6(a)-Compliance with LawSchedule 4.6(b)-Leased Real PropertySchedule 4.6(c)-EasementsSchedule 4.6(d)-Real Property LiensSchedule 4.7(a)-AssetsSchedule 4.7(b)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.13-Material ContractsSchedule 4.14-TaxesSchedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(a)-Intellectual PropertySchedule 4.19-Schedule PropertySchedule 4.19-TaxesSchedule 4.19-Taxes | | _ | |
| Schedule 3.4-Governmental Approvals (Seller)Schedule 3.6(a)-Ownership of Acquired InterestsSchedule 3.7-Brokers' Fees (Seller)Schedule 4.2-Non-Contravention (Acquired Companies)Schedule 4.3-Governmental Approvals (Acquired Companies)Schedule 4.4(a)-Ownership of Equity InterestsSchedule 4.5-Compliance with LawSchedule 4.6(b)-Leased Real PropertySchedule 4.6(c)-EasementsSchedule 4.6(d)-Real Property LiensSchedule 4.6(d)-Real Property LiensSchedule 4.7(a)-AssetsSchedule 4.8(b)-Communications from Material SuppliersSchedule 4.8(b)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.9(c)-DebtSchedule 4.10-DebtSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.16-TaxesSchedule 4.17-Absences of Certain Changes (Acquired Companies)Schedule 4.18-Intellectual PropertySchedule 4.19-InsuranceSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.14- <t< td=""><td></td><td>_</td><td></td></t<> | | _ | |
| Schedule 3.6(a)-Ownership of Acquired InterestsSchedule 3.7-Brokers' Fees (Seller)Schedule 4.2-Non-Contravention (Acquired Companies)Schedule 4.3-Governmental Approvals (Acquired Companies)Schedule 4.4(a)-Ownership of Equity InterestsSchedule 4.5-Compliance with LawSchedule 4.6(b)-Leased Real PropertySchedule 4.6(c)-EasementsSchedule 4.6(d)-Real Property LiensSchedule 4.7(a)-AssetsSchedule 4.7(c)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.9(b)-Conmunications from Material Customers & Material SuppliersSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.9(c)-Environmental MattersSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.14-TaxesSchedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.17-Bank AccountsSchedule 4.18(a)-Intellectual PropertySchedule 4.19-Information TechnologySchedule 4.12-Bank AccountsSchedule 4.14-Bank AccountsSchedule 4.15-Powers of AttorneySchedule 4.16-Taxes </td <td></td> <td></td> <td></td> | | | |
| Schedule 3.7-Brokers' Fees (Seller)Schedule 4.2-Non-Contravention (Acquired Companies)Schedule 4.3-Governmental Approvals (Acquired Companies)Schedule 4.4(a)-Ownership of Equity InterestsSchedule 4.6(a)-Ownership of Equity InterestsSchedule 4.6(b)-Leased Real PropertySchedule 4.6(c)-EasementsSchedule 4.6(d)-Real Property LiensSchedule 4.6(d)-Real Property LiensSchedule 4.7(a)-AssetsSchedule 4.7(b)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.14-Intellectual PropertySchedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.17-Bank AccountsSchedule 4.18(a)-Intellectual PropertySchedule 4.19-Intellectual PropertySchedule 4.10-PermitsSchedule 4.11-Bank AccountsSchedule 4.12-Powers of AttorneySchedule 4.13- <t< td=""><td></td><td></td><td></td></t<> | | | |
| Schedule 4.2-Non-Contravention (Acquired Companies)Schedule 4.3-Governmental Approvals (Acquired Companies)Schedule 4.4(a)-Ownership of Equity InterestsSchedule 4.5-Compliance with LawSchedule 4.6(b)-Leased Real PropertySchedule 4.6(c)-EasementsSchedule 4.6(d)-Real Property LiensSchedule 4.6(d)-Real Property LiensSchedule 4.7(a)-AssetsSchedule 4.8(a)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.19-InsuranceSchedule 4.19-InsuranceSchedule 4.19-Non-Contravention (GC and OCC)Schedule 4.19-Frees (Acquired Companies)Schedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 5.2(a)-Non-Contravention (O | 승규가 있는 것 같은 것 같 | | |
| Schedule 4.3-Governmental Approvals (Acquired Companies)Schedule 4.4(a)-Ownership of Equity InterestsSchedule 4.5-Compliance with LawSchedule 4.6(a)-Owned Real PropertySchedule 4.6(b)-Leased Real PropertySchedule 4.6(c)-EasementsSchedule 4.6(d)-Real Property LiensSchedule 4.7(a)-AssetsSchedule 4.7(c)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.9(d)-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.14-DebtSchedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.17-InsuranceSchedule 4.18(a)-Intellectual PropertySchedule 4.18(a)-InsuranceSchedule 4.19-InsuranceSchedule 4.19-Bark AccountsSchedule 4.19-Bark AccountsSchedule 4.20-Affiliate TransactionsSchedule 4.21-Powers of AttorneySchedule 4.23-Broke | | | 이 것 같은 것 같 |
| Schedule 4.4(a)-Ownership of Equity InterestsSchedule 4.6(a)-Compliance with LawSchedule 4.6(a)-Owned Real PropertySchedule 4.6(b)-Leased Real PropertySchedule 4.6(c)-EasementsSchedule 4.6(d)-Real Property LiensSchedule 4.6(d)-Real Property LiensSchedule 4.7(a)-AssetsSchedule 4.7(a)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.14-PermitsSchedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 4.24-Bank AccountsSchedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.2(b)- | | | |
| Schedule 4.5-Compliance with LawSchedule 4.6(a)-Owned Real PropertySchedule 4.6(b)-Leased Real PropertySchedule 4.6(c)-EasementsSchedule 4.6(d)-Real Property LiensSchedule 4.6(d)-ImprovementsSchedule 4.7(c)-Condition of AssetsSchedule 4.7(c)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.19-InsuranceSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 4.24-Affiliate TransactionsSchedule 4.25(a)-Non-Contravention (OGC and OCC)Schedule 4.21-Bank AccountsSchedule 4.23-< | | | |
| Schedule 4.6(a)–Owned Real PropertySchedule 4.6(b)–Leased Real PropertySchedule 4.6(c)–EasementsSchedule 4.6(d)–Real Property LiensSchedule 4.6(d)–ImprovementsSchedule 4.7(a)–AssetsSchedule 4.7(c)–Condition of AssetsSchedule 4.8(a)–Material Customers & Material SuppliersSchedule 4.8(b)–Communications from Material Customers & Material SuppliersSchedule 4.9(a)–Financial StatementsSchedule 4.9(b)–GAAPSchedule 4.9(c)–Undisclosed LiabilitiesSchedule 4.10–DebtSchedule 4.11–Absences of Certain Changes (Acquired Companies)Schedule 4.12–Environmental MattersSchedule 4.13–Legal Proceedings (Acquired Companies)Schedule 4.13–Intellectual PropertySchedule 4.16–TaxesSchedule 4.18(a)–Intellectual PropertySchedule 4.18(b)–Information TechnologySchedule 4.18(b)–Information TechnologySchedule 4.21–Bank AccountsSchedule 4.23–Powers of AttorneySchedule 4.23–Non-Contravention (OGC and OCC)Schedule 4.24–Governmental Approvals (OGC and OCC)Schedule 5.3–Capitalization (OGC and OCC)Schedule 5.4–Absence of Certain Changes (OGC and OCC)Schedule 5.4–A | 지 않고 그는 것 같은 것 같은 것은 것 같은 것 같은 것 같은 것은 것은 것은 것 같은 것 같 ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? ? | | |
| Schedule 4.6(b)-Leased Real PropertySchedule 4.6(c)-EasementsSchedule 4.6(d)-Real Property LiensSchedule 4.6(d)-ImprovementsSchedule 4.7(a)-AssetsSchedule 4.7(c)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(a)-Communications from Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.16-TaxesSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(a)-Information TechnologySchedule 4.19-Bank AccountsSchedule 4.20-Affiliate TransactionsSchedule 4.21-Powers of AttorneySchedule 4.22-Governmental Approvals (OGC and OCC)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceeding (OGC and OCC)Schedule 5.5-Legal Proceeding (OGC and OCC)Schedule 5.5-Legal Proceeding (OGC and OC | | | 그 아님에 집에 집에 집에 집에 가지 않는 것이 집에 집에 집에 집에 집에 집에 있다. |
| Schedule 4.6(c)–EasementsSchedule 4.6(d)–Real Property LiensSchedule 4.6(d)–ImprovementsSchedule 4.7(a)–AssetsSchedule 4.7(c)–Condition of AssetsSchedule 4.8(a)–Material Customers & Material SuppliersSchedule 4.8(b)–Communications from Material Customers & Material SuppliersSchedule 4.9(b)–GAAPSchedule 4.9(c)–Undisclosed LiabilitiesSchedule 4.10–DebtSchedule 4.11–Absences of Certain Changes (Acquired Companies)Schedule 4.12–Environmental MattersSchedule 4.13–Material ContractsSchedule 4.15–PermitsSchedule 4.16–TaxesSchedule 4.18(a)–Intellectual PropertySchedule 4.19–Information TechnologySchedule 4.19–Bank AccountsSchedule 4.20–Affiliate TransactionsSchedule 4.21–Bank AccountsSchedule 4.22–Governmental Approvals (OGC and OCC)Schedule 4.23–Governmental Approvals (OGC and OCC)Schedule 5.5–Legal Proceeding (OGC and O | | | |
| Schedule 4.6(d)-Real Property LiensSchedule 4.6(j)-ImprovementsSchedule 4.7(a)-AssetsSchedule 4.7(c)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.14-PermitsSchedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(a)-Intellectual PropertySchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.21-Powers of AttorneySchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 4.24-CommonicationsSchedule 4.25-Capitalization (OGC and OCC)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)< | | | |
| Schedule 4.6(j)-ImprovementsSchedule 4.7(a)-AssetsSchedule 4.7(c)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.9(c)-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.18(b)-Information TechnologySchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 4.23-Borkers' Fees (Acquired Companies)Schedule 4.24-Communication (OGC and OCC)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Pro | | | |
| Schedule 4.7(a)-AssetsSchedule 4.7(c)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.19-Information TechnologySchedule 4.19-InsuranceSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 4.21-Governmental Approvals (OGC and OCC)Schedule 4.23-Governmental Approvals (OGC and OCC)Schedule 5.2 (a)-Absence of Certain Changes (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC) | | | |
| Schedule 4.7(c)-Condition of AssetsSchedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.14-TaxesSchedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.19-InsuranceSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.4-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC) | u , | | |
| Schedule 4.8(a)-Material Customers & Material SuppliersSchedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.16-TaxesSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.19-Information TechnologySchedule 4.20-Affiliate TransactionsSchedule 4.21-Powers of AttorneySchedule 4.23-Bank AccountsSchedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC) | | | |
| Schedule 4.8(b)-Communications from Material Customers & Material SuppliersSchedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.21-Bank AccountsSchedule 4.21-Bank AccountsSchedule 4.21-Bank AccountsSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 4.21-Governmental Approvals (OGC and OCC)Schedule 5.2(a)-Copitalization (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proc | | | |
| Schedule 4.9(a)-Financial StatementsSchedule 4.9(b)-GAAPSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.16-TaxesSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.22-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.2(b)-Governmental Approvals (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC) | | | |
| Schedule 4.9(b)-GAAPSchedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.19-InsuranceSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.2(b)-Governmental Approvals (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC) | | | |
| Schedule 4.9(c)-Undisclosed LiabilitiesSchedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.19-InsuranceSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.2(b)-Governmental Approvals (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC) | 승규가가 … 것은 " 이 가 있었던 것 이 것 것 것 것 것 것 것 같아. 말 것 | | |
| Schedule 4.10-DebtSchedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.19-InsuranceSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.1-Subject Employees | 신경 양 전 집에 가지 않는 것이 같아요. 김 가슴이 많이 많이 많이 했다. | | |
| Schedule 4.11-Absences of Certain Changes (Acquired Companies)Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.19-InsuranceSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC) | | | |
| Schedule 4.12-Environmental MattersSchedule 4.13-Material ContractsSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.19-InsuranceSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Subject Employees | | | |
| Schedule 4.13-Material ContractsSchedule 4.13-Legal Proceedings (Acquired Companies)Schedule 4.15-PermitsSchedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.19-InsuranceSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Subject Employees | | | 이는 것이 있는 것은 |
| Schedule 4.13–Legal Proceedings (Acquired Companies)Schedule 4.15–PermitsSchedule 4.16–TaxesSchedule 4.18(a)–Intellectual PropertySchedule 4.18(b)–Information TechnologySchedule 4.19–InsuranceSchedule 4.20–Affiliate TransactionsSchedule 4.21–Bank AccountsSchedule 4.23–Brokers' Fees (Acquired Companies)Schedule 5.2(a)–Non-Contravention (OGC and OCC)Schedule 5.3–Capitalization (OGC and OCC)Schedule 5.4–Absence of Certain Changes (OGC and OCC)Schedule 5.5–Legal Proceedings (OGC and OCC)Schedule 5.5–Legal Proceedings (OGC and OCC)Schedule 5.5–Subject Employees | | | |
| Schedule 4.15–PermitsSchedule 4.16–TaxesSchedule 4.18(a)–Intellectual PropertySchedule 4.18(b)–Information TechnologySchedule 4.19–InsuranceSchedule 4.20–Affiliate TransactionsSchedule 4.21–Bank AccountsSchedule 4.21–Bank AccountsSchedule 4.23–Brokers' Fees (Acquired Companies)Schedule 5.2(a)–Non-Contravention (OGC and OCC)Schedule 5.2(b)–Governmental Approvals (OGC and OCC)Schedule 5.3–Capitalization (OGC and OCC)Schedule 5.4–Absence of Certain Changes (OGC and OCC)Schedule 5.5–Legal Proceedings (OGC and OCC)Schedule 5.5–Subject Employees | | | |
| Schedule 4.16-TaxesSchedule 4.18(a)-Intellectual PropertySchedule 4.18(b)-Information TechnologySchedule 4.19-InsuranceSchedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.21-Powers of AttorneySchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.2(b)-Governmental Approvals (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 5.5-Subject Employees | | | |
| Schedule 4.18(a)–Intellectual PropertySchedule 4.18(b)–Information TechnologySchedule 4.19–InsuranceSchedule 4.20–Affiliate TransactionsSchedule 4.21–Bank AccountsSchedule 4.21–Powers of AttorneySchedule 4.23–Brokers' Fees (Acquired Companies)Schedule 5.2(a)–Non-Contravention (OGC and OCC)Schedule 5.2(b)–Governmental Approvals (OGC and OCC)Schedule 5.3–Capitalization (OGC and OCC)Schedule 5.4–Absence of Certain Changes (OGC and OCC)Schedule 5.5–Legal Proceedings (OGC and OCC)Schedule 7.13–Subject Employees | | | |
| Schedule 4.18(b)–Information TechnologySchedule 4.19–InsuranceSchedule 4.20–Affiliate TransactionsSchedule 4.21–Bank AccountsSchedule 4.21–Powers of AttorneySchedule 4.23–Brokers' Fees (Acquired Companies)Schedule 5.2(a)–Non-Contravention (OGC and OCC)Schedule 5.2(b)–Governmental Approvals (OGC and OCC)Schedule 5.3–Capitalization (OGC and OCC)Schedule 5.4–Absence of Certain Changes (OGC and OCC)Schedule 5.5–Legal Proceedings (OGC and OCC)Schedule 7.13–Subject Employees | | | |
| Schedule 4.19–InsuranceSchedule 4.20–Affiliate TransactionsSchedule 4.21–Bank AccountsSchedule 4.21–Powers of AttorneySchedule 4.23–Brokers' Fees (Acquired Companies)Schedule 5.2(a)–Non-Contravention (OGC and OCC)Schedule 5.2(b)–Governmental Approvals (OGC and OCC)Schedule 5.3–Capitalization (OGC and OCC)Schedule 5.4–Absence of Certain Changes (OGC and OCC)Schedule 5.5–Legal Proceedings (OGC and OCC)Schedule 7.13–Subject Employees | | | |
| Schedule 4.20-Affiliate TransactionsSchedule 4.21-Bank AccountsSchedule 4.21-Powers of AttorneySchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.2(b)-Governmental Approvals (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 7.13-Subject Employees | | | |
| Schedule 4.21-Bank AccountsSchedule 4.21-Powers of AttorneySchedule 4.23-Brokers' Fees (Acquired Companies)Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.2(b)-Governmental Approvals (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 7.13-Subject Employees | | | |
| Schedule 4.21–Powers of AttorneySchedule 4.23–Brokers' Fees (Acquired Companies)Schedule 5.2(a)–Non-Contravention (OGC and OCC)Schedule 5.2(b)–Governmental Approvals (OGC and OCC)Schedule 5.3–Capitalization (OGC and OCC)Schedule 5.4–Absence of Certain Changes (OGC and OCC)Schedule 5.5–Legal Proceedings (OGC and OCC)Schedule 7.13–Subject Employees | | | |
| Schedule 4.23–Brokers' Fees (Acquired Companies)Schedule 5.2(a)–Non-Contravention (OGC and OCC)Schedule 5.2(b)–Governmental Approvals (OGC and OCC)Schedule 5.3–Capitalization (OGC and OCC)Schedule 5.4–Absence of Certain Changes (OGC and OCC)Schedule 5.5–Legal Proceedings (OGC and OCC)Schedule 7.13–Subject Employees | | - | |
| Schedule 5.2(a)-Non-Contravention (OGC and OCC)Schedule 5.2(b)-Governmental Approvals (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 7.13-Subject Employees | | - | 이 아이는 것이 아이에 가지 않는 것이 같은 것이 같은 것이 같은 것이 같다. |
| Schedule 5.2(b)-Governmental Approvals (OGC and OCC)Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 7.13-Subject Employees | | _ | |
| Schedule 5.3-Capitalization (OGC and OCC)Schedule 5.4-Absence of Certain Changes (OGC and OCC)Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 7.13-Subject Employees | | | |
| Schedule 5.4–Absence of Certain Changes (OGC and OCC)Schedule 5.5–Legal Proceedings (OGC and OCC)Schedule 7.13–Subject Employees | And an University of the second state of the second | _ | |
| Schedule 5.5-Legal Proceedings (OGC and OCC)Schedule 7.13-Subject Employees | | _ | |
| Schedule 7.13 – Subject Employees | | _ | |
| | | - | |
| Schedule 7.17 – Bonds | | - | |
| | Schedule 7.17 | - | Bonds |

i

| Schedule 7.18 | - | Insurance Policy Matters |
|---------------------|-------|--------------------------|
| Schedule 9.2(b)(vi) | 10.00 | Resignations Persons |

PURCHASE AND SALE AGREEMENT

THIS PURCHASE AND SALE AGREEMENT (this "Agreement"), dated as of March 22, 2024 (the "Execution Date"), is made and entered into by and among Summit Midstream Opco, LP, a Delaware limited partnership ("Seller"), and MarkWest Utica Operating Company, L.L.C., a Delaware limited liability company ("Buyer"), and solely for purposes of <u>Section 12.18</u> hereof, Summit Midstream Partners, L.P., a Delaware limited partnership ("Seller and Buyer is sometimes referred to individually in this Agreement as a "Party" and collectively as the "Parties."

RECITALS

WHEREAS, Seller owns all of the rights, title and interest in the Equity Interests of the Acquired Company which constitutes 100% of the Equity Interest in the Acquired Company (the "Acquired Interests");

WHEREAS, the Acquired Company directly owns the OGC Interests and the OCC Interests (collectively, the "Non-Op Interests");

WHEREAS, upon the terms and subject to the conditions of this Agreement, Seller desires to sell, and Buyer desires to purchase, the Acquired Interests at the Closing in exchange for payment of the consideration specified in this Agreement;

WHEREAS, Buyer and Seller desire to make certain representations, warranties and agreements in connection with, and establish various conditions to, the purchase and sale of the Acquired Interests; and

WHEREAS, simultaneously with the execution of this Agreement, the Parties and/or their respective Affiliates are entering into the Letter Agreement Amendment.

AGREEMENTS

NOW, THEREFORE, in consideration of the representations, warranties, agreements and covenants contained in this Agreement, and other good and valuable consideration, the receipt and legal sufficiency of which are hereby acknowledged, the Parties undertake and agree as follows:

ARTICLE I Definitions and Interpretations

1.1. Definitions. Capitalized terms used in this Agreement but not defined in the body of this Agreement shall have the meanings ascribed to them in this <u>Section 1.1</u>. Capitalized terms defined in the body of this Agreement are listed below with reference to the location of the definitions of such terms in the body of this Agreement.

"Accounting Firm" is defined in Section 2.3(b).

"Accounting Principles" means the accounting principles set forth on Annex 2.

"Acquired Company" means Summit Midstream Utica, LLC, a Delaware limited liability company.

"Acquired Company Group" means the Acquired Company, OGC and OCC, collectively.

"Acquired Company Group Member" means any member of the Acquired Company Group.

"Acquired Interests" is defined in the recitals to this Agreement.

"Adjustment Amount" means an amount of Dollars (expressed as a positive or a negative number, as applicable) determined as of the Measurement Time equal to the sum of (i) the Closing Date Working Capital Adjustment Amount, <u>minus</u> (ii) the Closing Debt Amount <u>plus</u> (iii) the Closing Date Cash.

"Affiliate" means a Person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, a specified Person; provided, however, that in no event shall any Affiliate of Seller, other than the Acquired Company Group, be deemed an Affiliate of the Acquired Company (other than for purposes of <u>Section 11.17</u> and the definition of Non-Recourse Party therein). The Acquired Company shall be considered an Affiliate of Seller prior to Closing and an Affiliate of Buyer after Closing.

"Affiliate Contracts" is defined in Section 4.20.

"Agreement" is defined in the preamble to this Agreement.

"Assets" of any Person means all properties of every kind, nature, character and description (whether real, personal or mixed, whether tangible or intangible and wherever situated), including the goodwill related thereto, operated, owned or leased by such Person.

"Assignment Agreement" means an assignment agreement, by and between Seller and Buyer, in substantially the form attached hereto as <u>Exhibit C</u>, evidencing the assignment and transfer to Buyer of the Acquired Interests.

"Balance Sheet Date" means December 31, 2023.

"Business" means, with respect to each Acquired Company Group Member, the ownership and operation by such Acquired Company Group Member, as applicable, of its respective Assets and other activities conducted by such Acquired Company Group Member that are incidental thereto as of the Closing Date.

"Business Day" means any day that is not a Saturday, Sunday or other day on which commercial banks in Delaware or Houston, Texas are authorized or obligated to be closed by applicable Laws.

"Buyer" is defined in the preamble to this Agreement.

"Buyer Disclosure Schedule" means the disclosure schedule to this Agreement prepared by Buyer and delivered to Seller on the Closing Date.

"Buyer Indemnified Party" is defined in Section 11.2.

"Cash and Cash Equivalents" means (a) money, currency or a credit balance in a deposit account at a financial institution, net of checks outstanding as of the time of determination, (b) marketable securities or direct obligations issued or unconditionally guaranteed by the United States government or issued by any agency thereof and backed by the full faith and credit of the United States, (c) marketable direct obligations issued by any state of the United States of America or any political subdivision of any such state or any public instrumentality thereof, (d) commercial paper issued by any bank or any bank holding company owning any bank, and (e) certificates of deposit or bankers' acceptances issued by any commercial bank organized under the applicable Laws of the United States of America, in each case, calculated in a manner consistent with GAAP. For the avoidance of doubt, Cash and Cash Equivalents shall include any security, escrow or similar deposits, and any deposits or cash held as a guarantee in respect of performance of Contracts or as collateral in respect of outstanding insurance policies, leases or letters of credit, or credit card receivables.

"Closing" is defined in Section 9.1.

"Closing Amount" is defined in Section 2.2.

"Closing Date" is defined in Section 9.1.

"Closing Date Cash" means, as of the Measurement Time, the aggregate amount of all Cash and Cash Equivalents held by the Acquired Company determined in accordance with the Accounting Principles.

"Closing Date Working Capital" means an amount of Dollars (expressed as a positive or negative number, as applicable) equal to (a) the Current Assets minus (b) the Current Liabilities.

"Closing Date Working Capital Adjustment Amount" means an amount of Dollars (expressed as a positive or negative number, as applicable) equal to the amount by which the Closing Date Working Capital is greater than or less than the Estimated Working Capital.

"Closing Debt Amount" means the aggregate amount of all unpaid Debt of the Acquired Company immediately prior to Closing as determined in accordance with the Accounting Principles.

"Closing Purchase Price" is defined in Section 2.3(a).

"Code" means the Internal Revenue Code of 1986, as amended.

"Confidentiality Agreement" means that certain confidentiality agreement, dated as of September 15, 2023, by and between MPLX LP and Summit Midstream Holdings, LLC. "Contract" means any legally binding oral or written agreement, lease, license, note, evidence of indebtedness, mortgage, security agreement, understanding, instrument, undertaking, or obligation, including any amendment, modification, or supplement thereto or other legally binding arrangement.

"control" means possession, directly or indirectly (through one or more intermediaries), of the power to direct or cause the direction of the management or policies (whether through ownership of securities or any partnership or other ownership interest, by contract or otherwise) of a Person.

"Corporate Encumbrances" shall mean, with respect to the Equity Interests of any Person, (a) any transfer restrictions imposed by federal or state securities Laws, (b) any transfer restrictions contained in the Organizational Documents of such Person, (c) Liens created by this Agreement, (d) Liens arising by, through or under Buyer or any of its Affiliates or (e) Liens that shall be released, waived or otherwise terminated in connection with the Closing.

"COVID-19" means the COVID-19 or SARS-CoV-2 virus, and any evolutions or mutations thereof or related or associated epidemics, pandemics, public health emergencies or disease outbreaks.

"Creditors' Rights" is defined in Section 3.2(b).

"Current Assets" means the current assets of the Acquired Company as of the Measurement Time, as illustrated on and specifically including the line items listed on <u>Annex 1</u> (excluding all Cash and Cash Equivalents), each determined in accordance with the Accounting Principles, but excluding any and all deferred Tax assets.

"Current Liabilities" means the current Liabilities of the Acquired Company as of the Measurement Time, as illustrated on and specifically including the line items listed on <u>Annex 1</u>, each determined in accordance with the Accounting Principles, but excluding (a) any and all deferred Tax Liabilities, (b) any Debt of the Acquired Company, including the Closing Debt Amount, and (c) any Transaction Expenses.

"D&O Claim" is defined in Section 7.4(a).

"D&O Indemnified Persons" is defined in Section 7.4(a).

"Debt" of any Person means, without duplication, (a) all indebtedness for borrowed money, (including all principal, accrued interest, premiums, late charges, collection fees, penalties, termination fees or breakage fees), (b) indebtedness evidenced by any note, bond, debenture, mortgage or other debt instrument or debt security, (c) any obligations of such Person for deferred and unpaid purchase price of property or services (other than trade payables and other obligations incurred in the ordinary course of business), conditional sale or title retention arrangements; (d) any reimbursement obligations of such Person with respect to letters of credit, banker's acceptances issued for account of the Acquired Company, to the extent drawn; (e) any Liabilities of such Person under any interest rate swaps, collars, caps, foreign currency exchange agreement or similar hedging agreement (including any breakage cost or termination fee); (f) any "earn out" or other similar payment obligations of such Person, or (g) guarantees with respect to any indebtedness or other obligation of any other Person of a type described in <u>clauses (a)</u> through (f) above; provided that Debt shall not include (x) any obligations of such Person as lessee under leases that have been or should be, in accordance with GAAP, recorded as capital or finance leases, (y) any accounts payable to trade creditors, purchase commitments incurred in the ordinary course of business, or accrued expenses and deferred revenues, in each case, to the extent included as Current Liabilities in the calculation of Closing Date Working Capital or (z) any Liabilities or obligations among any of the Acquired Company Group Members.

"Direct Claim" is defined in Section 11.5(b).

"Easements" means all easements, authorizations, rights of way, servitudes, property use agreements, line rights, surface leases, and real property licenses (including right of way Permits from railroads and road crossing Permits or other right of way Permits from any Governmental Authority) relating to Real Property used in the Business of the Acquired Company but owned by third parties.

"Employee Pension Benefit Plan" has the meaning set forth in Section 3(2) of ERISA.

"Employee Welfare Benefit Plan" has the meaning set forth in Section 3(1) of ERISA.

"Environmental Laws" means any and all Laws pertaining to prevention of pollution, protection of the environment (including natural resources), remediation of contamination (including the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. § 9601 et seq.) and workplace health and safety (to the extent such health and safety relates to exposure to Hazardous Substances).

"Equity Interests" means capital stock, partnership or membership interests or units (whether general or limited), and any other interest or participation that confers on a Person the right to receive a share of the profits and/or losses of, or distribution of assets of, the issuing entity.

"Equity Securities" means (a) Equity Interests, (b) subscriptions, calls, warrants, options or commitments of any kind or character relating to, or entitling any Person to acquire, any Equity Interests, and (c) securities convertible into or exercisable or exchangeable for Equity Interests.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"Estimated Adjustment Amount" is defined in Section 2.3(a).

"Estimated Settlement Statement" is defined in Section 2.3(a).

"Estimated Working Capital" means \$5,322,974.

"Evaluation Material" is defined in Section 7.2(a).

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Excluded Liability" has the meaning set forth on Exhibit F.

"Execution Date" is defined in the preamble to this Agreement.

"Final Adjustment Amount" is defined in Section 2.3(b).

"Final Purchase Price" is defined in Section 2.3(b).

"Financial Security" is defined in Section 12.18.

"Final Settlement Date" is defined in Section 2.3(b).

"Final Settlement Statement" is defined in Section 2.3(b).

"Financial Statements" are defined in Section 4.9(a).

"Formation Date" means October 29, 2014.

"Fraud" means, with respect to any Person, such Person's knowing and intentional common law fraud under the Laws of the State of Delaware with respect to the making by such Person of any of the representations and warranties contained in this Agreement; provided that, notwithstanding anything to the contrary, "Fraud" does not include any claim for equitable fraud, promissory fraud, unfair dealings fraud, or any torts (including a claim for fraud) based on negligence or recklessness.

"GAAP" means generally accepted accounting principles in the United States of America.

"Governmental Authority" means any executive, legislative, judicial, regulatory or administrative agency, body, commission, department, board, court, tribunal, arbitration body or authority of the United States or any foreign country, or any state, local or other governmental subdivision thereof.

"Governmental Consents" is defined in Section 3.4.

"Hazardous Substances" means any material, substance or waste which is hazardous, acutely hazardous, toxic, or of similar effect based on words of similar import or regulatory effect under Environmental Law regulated by, or for which Liability or standards of conduct may be imposed under, Environmental Law, including any material, substance or waste defined or regulated in any Environmental Law as a hazardous substance, hazardous waste, hazardous material, solid waste, toxic substance, toxic waste, pollutant or contaminant, and including petroleum or any fraction thereof, petroleum products, natural gas, natural gas liquids, asbestos and asbestos-containing materials, radon, per- or polyfluoroalkyl substances, urea formaldehyde and polychlorinated biphenyls.

"Hire Date" is defined in Section 7.13(a).

"Hired Employee" is defined in Section 7.13(a).

"HSR Act" means Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Indemnified Party" is defined in Section 11.5(a).

"Indemnifying Party" is defined in Section 11.5(a).

"Intellectual Property" means all intellectual property rights arising from or in respect of the following: (a) patents and patent applications, including all reissuances, continuations, continuations-in-part, divisions, supplementary protection certificates, extensions and reexaminations thereof; (b) trademarks, service marks, logos, designs, trade names, trade dress, Internet domain names, and registrations and applications therefore, including the associated goodwill therewith; (c) copyrights, other rights in copyrightable works of authorship and registrations and applications therefore; and (d) trade secrets, know-how, and other confidential and proprietary information.

"Knowledge" means (a) with respect to Seller, the actual knowledge, after reasonable inquiry of their respective direct reports, of Heath Deneke and Bill Mault and (b) with respect to Buyer, the actual knowledge, after due investigation and inquiry, of Greg Floerke.

"Law" means any applicable law, statute, code, regulation, ordinance, order, rule, rule of common law, judgment, decree, injunction, franchise, permit, certificate, license or authorization of any Governmental Authority.

"Leased Real Property" is defined in Section 4.6(b).

"Letter Agreement Amendment" means that certain Letter Agreement Amendment, dated as of the Closing Date, by and between MarkWest Liberty Midstream & Resources, L.L.C. and Summit Parent, attached hereto as <u>Exhibit E</u>, effectuating the amendment of that certain Letter Agreement, dated as of June 4, 2013, by and between MarkWest Liberty Midstream & Resources, L.L.C. and Summit Parent.

"Liability" or "Liabilities" means all direct and indirect debts, liabilities, Taxes, losses, damages, claims, demands, costs, expenses, fines, awards, judgments, penalties, interest, guarantees, commitments and obligations of any kind, whether fixed, contingent or absolute, matured or unmature, asserted or unasserted, accrued or not accrued, known or unknown, due or to become due, whenever or however arising (including out of any Contract, under any Law or Proceeding, or tort based on negligence or strict liability).

"Lien" means, with respect to any property or asset, any mortgage, deed of trust, pledge, security interest, lien or other similar property interest or encumbrance in respect of such property or asset.

"Lien Release Letter" means the duly executed letters or other instruments from the administrative agent, collateral agent, collateral trustee and/or trustee, as applicable, releasing all Liens on the Acquired Interests and Non-Op Interests filed in connection with any indebtedness of the Seller involving the Acquired Interests and Non-Op Interests, and authorizing the filing of UCC-3 amendments reflecting such release, in each case at the Closing to be held in trust by Locke Lord LLP then released after the Seller's receipt of the Purchase Price.

"Loss" means any and all Liabilities, amounts paid in settlement, costs or expenses (including reasonable attorneys', consultants' fees and court costs). For the avoidance of doubt, any retention amount owed under the R&W Policy shall not be considered in the calculation of Loss.

"Material Adverse Effect" means any effect, event, change or development that, individually or in the aggregate with any such other effects, events, changes, occurrences, facts, circumstances or developments, is or would reasonably be expected to be materially adverse to (a) the Businesses, operations, Assets, financial condition or results of operations of the Acquired Company Group Members or (b) the ability of Seller to consummate any transaction contemplated by this Agreement or any Transaction Documents; provided, however, that "Material Adverse Effect" shall not include any effect, event, change or development, directly or indirectly, arising out of or attributable to: (i) any effect, event, change or development affecting the economy of the United States generally, including changes in the credit, debt, capital or financial markets (including changes in interest or exchange rates) or affecting the oil and gas industry (including changes in commodity prices, general market prices and regulatory changes affecting such industry generally and drilling, producing, gathering, processing, transportation, storing and marketing activity, costs or margins) generally or regionally; (ii) any changes in financial or securities markets in general and any fluctuations in currency exchange rates; (iii) any global, national or regional political or social actions or conditions, including acts of war (whether or not declared), armed hostilities or terrorism (including cyberterrorism), or the escalation or worsening thereof; (iv) acts of God, earthquakes, any weather-related or other force majeure event or natural disasters; (v) health conditions (including any epidemic, pandemic, or disease outbreak (including COVID-19), including any worsening of such conditions; (vi) a change in Laws or accounting rules (including GAAP) and any interpretations thereof from and after the Closing Date; (vii) any failure, in and of itself, by any Acquired Company Group Member to meet any internal or published projections, forecasts or revenue or earnings predictions for any period (it being understood that the facts or occurrences giving rise to such failure may be taken into account in determining whether there has been or will be, a Material Adverse Effect; (viii) any action required or permitted by this Agreement; (ix) the public announcement, pendency or completion of the transactions contemplated by this Agreement; or (x) action taken by Seller or any Affiliate of Seller with Buyer's written consent or that are otherwise permitted or prescribed hereunder; provided further, however, that any effect, event, change or development referred to in clauses (i) through (vi) immediately above shall be taken into account in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur to the extent that such effect, event, change or development has a material disproportionate and adverse effect on the Acquired Company Group Members, taken as a whole, compared to other participants in the industries in which the Acquired Company Group Members conduct their respective Businesses.

"Material Contracts" is defined in Section 4.13(b).

"Material Customers" is defined in Section 4.8(a).

"Material Permits" is defined in Section 4.14.

"Material Suppliers" is defined in Section 4.8(a).

"Measurement Time" means 11:59 p.m. Central Time on the day immediately prior to the Closing Date.

"Merits Hearing" is defined in Section 2.3(b).

"Non-Op Interests" is defined in the recitals to this Agreement.

"Non-Recourse Party" is defined in Section 12.15.

"Objection Notice" is defined in Section 2.3(b).

"OCC" means Ohio Condensate Company, L.L.C., a Delaware limited liability company.

"OCC Interests" means the issued and outstanding Equity Interests of OCC owned directly by the Acquired Company and set forth on <u>Schedule 1.1(a)</u> of the Seller Disclosure.

"OCC LLC Agreement" means that certain Second Amended and Restated Limited Liability Company Agreement of OCC, dated as of August 1, 2014, as amended.

"OGC" means Ohio Gathering Company, L.L.C., a Delaware limited liability company.

"OGC Interests" means the issued and outstanding Equity Interests of OGC owned directly by the Acquired Company and set forth on <u>Schedule 1.1(a)</u> of the Seller Disclosure Schedule.

"OGC LLC Agreement" means that certain Third Amended and Restated Limited Liability Company Agreement of OGC, dated as of August 1, 2014, as amended.

"Offered Employee" is defined in Section 7.13(a).

"Operations Services" has the meaning given to it in Section 2.6.

"Organizational Documents" means, with respect to any Person, the articles of incorporation, certificate of incorporation, certificate of formation, certificate of limited partnership, bylaws, limited liability company agreement, operating agreement, partnership agreement, stockholders' agreement and all other similar documents, instruments or certificates executed, adopted or filed in connection with the creation, formation or organization of such Person, including any amendments thereto.

"Other Ohio Interests" means the Equity Interests in each of OCC and OGC, other than the Non-Op Interests.

"Owned Intellectual Property" means all Intellectual Property owned by the Acquired Company and used in its Business.

"Owned Real Property" is defined in Section 4.6(a).

"Party" and "Parties" are defined in the preamble to this Agreement.

"Permit" means any permit, approval, consent, license, franchise, exemption and any other authorization, consent and approval of or from Governmental Authorities.

"Permitted Liens" means (a) statutory Liens for Taxes not yet due and payable or the amount or validity of which is being contested in good faith by appropriate proceedings and for which appropriate reserves have been established in accordance with GAAP, (b) mechanics', carriers', workers', repairers' and similar statutory Liens arising or incurred in the ordinary course of business not yet due and payable or the amount or validity of which is being contested in good faith by appropriate proceedings, (c) Liens contained in the Organizational Documents of an Acquired Company Group Member, (d) Liens created by the Buyer or its Affiliates, (e) Liens securing obligations reflected as liabilities in the Financial Statements or referred to in notes to the Financial Statements, (f) with respect to the Real Property, (i) zoning, entitlement, building and other land use regulations imposed by Governmental Authorities having jurisdiction over the Real Property and not violated by the current use and operation of the Real Property, and (ii) covenants, conditions, restrictions, easements and other similar matters that would be disclosed by an inspection or accurate survey of any parcel of Real Property; provided that such covenants, conditions, restrictions, easements and other similar matters, individually and in the aggregate, do not impair in any material respect the occupancy, use or operation of any Real Property for purposes of the Business, (g) Liens that shall be released, waived or otherwise terminated in connection with the Closing, and (h) those matters identified on Schedule 1.1(a).

"Person" means any natural person, corporation, limited partnership, general partnership, limited liability company, joint stock company, joint venture, association, company, estate, trust, bank trust company, land trust, business trust, or other organization, whether or not a legal entity, custodian, trustee-executor, administrator, nominee or entity in a representative capacity and any Governmental Authority.

"Personal Property" means the personal property interests owned by the applicable Acquired Company and directly or indirectly used or held for use in its Business (whether or not located on, over or across the Real Property), but specifically excluding the Real Property itself.

"Plan" means each (a) "employee benefit plan," as such term is defined in Section 3(3) of ERISA; (b) plan that would be an employee benefit plan described in <u>clause (a)</u> of this sentence if it was subject to ERISA, such as foreign plans and plans for directors; (c) equity bonus, equity ownership, equity option, restricted equity, equity purchase, equity appreciation rights, phantom equity or other equity-based compensation plan or arrangement; (d) bonus plan or arrangement, incentive award plan or arrangement, deferred compensation agreement or arrangement, change in control or retention plan or arrangement, executive compensation or supplemental income arrangement, personnel policy, vacation policy, severance pay plan, policy or agreement, consulting agreement or employment agreement; and (e) other employee benefit plan, agreement, arrangement, program, practice or understanding.

"Privileged Communications" is defined in Section 12.14.

"Proceeding" means any action, claim, counterclaim, complaint, investigation, mediation, petition, suit, arbitration proceeding, administrative or regulatory investigation, review, audit, proceeding, citation, summons or subpoena of any nature (civil, criminal, regulatory or otherwise) in Law or in equity.

"Purchase Price" is defined in Section 2.2.

"Real Property" means all real property used or held for use by the Acquired Company in the ownership, operation, or maintenance of the Assets and the Business, including the Owned Real Property, Leased Real Property and Easements.

"Real Property Leases" is defined in Section 4.6(b).

"Registered Intellectual Property" means all patents, patent applications, trademark registrations, trademark applications, copyright registrations, copyright applications, and domain names issued by a Governmental Authority and owned by and used in the Business.

"Release" means any depositing, spilling, leaking, pumping, pouring, placing, emitting, discarding, abandoning, emptying, discharging, migrating, injecting, escaping, leaching, dumping, or disposing.

"Representatives" means (a) partners, employees, officers, directors, members, equity owners and counsel of a Party or any of its Affiliates or any prospective purchaser of a Party or an interest in a Party; (b) any consultant, attorney, accountant or agent retained by a Party or the parties listed in <u>subsection (a)</u> above; and (c) any bank, other financial institution or entity funding, or proposing to fund, such Party's operations, including any consultant retained by such bank, other financial institution or entity.

"Review Period" is defined in Section 2.3(b).

"R&W Policy" means the buyer-side representations and warranties insurance policy and binder purchased and bound by Buyer as of the Closing Date, attached hereto as <u>Exhibit B</u>.

"Schedules" means the Buyer Disclosure Schedule and the Seller Disclosure Schedule, collectively.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Seller" is defined in the preamble to this Agreement.

"Seller Disclosure Schedule" means the disclosure schedule to this Agreement prepared by Seller and delivered to Buyer on the Closing Date.

"Seller Indemnified Party" is defined in Section 11.3.

"Seller Insurance Policies" is defined in Section 7.18(a).

"Seller Parent" is defined in the preamble to this Agreement.

"Seller's Marks" is defined in Section 7.15(a).

"Site Employees" is defined in Section 4.17(a).

"Subject Employee" means those certain employees listed on Schedule 7.13.

"Submission" is defined in Section 2.3(b).

"Subsidiary" or "Subsidiaries" means, with respect to any Person, another Person in which such first Person owns, directly or indirectly, an amount of the voting securities, other voting ownership or voting partnership interests of which is sufficient to elect at least a majority of its board of directors or other governing body (or, if there are no such voting interests, fifty percent (50%) or more of the Equity Interests of such Person).

"Summit Parent" means Seller Parent.

"Tax" or "Taxes" means any taxes and other similar governmental charges imposed by any Taxing Authority, including income, profits, gross receipts, net proceeds, alternative or add on minimum, ad valorem, value added, turnover, sales, use, property, personal property (tangible and intangible), environmental, stamp, leasing, lease, user, excise, duty, franchise, capital stock, transfer, registration, license, withholding, social security (or similar), unemployment, disability, payroll, employment, social contributions, fuel, excess profits, occupational, premium, windfall profit, severance, estimated, or similar other charge of any kind whatsoever, including any interest, penalty, or addition thereto, whether disputed or not.

"Tax Proceeding" is defined in Section 7.10(b).

"Tax Return" means any declaration, report, statement, form, return or other document or information required to be supplied to a Taxing Authority in connection with Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Taxing Authority" means, with respect to any Tax, the Governmental Authority or political subdivision thereof that imposes such Tax, and the agency (if any) charged with collection of such Tax for such entity or subdivision.

"third party" means a Person that is not a Party to this Agreement or an Affiliate of a Party to this Agreement.

"Third-Party Claim" is defined in Section 11.5(a).

"Transaction Documents" means this Agreement, the Assignment Agreement, the Transition Services Agreement and each other agreement, document and instrument required to be executed and delivered in accordance with this Agreement except for the Letter Agreement Amendment.

"Transaction Expenses" means (a) all investment banking fees, costs and expenses and legal fees, costs and expenses incurred by the Acquired Company and Seller in connection with the preparation for, negotiating or consummation of the transactions contemplated by this Agreement and the other Transaction Documents and (b) any transaction, retention, or change in control bonus in connection with the transactions contemplated by this Agreement, or similar compensatory amounts payable to any employees or service providers of the Acquired Company that become payable by any Acquired Company pursuant to an agreement with such Acquired Company, Seller or any of their respective Affiliates (in each case, excluding those at the express request of Buyer) solely as a result of the consummation of the transactions contemplated hereby

(including, in each case, the employer's share of any Tax withholding and any gross-up or similar payments for another Person's Taxes required to be paid in connection therewith).

"Transfer Taxes" is defined in Section 7.9.

"Transition Services Agreement" is defined in Section 9.2(a)(iii).

"Treasury Regulations" means the regulations (including temporary regulations) promulgated by the United States Department of the Treasury pursuant to and in respect of provisions of the Code. All references herein to Sections of the Treasury Regulations shall include any corresponding provision or provisions of succeeding, similar or substitute, temporary or final Treasury Regulations.

1.2. Interpretations. The rules of construction set forth in this Section 1.2 shall apply to the interpretation of this Agreement. All references in this Agreement to Exhibits, Annexes, Schedules, Articles, Sections, subsections, and other subdivisions of or to this Agreement refer to the corresponding Exhibits, Annexes, Schedules, Articles, Sections, subsections, and other subdivisions of or to this Agreement unless expressly provided otherwise. The Exhibits, Annexes and Schedules attached to this Agreement constitute a part of this Agreement and are incorporated herein for all purposes. Titles appearing at the beginning of any Articles, Sections, subsections, and other subdivisions of or to this Agreement are for convenience only, do not constitute any part of this Agreement, and shall be disregarded in construing the language hereof. The words "this Agreement," "herein," "hereby," "hereunder," and "hereof," and words of similar import, refer to this Agreement as a whole and not to any particular Article, Section, subsection, or other subdivision of or to this Agreement unless expressly so limited. The words "this Article," "this Section," and "this subsection," and words of similar import, refer only to the Article, Section or subsection hereof in which such words occur. Wherever the word "including" (in its various forms) is used in this Agreement, it shall be deemed to be followed by the words "without limiting the foregoing in any respect" and shall not be construed to limit any general statement that it follows to the specific items or matters immediately following it. Unless expressly provided to the contrary, if a word or phrase is defined, its other grammatical forms have a corresponding meaning. The words "shall" and "will" have the equal force and effect. Unless expressly provided to the contrary, the word "or", "either" or "any" shall not be exclusive. All references to "\$" or "Dollars" shall be deemed references to United States Dollars. Except as expressly provided herein, each accounting term not defined herein will have the meaning given to it under the Accounting Principles. Pronouns in masculine, feminine, or neuter genders shall be construed to state and include any other gender, and words, terms, and titles (including terms defined herein) in the singular form shall be construed to include the plural and vice versa, unless the context otherwise requires. Reference herein to any federal, state, local, or foreign Law shall be deemed to also refer to all rules and regulations promulgated thereunder, unless the context requires otherwise, and any reference to any Law in this Agreement shall only be a reference to such Law as of the Closing Date. Any reference herein to any Contract or other agreement or instrument means such Contract or other agreement or instrument as from time to time amended, modified, or supplemented, including, in the case of agreements or instruments, by waiver or consent. If any period of days referred to in this Agreement shall end on a day that is not a Business Day, then the expiration of such period shall automatically be extended until the end of the first succeeding Business Day. References to a Person are also to its permitted successors and permitted assigns. Unless otherwise indicated, with respect to Seller or an Acquired Company Group Member, the terms "ordinary course of business" or "ordinary course" shall be deemed to refer to the ordinary conduct of business in a manner consistent with the past practices and customs of Seller or the applicable Acquired Company Group Member. The phrase "made available" or "provided to" or similar phrases in respect of documents being made available or provided to Buyer means that any of Buyer, its Affiliates or its Representatives has had the opportunity to review such documents or materials at the offices of the Acquired Company Group Members or any of their respective Representatives or electronically by virtue of the electronic data room established by Seller or its Representatives in connection with the transactions contemplated hereby or any other physical or electronic means provided by Seller. Each Party acknowledges that it and its attorneys have been given an equal opportunity to negotiate the terms and conditions of this Agreement and that any rule of construction to the effect that ambiguities are to be resolved against the drafting Party or any similar rule operating against the drafter of an agreement shall not be applicable to the construction or interpretation of this Agreement.

ARTICLE II

Purchase and Sale of the Acquired Interests; Closing

2.1. Purchase and Sale of the Acquired Interests. Upon the terms and subject to the conditions of this Agreement, at the Closing, Seller shall sell, assign, convey, transfer and deliver the Acquired Interests to Buyer, and Buyer shall purchase and accept from Seller, the Acquired Interests, in exchange for the consideration in <u>Section 2.2</u>.

2.2. Purchase Price. Subject to the other terms and conditions of this Agreement, as full consideration for the sale, assignment, conveyance, transfer and delivery of the Acquired Interests, the aggregate consideration to be paid by Buyer to Seller is \$625,000,000 (the "Purchase Price"), subject to adjustment in accordance with the terms of this Agreement. At the Closing, an aggregate amount equal to the sum of (A) the Purchase Price, <u>plus</u> (B) the Estimated Adjustment Amount (such resulting sum, the "Closing Amount"), shall be paid by Buyer to Seller by wire transfer of immediately available funds in accordance with the wire transfer instructions provided by Seller to Buyer in advance of Closing.

2.3. Purchase Price Adjustments.

(a) Seller and Buyer have agreed to a preliminary settlement statement in substantially the form attached hereto as <u>Exhibit A</u> (the "Estimated Settlement Statement"), setting forth the estimated calculation of the Adjustment Amount (such estimated amount, the "Estimated Adjustment Amount") determined in accordance with the Accounting Principles, together with applicable wiring instructions. The Estimated Adjustment Amount was used for purposes of calculating the Purchase Price delivered at Closing, including the Closing Amount, and shall be subject to further adjustment pursuant to the provisions of <u>Section 2.3(b)</u>. The Purchase Price as paid to Seller at the Closing pursuant to this <u>Section 2.3(a)</u> shall be referred to as the "Closing Purchase Price".

(b) Not later than the sixtieth (60th) day following the Closing Date, Buyer shall prepare and deliver to Seller a statement in substantially the form attached hereto as <u>Exhibit A</u> and

attaching reasonable supporting schedules, working papers and all other relevant details in Buyer's possession to enable a review thereof by Seller (the "Final Settlement Statement") setting forth Buyer's good faith estimate of the final calculation of the Adjustment Amount (the "Final Adjustment Amount") determined in accordance with the Accounting Principles. Such process is not intended to permit the introduction of different components, judgments, accounting methods, policies, principles, practices, procedures, classifications or estimation methodologies for the purpose of preparing the Final Settlement Statement or determining the Final Adjustment Amount from the components, judgments, accounting methods, policies, principles, practices, procedures, classifications or estimation methodologies described in the Accounting Principles or used in determining the Estimated Working Capital and the Estimated Adjustment Amount. The Final Settlement Statement shall be determined in accordance with this Agreement and shall not take into account or otherwise give effect to the transactions to occur on or after the Closing Date. Buyer shall provide to Seller access to records of the Acquired Company as is reasonably necessary to enable Seller to review the Final Settlement Statement and the determinations to be contained therein. At any time during the thirty (30) day period following Seller's receipt of the Final Settlement Statement (the "Review Period"), Seller may deliver to Buyer one or more written reports or supplements thereto containing any changes that Seller proposes be made to the Final Settlement Statement (such written report, an "Objection Notice"). Seller shall be deemed to have waived any rights to object to the Final Settlement Statement unless Seller delivers an Objection Notice to Buver within the Review Period and, if the Review Period expires without Seller so delivering an Objection Notice, then the Final Settlement Statement and Final Adjustment Amount shall become final and binding on the Parties for all purposes of this Agreement and not subject to further audit or arbitration. If Seller timely delivers to Buyer an Objection Notice, all items in the Final Settlement Statement, other than such matters that are specifically disputed in the Objection Notice, shall be deemed to be irrevocably accepted and agreed to by Buyer and Seller. If Seller delivers an Objection Notice to Buyer during the Review Period, then Buyer and Seller shall undertake to agree on the Final Adjustment Amount no later than thirty (30) days after the date on which Seller delivered such Objection Notice to Buyer. In the event that such Parties cannot reach agreement within such thirty (30) day period, the Parties shall, within ten (10) days following the end of such thirty (30) day period, mutually engage and refer the remaining disputed matters to Ernst and Young, or if Ernst and Young is unable or unwilling to perform its obligations under this Section 2.3(b), such other nationally-recognized independent accounting firm as is mutually agreed on by Seller and Buyer or if Buyer and Seller cannot so agree within such time period then such other nationally-recognized independent accounting firm appointed by the Houston office of the American Arbitration Association as requested by Buyer or Seller (such firm that agrees to serve hereunder, the "Accounting Firm"). In connection with the engagement of the Accounting Firm, each Party will execute such engagement, indemnity and other agreements as the Accounting Firm and the American Arbitration Association may reasonably require as a condition to such engagement. Each Party will use reasonable efforts to cause the Accounting Firm to render its decision as soon as practicable after such engagement, including by promptly complying with all reasonable requests by the Accounting Firm for information, books, records and similar items; provided, however, notwithstanding anything to the contrary herein, no Party will disclose to the Accounting Firm, and the Accounting Firm will not consider for any purpose, any settlement discussions or settlement offer made by any Party. Within ten (10) days following the agreement of the Accounting Firm to serve hereunder, Buyer and Seller shall deliver to the Accounting Firm and Buyer or Seller, as applicable, (i) the Objection Notice and such work papers, invoices and other reports and information relating to the disputed matters as the Accounting Firm may request and (ii) Buyer's or Seller's, as applicable, proposed resolution of the disputed matters, limited to twenty (20) pages or less, and any materials it wishes to present to justify the resolution it so presents (the foregoing items (i) and (ii) together forming Buyer's or Seller's, as applicable, "Submission"). Within five (5) Business Days of receipt of the other party's Submission, each Party shall submit to the Accounting Firm a reply brief of ten (10) pages or less (exclusive of exhibits, schedules, or other attachments), with a simultaneous copy to the other party. If requested by either Party, within ten (10) Business days of the submission of the reply briefs pursuant to the foregoing sentence, the Accounting Firm shall conduct a hearing limited to one (1) day with time to be split equally between the Parties (the "Merits Hearing"). The Accounting Firm shall conduct the Merits Hearing in Houston, Texas in accordance with the Commercial Arbitration Rules of the American Arbitration Association, to the extent that such rules do not conflict with the terms of this Section 2.3(b). The Accounting Firm shall agree in writing to keep strictly confidential the specifics and existence of any matters submitted as well as all proprietary records of the Parties, if any, reviewed by the Accounting Firm in the process of resolving such disputes. Neither Party may have any exparte communications with the Accounting Firm concerning the Accounting Firm's determination of the disputed matters. The Accounting Firm shall act as an arbitrator for the limited purpose of determining the specific disputed matters submitted by either Seller or Buyer in their respective Submissions and reply briefs to the Accounting Firm, and whether and to what extent, if any, the Purchase Price requires adjustment as a result of the resolution of those disputed matters; provided, however, that if any of the disputed matters relate to the interpretation of the Parties' legal rights or obligations under this Agreement or the Transaction Documents, rather than financial or accounting matters pertinent to the calculation of the Final Adjustment Amount, such disputed matter shall instead be resolved in the manner set forth in Section 12.3 (with any dispute as to whether a disputed matter is legal or financial, or accounting-related in nature to be resolved solely by the Accounting Firm in its capacity as an arbitrator). The Accounting Firm may not award interest, damages or penalties. The Accounting Firm's determination shall be made within thirty (30) days after receipt of the Submissions (or after the Merits Hearing, as applicable) and, absent fraud or manifest error, shall be final and binding on Buyer and Seller, without right of appeal, and shall constitute an arbitral award upon which a judgment may be entered in any court having jurisdiction thereof. The Accounting Firm's determination shall be limited solely to resolution of the disputed items in accordance with this Section 2.3(b). In resolving any such disputed item, the Accounting Firm shall not assign a value to any item greater than the greatest value for such item proposed by Buyer or Seller nor assign a value to any item less than the smallest value for such item proposed by Buyer or Seller, as set forth in their respective Submissions, as applicable. The costs and expenses of the Accounting Firm in connection with resolving such disputed matters shall be borne by Buyer, on the one hand, and by Seller, on the other, based upon the percentage of which the portion of the contested amount not awarded to each Party bears to the amount actually contested by such Party. For instance, if Seller challenges the calculation of the Final Adjustment Amount in the Final Settlement Statement by an amount of \$100,000, but the Accounting Firm determines that Seller has a valid claim for only \$40,000, Seller shall bear 60% (i.e., \$60,000/\$100,000) of the fees and expenses of the Accounting Firm and Buyer shall bear the other 40% of such fees and expenses (i.e., \$40,000/\$100,000). The date upon which all adjustments and amounts in the Final Settlement Statement are agreed to (or deemed agreed to) by the Parties pursuant to this Section 2.3(b) or otherwise fully and finally determined by the Accounting Firm as set forth in this Section 2.3(b) shall be referred to as the "Final Settlement Date" and the final aggregate Purchase Price as adjusted by the Final Adjustment Amount determined accordingly shall be referred to as the "Final Purchase Price".

(c) Within ten (10) days after the Final Settlement Date, (i) if the Final Adjustment Amount is greater than the Estimated Adjustment Amount, then Buyer will pay to Seller the difference in Dollars between the Final Adjustment Amount and the Estimated Adjustment Amount and (ii) if the Final Adjustment Amount is less than the Estimated Adjustment Amount, then Seller will pay to Buyer the deficit by which the Final Adjustment Amount is less than the Estimated Adjustment Amount. The payment required by the preceding sentence shall be made by wire transfer in immediately available funds to the account or accounts designated in writing by the applicable Party within five (5) Business Days of the Final Settlement Date.

(d) If Buyer does not deliver the Final Settlement Statement within sixty (60) days after the Closing Date, the Estimated Settlement Statement shall be deemed to be the Final Settlement Statement for all purposes under this Agreement and the Parties shall be deemed to have waived any and all objection rights in connection therewith unless Seller, in its sole discretion, on the sixty-first (61^{tst}) day after Closing notifies the Buyer that it elects to prepare and deliver a draft Final Settlement Statement, in accordance with the guidelines agreed by the Parties in this Section 2.3, and delivers the same to Buyer on or prior to the ninetieth (90th) day after the Closing Date, and, in such case, Seller's objection rights under this Section 2.3 shall apply to Buyer mutatis mutandis.

2.4. Intentionally Deleted.

2.5. Withholding. Buyer shall be entitled to deduct and withhold from the consideration otherwise payable to any Person in connection with the transactions contemplated by this Agreement such amounts that Buyer is required to deduct and withhold with respect to the making of the relevant payment under applicable Law; provided, however, that other than any amounts treated as wages for income Tax purposes, Buyer shall use commercially reasonable efforts to notify Seller upon becoming aware of any Tax that Buyer intends to withhold pursuant to this <u>Section 2.5</u> and to provide the applicable payee with a reasonable opportunity to provide any certifications, documentation, or other information that may mitigate or eliminate such withholding. To the extent that amounts are withheld consistent with the terms of this <u>Section 2.5</u>, and duly and timely deposited with or otherwise paid to the appropriate Taxing Authority by Buyer, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made.

2.6. Express Exclusion of Operations Services from the Purchase and Sale. The Acquired Company utilizes, and its Business is currently dependent upon, certain operations, scheduling and distribution, accounting, contract administration, information technology, environmental, health and safety, land management and administration support services and software, including, but not limited to, gas and liquids monitoring and leak detection programs, processes and programs for gas and liquids measurement, allocation, settlement and distribution of monthly statements to producers, collection and disbursement of applicable Taxes, supervisory control and data acquisition, and various management systems for maintaining and monitoring compliance with applicable regulatory requirements and contractual obligations (collectively, the

"Operations Services") that have been implemented in connection with Summit Parent's and its Affiliates' enterprise-wide management program for midstream operations. The Operations Services are proprietary, and non-transferable; therefore, it is expressly agreed and acknowledged that Buyer is not receiving the Operations Services in connection with this Agreement and Buyer will rely solely upon its own enterprise-wide management program for operations, accounting, contract administration, information technology, environmental, health and safety, land management and administration support services and software. Notwithstanding the foregoing, Buyer shall receive all the tangible hardware, communications, networking and measurement instrumentation and equipment associated with the Acquired Company's Business operations conducted in the field, including, without limitation, such systems as set forth on <u>Schedule 2.6</u> hereto.

ARTICLE III

Representations and Warranties of Seller

Subject to the provisions of this <u>Article III</u> and the other terms and conditions of this Agreement and the exceptions and matters set forth on the Seller Disclosure Schedule, Seller hereby represents and warrants to Buyer as follows:

3.1. Organization; Qualification. Seller is a limited partnership duly formed, validly existing and in good standing under the Laws of the State of Delaware and has all requisite organizational power and authority to own, lease and operate its properties and to carry on its business as it is now being conducted, and is duly qualified, registered or licensed to do business as a foreign entity and is in good standing in each jurisdiction in which the property owned, leased or operated by it or the nature of its business makes such qualification necessary, except where the failure to be so duly qualified, registered or licensed and in good standing would not reasonably be expected to prevent or materially delay the consummation of the transactions contemplated by the Transaction Documents to which it is, or will be, a party or to materially impair its ability to perform its obligations under the Transaction Documents to which it is, or will be, a party.

3.2. Authority; Enforceability.

(a) Seller has the requisite power and authority to execute and deliver the Transaction Documents to which it is, or will be, a party, to carry out its obligations hereunder, and to consummate the transactions contemplated thereby and hereby. The execution, delivery and performance by Seller of the Transaction Documents to which it is, or will be, a party, and the consummation by it of the transactions contemplated thereby, have been duly and validly authorized by Seller, and no other limited liability company proceedings on the part of Seller are necessary to authorize the Transaction Documents to which it is, or will be, a party or to consummate the transactions contemplated by the Transaction Documents to which it is, or will be, a party.

(b) The Transaction Documents to which Seller is, or will be, a party have been (or will be, when executed and delivered at the Closing) duly executed and delivered by Seller, and, assuming the due authorization, execution and delivery by the other parties thereto, each Transaction Document to which Seller is, or will be, a party constitutes (or will constitute, when

executed and delivered at the Closing) the valid and binding agreement of Seller, enforceable against Seller in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium and similar Laws relating to or affecting creditors' rights generally and subject, as to enforceability, to legal principles of general applicability governing the availability of equitable remedies, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether such enforceability is considered in a proceeding in equity or at law) (collectively, "Creditors' Rights").

3.3. Non-Contravention. Except as set forth on Schedule 3.3 of the Seller Disclosure Schedule, the execution, delivery and performance of the Transaction Documents to which Seller is, or will be, a party by Seller and the consummation by Seller of the transactions contemplated thereby does not and will not: (a) result in any breach of any provision of the Organizational Documents of Seller; (b) require authorization or consent of, approval by, filing with or notice under, conflict with, constitute a default (or an event that with notice or passage of time or both would give rise to a default), or give rise to any right of termination, cancellation, amendment or acceleration (with or without the giving of notice, or the passage of time or both), under any of the terms, conditions or provisions of any Contract to which Seller is a party or by which any property or asset of Seller is bound or affected; or (c) except for any Governmental Consents (assuming compliance with the matters referred to in Section 3.4), violate any Law to which Seller is subject or by which any of Seller's properties or assets are bound, except, in the cases of clauses (b) and (c), for such defaults or rights of termination, cancellation, amendment, or acceleration or violations as would not reasonably be expected to prevent or materially delay the consummation of the transactions contemplated by the Transaction Documents to which Seller is, or will be, a party or to materially impair Seller's ability to perform its obligations under the Transaction Documents to which it is, or will be, a party.

3.4. Governmental Approvals. Except as set forth on <u>Schedule 3.4</u> of the Seller Disclosure Schedule, assuming the accuracy of Buyer's representation in Section 6.5 (HSR Act Approval) in this Agreement, no declaration, filing, registration, notice, waiver, license, or exemption with, or notice to, or authorization, consent or approval of, any Governmental Authority (collectively, "Governmental Consents") is necessary for the consummation by Seller of the transactions contemplated by the Transaction Documents to which it is a party, (a) other than such declarations, filings, registrations, notices, authorizations, consents, licenses, exemptions or approvals which (i) are customarily made or obtained after the Closing or (ii) if not obtained or made, would not reasonably be expected to prevent or materially delay the consummation of the transactions contemplated by the Transaction Documents to which Seller is, or will be, a party or to materially impair Seller's ability to perform its obligations under the Transaction Documents to which it is, or will be, a party.

3.5. Legal Proceedings. There are no Proceedings pending or, to the Knowledge of Seller, threatened against Seller that would reasonably be expected to prevent, enjoin or delay the consummation of the transactions contemplated by the Transaction Documents to which Seller is, or will be, a party or to materially impair Seller's ability to perform its obligations under the Transaction Documents to which it is, or will be, a party.

3.6. Ownership of Acquired Interests.

(a) Seller owns, directly or indirectly, the Acquired Interests, free and clear of all Liens, other than Corporate Encumbrances.

(b) Seller is not a party to any agreements, arrangements or commitments obligating Seller to grant, deliver, sell, or otherwise dispose of, or cause to be granted, delivered, sold, or otherwise dispose of the Acquired Interests, by sale, lease, license, grant or otherwise, other than this Agreement.

(c) There are no voting trusts, proxies or other agreements or understandings to which Seller is bound with respect to the voting of the Acquired Interests.

3.7. Brokers' Fees. Except as set forth on <u>Schedule 3.7</u> of the Seller Disclosure Schedule, no broker, investment banker, financial advisor or other Person is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Seller for which Buyer or the Acquired Company shall have any responsibility.

3.8. Bankruptcy. There are no bankruptcy, reorganization or receivership Proceedings pending, being contemplated by or, to Seller's Knowledge, threatened against Seller.

3.9. No Other Representations and Warranties. THE REPRESENTATIONS AND WARRANTIES OF SELLER CONTAINED IN ARTICLE III, ARTICLE IV AND ARTICLE V CONSTITUTE THE SOLE AND EXCLUSIVE REPRESENTATIONS AND WARRANTIES OF SELLER TO BUYER IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EXCEPT FOR SUCH REPRESENTATIONS AND WARRANTIES, NO PARTY OR ANY OTHER PERSON MAKES ANY OTHER EXPRESS OR IMPLIED REPRESENTATION OR WARRANTY WITH RESPECT TO SELLER OR THE ACQUIRED COMPANY GROUP, THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR THE ACQUIRED COMPANY GROUP'S BUSINESS, ASSETS, LIABILITIES, OPERATIONS, PROSPECTS OR CONDITION. EXCEPT FOR SUCH REPRESENTATIONS AND WARRANTIES, SELLER DISCLAIMS ALL LIABILITY AND RESPONSIBILITY FOR ANY REPRESENTATION, WARRANTY, PROJECTION. FORECAST, STATEMENT. OR **INFORMATION** MADE. COMMUNICATED, OR FURNISHED (ORALLY OR IN WRITING) TO ANY OTHER PARTY OR ITS AFFILIATES, OFFICERS, DIRECTORS, EMPLOYEES, AGENTS OR REPRESENTATIVES (INCLUDING OPINION, INFORMATION, PROJECTION OR ADVICE THAT MAY HAVE BEEN OR MAY BE PROVIDED TO ANY PARTY OR ANY OFFICER, DIRECTOR, EMPLOYEE, AGENT OR REPRESENTATIVE OF SUCH PARTY OR ANY OF ITS AFFILIATES).

ARTICLE IV

Representations and Warranties Regarding the Acquired Company

Subject to the provisions of this <u>Article IV</u> and the other terms and conditions of this Agreement and the exceptions and matters set forth on the Seller Disclosure Schedule, Seller hereby represents and warrants to Buyer, as follows; provided, however, that notwithstanding

anything to the contrary herein, none of the representations or warranties in this <u>Article IV</u> shall include or apply to OGC, OCC, the Non-Op Interests, or the Assets or the Business of OGC or OCC in any respect:

4.1. Organization; Qualification. The Acquired Company is a limited liability company duly formed, validly existing and in good standing under the laws of the State of Delaware. The Acquired Company has all requisite organizational power and authority to own, lease and operate its properties and to carry on its business as it is now being conducted, and is duly qualified, registered or licensed to do business as a foreign entity and is in good standing in each jurisdiction in which the property owned, leased or operated by it or the nature of its business makes such qualification necessary as required by Law. Seller has made available to Buyer true and complete copies of the Organizational Documents of the Acquired Company in effect on the Closing Date.

4.2. Non-Contravention. Except as set forth on <u>Schedule 4.2</u> of the Seller Disclosure Schedule, the execution, delivery and performance of the Transaction Documents by Seller, and the consummation of the transactions contemplated thereby by Seller, does not and will not: (a) result in any breach of any provision of the Organizational Documents of the Acquired Company; (b) require authorization or consent of, approval by, filing with or notice under, conflict with, constitute a default (or an event that with notice or passage of time or both would give rise to a default), or give rise to any right of termination, cancellation, amendment or acceleration (with or without the giving of notice, or the passage of time or both), under any of the terms, conditions or provisions of any Material Contract or Material Permit to which the Acquired Company is a party or by which any Asset of the Acquired Company is bound or affected; (c) except for any requisite Governmental Consents (assuming compliance with the matters referred to in Section 4.3), violate any Law to which the Acquired Company is subject or by which the Acquired Company's Assets are bound; or (d) constitute (with or without the giving of notice or the passage of time or both) an event which would result in the creation of any Lien (other than Permitted Liens) on the Acquired Company, its Business or Assets, except, in the cases of clauses (b), (c) and (d), for such defaults or rights of termination, cancellation, amendment, or acceleration, violations or Liens, as would not reasonably be expected to be material to the Acquired Company.

4.3. Governmental Approvals. Except as set forth on <u>Schedule 4.3</u> of the Seller Disclosure Schedule, no Governmental Consent needs to be obtained by the Acquired Company in connection with the consummation by Seller of the transactions contemplated by the Transaction Documents to which is, or will be, a party, other than such declarations, filings, registrations, notices, authorizations, consents, licenses, exemption or approvals which are customarily made or obtained after the Closing.

4.4. Capitalization; Subsidiaries.

(a) Seller owns, directly or indirectly, one hundred percent (100%) of the issued and outstanding Equity Securities of the Acquired Company, free and clear of all Liens, other than Corporate Encumbrances. <u>Schedule 4.4(a)</u> of the Seller Disclosure Schedule lists all of the issued and outstanding Equity Interests in the Acquired Company, including the names of the holders thereof and number or percentage of the Equity Interests held by each such holder. There are no

Equity Securities issued or outstanding in the Acquired Company other than as set forth on <u>Schedule 4.4(a)</u> of the Seller Disclosure Schedule.

(b) The Acquired Interests have been duly authorized, are validly issued, fully paid and nonassessable, and were not issued in violation of, and are not subject to, any preemptive rights, rights of first refusal, rights of first offer, purchase options, call options or other similar rights of any Person. None of the Acquired Interests are subject to any voting trust, member or partnership agreement or voting agreement or other agreement, right, instrument or understanding with respect to any purchase, sale, issuance, transfer, repurchase, redemption or voting of any Equity Securities. Except for the Acquired Interests, there are no limited liability company interests or other equity interest or voting securities of the Acquired Company outstanding.

(c) Except for this Agreement, (i) there are no (x) options, warrants, purchase rights, subscription rights, preemptive rights, conversion rights, exchange rights, calls, puts, rights of first refusal, or other Contracts or rights of similar nature that may require the Acquired Company to issue, sell, or otherwise cause to become outstanding or to acquire, repurchase, or redeem, any securities of the Acquired Company (including any securities convertible into or exchangeable or exercisable for any equity interest of the Acquired Company); or (y) stock appreciation, phantom stock, profit participation, or similar rights with respect to the Acquired Company, and (ii) there are no outstanding contractual obligations of the Acquired Company to repurchase, redeem or otherwise acquire any Equity Interest in the Acquired Company.

(d) Except for the Non-Op Interests, the Acquired Company does not own any Equity Securities or similar interest in, or any interest convertible into or exchangeable or exercisable for, at any time, any Equity Security or similar interest in, or any subscriptions, options, warrants, rights, calls, or other agreements or commitments for any Equity Security or similar interest in any Person.

4.5. Compliance with Law. Except as set forth on <u>Schedule 4.5</u> of the Seller Disclosure Schedule, (a) the Acquired Company has not received, in the seven (7) year period preceding the Closing Date, written notice of any material violation of or non-compliance with any Laws, including with respect to its ownership or operation of the Acquired Company's Assets or the operation of its Business, nor (b) is the Acquired Company, nor has it been during the seven (7) year period preceding the Closing Date, in violation of or non-compliance with any Laws in any material respect, including with respect to its ownership or operation of the Acquired Company's Assets or the operation of its Business; provided, however, that this <u>Section 4.5</u> does not address compliance with Environmental Laws, which is exclusively addressed by <u>Section 4.12</u>, matters relating to Taxes, which are exclusively addressed by <u>Section 4.16</u>.

4.6. Real Property.

(a) <u>Schedule 4.6(a)</u> of the Seller Disclosure Schedule contains a true and complete list of all of the real property and interests in real property owned in fee simple by the Acquired Company (the "Owned Real Property"). Seller has made available to Buyer true and complete copies of the vesting deeds with respect to the Owned Real Property listed on Schedule 4.6(a). (b) <u>Schedule 4.6(b)</u> of the Seller Disclosure Schedule contains a true and complete list of all of the leases, subleases, surface leases, or licenses of real property (the "Real Property Leases") by which the Acquired Company holds a leasehold interest (excluding for purposes hereof, any Easements or similar interests) (the "Leased Real Property"). Except as otherwise set forth on <u>Schedule 4.6(b)</u> of the Seller Disclosure Schedule, there exists no default (and no event has occurred which, with notice or lapse of time, would constitute a default) by the Acquired Company or, to the Knowledge of Seller, any other party, under any Real Property Lease. The Acquired Company has a valid, binding and enforceable leasehold interest in and to all of the Leased Real Property, free and clear of all Liens (except for Permitted Liens). Seller has made available to Buyer true and complete copies of the Real Property Leases listed on <u>Schedule 4.6(b)</u>.

(c) <u>Schedule 4.6(c)</u> of the Seller Disclosure Schedule contains a true and complete list of all of Easements not covered by the Real Property Leases required by the Acquired Company for the current ownership and operation of the Business, which if not held would materially adversely affect the current ownership or operation of the Business or the Assets. The Acquired Company has good and valid rights to utilize such Easements set forth on <u>Schedule 4.6(c)</u> of the Seller Disclosure Schedule, all of such Easements being free and clear of any Liens, except for Permitted Liens. Except as otherwise set forth on <u>Schedule 4.6(c)</u> of the Seller Disclosure Schedule, there exists no material default (and no event has occurred which, with notice or lapse of time, would constitute a material default) by the Acquired Company or, to the Knowledge of Seller, any other party, under any Easement set forth on <u>Schedule 4.6(c)</u> of the Seller Disclosure Schedule.

(d) The Acquired Company is the sole owner of and has good, insurable and marketable fee simple title to the Owned Real Property and a valid, binding and enforceable leasehold interest in and to all of the Leased Real Property and the Easements set forth on <u>Schedule</u> <u>4.6(c)</u> of the Seller Disclosure Schedule, free and clear of all Liens, except for Permitted Liens.

(e) The Acquired Company is not obligated under any right of first refusal, option or other contractual right to sell, assign or otherwise dispose of any Real Property, or any interest therein, to any Person other than Buyer.

(f) There are no leases, subleases, easements, licenses, concessions or other agreements granting to any Person other than the Acquired Company the right of use or occupancy of any portion of the Owned Real Property or the Leased Real Property, and no Person other than the Acquired Company is in possession of or has access to any portion of the Owned Real Property or the Leased Real Property.

(g) The Acquired Company has valid and adequate rights of ingress and egress via public rights-of-way with respect to the Real Property and all improvements and fixtures thereon. All public utilities, including water and wastewater, have been extended to a boundary line of each tract of the Real Property to the extent necessary for operation through adjoining public rights-of-way, or if they pass through adjoining private land, do so in accordance with valid Easements permitting such use, and all installation and connection charges necessary to use such utilities have been paid in full. All facilities located on the Owned Real Property or the Leased Real Property are supplied with utilities and other services, including without limitation gas, electricity, water, telephone, internet, sanitary sewer, and storm sewer as necessary for their current use, all of which

services are provided in accordance with Laws via public rights-of-way or valid easements benefiting such Real Property.

(h) Except as disclosed on Schedule 4.6(h), the Real Property includes all of the Owned Real Property, Real Property Leases and Easements necessary to access and operate the Assets of the Acquired Company as currently owned and operated, in all material respects, and constitutes all of the real property interests used in connection with the ownership and operation of the Acquired Company's Business as currently conducted. Except as disclosed on Schedule 4.6(h), neither Seller nor the Acquired Company has received any written notice with respect to any Real Property requiring the performance of any material structural or other material alterations to such Real Property, in each case, that have not been completed.

(i) Seller has not received any written notice from any Governmental Authority having jurisdiction over such Real Property or, to the Knowledge of Seller, any third Person, that the ownership or operation of any Real Property is currently in violation of any applicable Law(s). The operation of the Acquired Company's Business on the Real Property complies in all material respects with all applicable zoning Laws, building codes, Easements, covenants, conditions, restrictions and other instruments (whether recorded or unrecorded) affecting the Real Property.

(j) Except as set forth on <u>Schedule 4.6(j)</u>, all improvements, fixtures, and tangible Assets of the Acquired Company are located entirely within the boundaries of the Real Property and do not encroach on any setback, easement, or other real property interest of any other Person in any material respect.

(k) Neither Seller nor the Acquired Company has received any written notices with respect to any pending or, to Seller's Knowledge, threatened, Proceeding of condemnation, special assessment, or rezoning of any of the Real Property or improvements thereon.

(I) The Acquired Company (i) does not lease any real property other than the Leased Real Property and Easements and (ii) does not own any real property other than the Owned Real Property.

(m) Except as set forth on <u>Schedule 3.3</u> and the Permitted Liens, the Real Property is free and clear of all Liens.

4.7. Title to Properties and Assets; Condition of Assets.

(a) The Personal Property with a book value over \$50,000 is listed on <u>Schedule 4.7(a)</u>. Other than the assets associated with the Operations Services, the Personal Property constitutes, in all material respects, all of the tangible property necessary to conduct the operations of the Acquired Company's Business as the same is currently conducted.

(b) The Acquired Company has good and valid title to or a valid leasehold interest in all of the material Personal Property, in each case, free and clear of all Liens, except for Permitted Liens.

(c) Except as set forth on <u>Schedule 4.7(c)</u>, the Assets of the Acquired Company, including the improvements, buildings and facilities located on the Owned Real Property or the Leased Real Property, (i) have been maintained in working order and good operating condition generally consistent with assets of comparable age and usage based on Seller's experience in all material respects, except for ordinary wear and tear and ordinary and routine repairs and maintenance requirements, and (ii) are not in need of any repairs, which, if not made, would materially and adversely affect the integrity or safety of such Asset or affect the Acquired Company's ability to own, operate and maintain its Business after the Closing, in each case, in the same manner as owned, operated, and maintained immediately prior to the Closing Date.

4.8. Customers and Suppliers.

(a) <u>Schedule 4.8(a)</u> of the Seller Disclosure Schedule sets forth a list of the ten (10) largest customers ("Material Customers") and the ten (10) largest suppliers ("Material Suppliers") of the Acquired Company, as measured by the dollar amount of revenue received therefrom or purchased thereby, during the twelve (12)-month period ended December 31, 2023, showing the approximate total revenue received from each such Material Customer and the approximate total purchases by the Acquired Company from each such Material Supplier, as applicable, during such period.

(b) Except as listed in <u>Schedule 4.8(b)</u> of the Seller Disclosure Schedule, the Acquired Company has not received (i) any written notice that any of the Material Customers has ceased, or intends to cease after the Closing, to use the goods or services of the Acquired Company or to otherwise terminate or materially reduce its relationship with the Acquired Company or (ii) any written notice that any of the Material Suppliers has ceased, or intends to cease after the Closing, to supply goods or services to the Acquired Company or to otherwise terminate or materially reduce its relationship with otherwise terminate or materially reduce its relationship with the Acquired Company or to otherwise terminate or materially reduce its relationship with the Acquired Company.

4.9. Financial Statements.

(a) <u>Schedule 4.9(a)</u> of the Seller Disclosure Schedule contains the unaudited balance sheets of the Acquired Company and the related unaudited statements of income and members' equity as of the periods ended December 31, 2021, December 31, 2022, and December 31, 2023, and (ii) the unaudited balance sheets of the Acquired Company and the related unaudited statements of income and members' equity as of the Balance Sheet Date (collectively, the "Financial Statements").

(b) Except as set forth on <u>Schedule 4.9(b)</u> of the Seller Disclosure Schedule, the Financial Statements (i) have been prepared in accordance with GAAP applied on a consistent basis, and (ii) present fairly, in all material respects, the financial position and operating results, assets, combined statement of income and equity of the Acquired Company as of, and for the periods ended on, the respective dates thereof, subject, however, in the case of the unaudited Financial Statements, to normal year-end adjustments and accruals (the effect of which will not result in Material Adverse Effect) and the absence of notes and other textual disclosures (that, if presented, would not differ materially from those presented in the audited Financial Statements).

(c) Except as set forth on <u>Schedule 4.9(c)</u> of the Seller Disclosure Schedule, the Acquired Company does not have any Liability that would be required to be included in the Financial Statements prepared in accordance with GAAP except for (i) Liabilities adequately reflected or reserved against in the Financial Statements dated as of the Balance Sheet Date; or (ii) Liabilities that have arisen since the Balance Sheet Date in the ordinary course of business that are not, individually or in the aggregate, material.

(d) Since the Balance Sheet Date, Seller has not adopted or effected any material change in any method of accounting or accounting practice with respect to the Financial Statements, except for any change required because of a change in GAAP since the Balance Sheet Date.

4.10. Debt. Other than as set forth on <u>Schedule 4.10</u> of the Seller Disclosure Schedule, the Acquired Company does not have any Debt outstanding.

4.11. Absence of Certain Changes. Except as set forth on <u>Schedule 4.11</u> of the Seller Disclosure Schedule, since the Balance Sheet Date, (a) the Acquired Company has not suffered a Material Adverse Effect and (b) the Business of the Acquired Company has been operated and maintained substantially in the ordinary course of business of the Acquired Company.

4.12. Environmental Matters. Except as to matters set forth on <u>Schedule 4.12</u> of the Seller Disclosure Schedule:

(a) The Acquired Company is, and has been for the seven (7) years immediately preceding the Closing Date, in compliance with all Environmental Laws in all material respects.

(b) For the seven (7) years immediately preceding the Closing Date, the Acquired Company has obtained and possessed, as and when required pursuant to applicable Environmental Law, such Permits required under Environmental Law to operate its Business currently conducted and such Permits are in full force and effect, and applications for renewals or amendment thereof have been timely filed, and, for the seven (7) years immediately preceding the Closing Date, the Acquired Company has been and is in compliance in all material respects with the terms of all such Permits under which the Acquired Company operates.

(c) No Proceeding is pending, or, to the Knowledge of Seller, threatened, under Environmental Law regarding the Acquired Company's Business or its Assets, or, with respect to Permits under Environmental Laws, which would reasonably be expected to result in revocation of, or additional material restrictions in, any Permits held by the Acquired Company.

(d) For the seven (7) years immediately preceding the Closing Date, neither Seller with respect to the Acquired Company, nor the Acquired Company, has received a written notice of Proceedings or written information request asserting an alleged violation or other Liability of the Acquired Company under Environmental Law or any unauthorized Release by the Acquired Company of Hazardous Substances arising out of the ownership or operation of the Acquired Company's Business or Assets, other than notices with respect to matters that have been fully and finally resolved. (e) For the seven (7) years immediately preceding the Closing Date, there has been no Release of Hazardous Substances on, at, under, or from (i) any of the Real Properties of the Acquired Company, or (ii) any real property offsite the Real Properties where such Hazardous Substances were transported or disposed by the Acquired Company that, in each case of (i) or (ii), would reasonably be expected to give rise to any material remedial Liability pursuant to any Environmental Law for which Seller or the Acquired Company would have Liability.

(f) There is no pending order, judgment or decree (including any consent agreement or decree) or similar ruling with a Governmental Authority, applicable to the Acquired Company or the Real Property, arising from any violation by the Acquired Company of Environmental Laws or any Release by an Acquired Company of Hazardous Substances that has not been fully and finally resolved.

(g) Except as set forth in the Material Contracts and except for routine environmental covenants and indemnities in Contracts entered into in the ordinary course of business, the primary purpose of which is not the assumption of Liabilities under Environmental Laws, the Acquired Company has not, either expressly or by operation of Law, assumed or undertaken any material Liability, including any obligation for corrective or remedial action, of any third party relating to Environmental Laws.

4.13. Material Contracts.

(a) <u>Schedule 4.13</u> of the Seller Disclosure Schedule contains a true and complete listing of each of the following Contracts in effect to which the Acquired Company is a party:

(i) each hydrocarbon purchase and sale, gathering, transportation, treating, dehydration, processing or similar Contract and any Contract for the provision of services relating to gathering, compression, collection, processing, treating, disposal or transportation of natural gas, other hydrocarbons and saltwater involving annual expenditures or revenues in excess of \$200,000;

 (ii) each Contract that constitutes a pipeline interconnect or facility operating agreement;

(iii) each Contract involving a remaining commitment to pay capital expenditures in excess of \$200,000 in the aggregate;

(iv) each Contract that is a limited liability company, partnership, joint venture or similar agreement;

(v) each Contract relating to the acquisition or disposition of any business (whether by merger, sale of stock, sale of assets or otherwise) or granting to any Person a right of first refusal, first offer or right to purchase any of the Assets material to the conduct of the Acquired Company's Business; (vi) each Contract that impairs or restricts the ability of Seller (solely with respect to the Acquired Company's Business) or the Acquired Company (A) to compete in any line of business or in any geographic area during any period of time after the Closing, (B) solicit, hire or retain any Person as an employee, consultant or independent contractor, (C) acquire any product or services from any other Person, (D) solicit any customer of any Person, or (E) acquire any product or services from any other Person;

(vii) each Contract that licenses material Intellectual Property from a third party, other than "off the shelf" software licenses that are generally commercially available;

(viii) each swap, option, hedge or future Contracts;

(ix) each Contract that provides for "earn-outs" or other contingent payments which are outstanding or may become due and payable after the Closing;

(x) each settlement agreement involving the ongoing or future payment obligation by or to Acquired Company of \$200,000 or more;

(xi) each Contract relating to the incurrence, assumption or guarantee by Acquired Company of Debt, or under which any of the Assets are subjected to a Lien securing outstanding Debt or any other Contract that is a guaranty, performance, bid or completion or surety bond, in each case, in excess of \$200,000;

(xii) each Contract entered into during the prior five (5) year period that relates to the acquisition or disposition of Equity Interests or Assets of the Acquired Company (whether by merger, sale of stock, sale of assets. change of control or otherwise);

(xiii) each Contract for the lease or sublease (either as lessee, sublessee, lessor or sublessor) of (A) Personal Property or (B) Real Property, in each case which requires payments by the Acquired Company of more than \$500,000 in any calendar year;

(xiv) each Contract for (A) employment or (B) consulting services;

(xv) each Contract between (A) the Acquired Company, on the one hand, and any Governmental Authority, on the other, or (B) Acquired Company, on the one hand, and any prime contractor, higher tier subcontractor, or resellers to any Governmental Authority, on the other hand, by which Acquired Company agrees to provide goods or services that will ultimately be delivered to or otherwise benefit such Governmental Authority; (xvi) each Contract involving aggregate payments or receipts reasonably expected to be in excess of \$200,000 during the twelve (12) month period following the Closing Date that cannot be terminated by such Person or the Acquired Company, as applicable, upon sixty (60) days or less notice without payment of a material penalty or other material Liability;

(xvii)each Contract between the Seller and its Affiliates, on the one hand, and the Acquired Company, on the other hand;

(xviii) each Contract that provides as its primary purpose for the indemnification by the Acquired Company of any Person or the assumption of any Tax, environmental or other Liability of any Person; and

(xix) each outstanding written commitment to enter into any Contract of the type described in subsections (i) through (xviii) of this <u>Section 4.13(a)</u>.

(b) Each Contract set forth on <u>Schedule 4.13</u> of the Seller Disclosure Schedule (collectively, the "Material Contracts") is a valid and binding obligation of the Acquired Company, and is in full force and effect and enforceable in accordance with its terms against the Acquired Company and, to the Knowledge of Seller, the other parties thereto in accordance with its respective terms (assuming due authorization, execution and delivery by each counterparty to such Material Contract and that such Material Contract is a legal, valid and binding obligation on such counterparty), except, as enforcement may be limited by Creditors' Rights. Seller has made available to Buyer a true and complete copy of each Material Contract to which Seller has the right to provide to Buyer or, in the case of any oral Material Contract, a summary of all material terms.

(c) Except as set forth on <u>Schedule 4.13</u> of the Seller Disclosure Schedule, (i) the Acquired Company has performed in all material respects all obligations required to be performed by it under the Material Contracts; (ii) no counterparty to any Material Contract has exercised in writing any termination or non-renewal rights with respect thereto or has provided written notice to Seller or the Acquired Company of any dispute with respect to such Material Contract, in each case which has not been resolved; and (iii) neither Seller nor Acquired Company has received any written notice of the intention of any party to terminate, modify, amend, or accelerate any Material Contract. Neither the Acquired Company nor, to the Knowledge of Seller, any other party to any Material Contract is in default or breach in any material respect under the terms of such Material Contract and, to the Knowledge of Seller, no event has occurred that with the giving of notice or the passage of time or both would constitute a material breach or default of, or permit termination, modification, acceleration or cancellation of, any Material Contract or of any material right or obligation thereunder.

4.14. Legal Proceedings. Except as set forth on <u>Schedule 4.14</u> of the Seller Disclosure Schedule, (a) there are no pending Proceedings against the Acquired Company or to which the Acquired Interests or the Acquired Company's Assets are subject and, to the Knowledge of Seller, no such Proceeding has been threatened against the Acquired Company,

the Acquired Interests or the Acquired Company's Assets, and (b) there exist no unsatisfied orders, decrees, injunctions, and judgments which remain outstanding against the Acquired Company, the Acquired Interests or the Acquired Company's Assets. None of the Proceedings set forth on <u>Schedule 4.14(a)</u> has a Material Adverse Effect or (b) challenges or seeks to prevent, enjoin, alter or delay the transactions contemplated by this Agreement.

4.15. Permits. Except as set forth on Schedule 4.15 of the Seller Disclosure Schedule, the Acquired Company validly holds, and during the seven (7) year period preceding the Closing Date, has validly held when required, all material Permits necessary for the lawful conduct of the Acquired Company's Business as and when conducted, and the ownership and use of the Acquired Company's Assets (each such Permit, a "Material Permit"). Schedule 4.15 sets forth a list of all Material Permits. Each of the Material Permits is in full force and effect as of the Closing Date. The Acquired Company is, and during the seven (7) year period preceding the Closing Date, has been, in compliance in all material respects with each such Material Permit and no Proceeding is pending or, to Knowledge of Seller, threatened, to suspend, revoke, withdraw, adversely modify or limit any such Material Permit in a manner that has had or would reasonably be expected to have a material impact on the ability of the Acquired Company to use such Material Permit, conduct its operations in compliance with applicable Law or that would result in the termination, revocation, suspension, withdrawal or restriction of any such Material Permit or the imposition of any fine, penalty or other sanctions for violation of any requirements relating to any such Material Permit. Seller has made available to Buyer true and complete copies of all Material Permits.

4.16. Taxes. Except as set forth on <u>Schedule 4.16</u> of the Seller Disclosure Schedule:

(a) The Acquired Company has filed all material Tax Returns required to be filed by the Acquired Company when due (taking into account any applicable extensions), and all such Tax Returns are true, correct and complete in all material respects.

(b) The Acquired Company has timely paid in full all Taxes which are due and payable by the Acquired Company whether or not shown on any Tax Return.

(c) No audits or Proceedings are pending, have been conducted or, to Knowledge of Seller, have been threatened with respect to Taxes of the Acquired Company.

(d) The Acquired Company has not received any written (i) notice indicating an intent to open an audit or other review (ii) request for information related to Tax matters; or (iii) notice of deficiency or proposed adjustment for any amount of Tax proposed, asserted, or assessed by any Tax Authority against the Acquired Company.

(e) There are no Liens for Taxes (other than Permitted Liens) upon any of the Acquired Company's Assets.

(f) No written agreement or other document extending, waiving, or having the effect of extending, the period of assessment or collection of any Taxes payable by the Acquired Company has been executed or filed with the United States Internal Revenue Service or any other Taxing Authority that is currently in effect. The Acquired Company is not the beneficiary of any extension of time (other than an automatic extension of time not requiring the consent of the United States Internal Revenue Service or any other Taxing Authority) within which to file any Tax Return not previously filed.

(g) All material Taxes required to be withheld, collected or deposited by or with respect to the Acquired Company have been timely withheld, collected or deposited as the case may be, and to the extent required, have been paid to the relevant Taxing Authority.

(h) The Acquired Company is not nor has been a party to any "listed transaction" (as defined in Section 6707A(c)(2) of the Code) or any similar or corresponding transaction described under applicable state or local Law.

(i) No written claim has ever been made by a Taxing Authority to Acquired Company that it is or may be subject to Tax or required to file a Tax Return in a jurisdiction where Acquired Company does not currently file Tax Returns, which claim has not been resolved.

(j) The Acquired Company (i) is not a party to any income Tax allocation, sharing, or indemnification agreement (other than agreements or arrangements not primarily related to Taxes),
 (ii) is not nor has been a member of an affiliated group filing consolidated or combined Tax Returns (other than groups the common parent of which is or was Seller) or (iii) otherwise has no liability for the Taxes of any Person (other than the Acquired Company).

(k) The Acquired Company has not (i) elected to defer the payment of any "applicable employment taxes" (as defined in Section 2302(d)(1) of the Coronavirus Aid, Relief and Economic Security Act (the "CARES Act")) pursuant to Section 2302 of the CARES Act or (ii) claimed any "employee retention credit" pursuant to Section 2301 of the CARES Act.

(I) The Acquired Company has not applied for or obtained a loan pursuant to the Paycheck Protection Program.

(m) The Acquired Company is and has been at all times since formation treated as either a partnership or "disregarded entity" for U.S. federal income Tax purposes.

4.17. Employee Benefits; Employment and Labor Matters.

(a) The Acquired Company (i) does not have now, (ii) did not have during the five (5) year period preceding the Closing Date and (iii) to the Knowledge of Seller, since the Formation Date has not had, any employees or a Contract with any individual providing services as an independent contractor or as a consultant. All of the Site Employees are employed by one of Seller's other Affiliates (other than the Acquired Company or any other Acquired Company Group Member). No Site Employee is covered by any collective bargaining agreement or other labor union contract, and, to the Knowledge of Seller, there are no organizational campaigns, petitions or other unionization activities involving Site Employees seeking recognition of a collective bargaining unit. There are no strikes, material slowdowns or material work stoppages pending or, to the Knowledge of Seller, threatened involving the Site Employees. "Site Employees" means employees whose primary employment obligations are those related to the Assets and the Business of the Acquired Company.

(b) The Acquired Company does not sponsor, maintain, contribute to or have an obligation to contribute to, or has ever sponsored, maintained, contributed to or had an obligation to contribute to, any Plan, and the Acquired Company has no current or potential Liability or obligation under or with respect to any Plan. There does not now exist, nor do any circumstances exist that could reasonably be expected to result in, any Liability of the Acquired Company under (i) Title IV of ERISA, (ii) Sections 206(g), 302 or 303 of ERISA, or (iii) Sections 412, 430, 431, 436 or 4971 of the Code. There is no Lien pursuant to Section 430(k) of the Code or Sections 303(k) or 4068 of ERISA in favor of or enforceable by the Pension Benefit Guaranty Corporation or any other Governmental Authority with respect to any of its Affiliates for the benefit of the Site Employees has been administered in accordance with its terms and all applicable Laws, including ERISA and the Code, and all contributions required to be made with respect to any such Plans have been made or properly accrued.

4.18. Intellectual Property; IT.

Schedule 4.18(a) of the Seller Disclosure Schedule contains a true and complete list (a) of all Registered Intellectual Property. Except as set forth on Schedule 4.18(a) of the Seller Disclosure Schedule, the Acquired Company either owns or has a license or other right to use all material Owned Intellectual Property and other material Intellectual Property used in the Business of the Acquired Company as currently conducted. During the five (5) year period preceding the Closing Date, the Acquired Company has not infringed, nor has Seller or Acquired Company received a written claim that Acquired Company has infringed, on any Intellectual Property of any Person. There is no Proceeding pending or, to the Knowledge of Seller, threatened that challenges the validity, enforceability, registration, ownership or use of any Intellectual Property of the Acquired Company. All Intellectual Property of Acquired Company is valid, subsisting (except for any Intellectual Property designated as expired or abandoned on Schedule 4.18(a) of the Seller Disclosure Schedule), and enforceable. The Acquired Company has taken commercially reasonable measures to protect the confidentiality of the material trade secrets and confidential information owned by Acquired Company used in the Acquired Company's Business and of any third parties who have licensed material trade secrets and confidential information to the Acquired Company for use in the Acquired Company's Business.

(b) Except as set forth on <u>Schedule 4.18(b)</u> of the Seller Disclosure Schedule, the Intellectual Property that the Acquired Company owns, licenses or otherwise has a valid right to use, constitutes all the Intellectual Property necessary or used to operate and conduct the Acquired Company's Business as currently conducted.

(c) During the five (5) year period preceding the Closing Date, there has been no material failure or other material substandard performance of any computer systems of Acquired Company which has caused any material disruption to the Acquired Company's Business. The Acquired Company has taken commercially reasonable actions to protect the integrity and security of the computer systems and the software information stored thereon from unauthorized use, access, or modification by third parties, and to the Knowledge of Seller, no such third party has obtained unauthorized access to such computer or software systems in the five (5) year period preceding the Closing Date. During the five (5) year period preceding the Closing Date, there has been no actual or alleged security breach, or unauthorized use, access or intrusion, of any

information technology system or any personal information, payment card information, confidential information, trade secret or any other such information collected, maintained or stored by or on behalf of the Acquired Company (or any loss, destruction, compromise or unauthorized disclosure thereof).

Insurance. Set forth on Schedule 4.19 of the Seller Disclosure Schedule is 4.19. a complete and accurate list of all material property, general liability, third-party offsite pollution liability, automobile liability, workers' compensation and employers' liability, umbrella/excess liability and directors' and officers' liability insurance currently held by or on behalf of the Acquired Company. In connection with the operation of the Acquired Company's Business, Seller or the Acquired Company maintains insurance (including self-insurance) in such amounts and against such risks as (a) Seller or the Acquired Company has determined to be sufficient for the conduct of the Acquired Company's Business, (b) may be required under any Real Property Lease, Easement or Material Contract, and (c) as may be required by applicable Laws. Seller has made available to Buyer true and complete copies of all such insurance policies. Except as set forth on Schedule 4.19 of the Seller Disclosure Schedule, all such policies and contracts of insurance are in full force and effect, all premiums due and owing thereon by the Acquired Company (covering all periods up to and including the Closing Date) have been paid by Seller or the Acquired Company and the Acquired Company is otherwise in compliance with the terms and provisions of such policies and is not in default in any material respect under any such insurance policy. Except as set forth on Schedule 4.19 of the Seller Disclosure Schedule, there are no outstanding claims under any such insurance policy. There are no claims pending under any such policies as to which coverage has been guestioned, denied or disputed or in respect of which there is an outstanding reservation of rights. No written notice has been received by the Acquired Company that would reasonably be expected to be followed by a written notice of cancellation, alteration of coverage or non-renewal of any insurance policy set forth on Schedule 4.19 of the Seller Disclosure Schedule.

4.20. Affiliate Transactions. Except as set forth on <u>Schedule 4.20</u> of the Seller Disclosure Schedule, (a) none of Seller, any Affiliate of Seller or any director, manager or officer of Seller, any Affiliate of Seller or the Acquired Company: (i) is a party to any Contract with the Acquired Company, or (ii) owns or leases any material Asset, property or right which is used by the Acquired Company in the ownership or operation of the Acquired Company's Business or the Assets and (b) none of Seller, any Affiliate of Seller or any director, manager or officer of Seller, or any Affiliate of Seller (other than Acquired Company) is a party to a Contract which benefits, binds or proposes to bind the Acquired Company (collectively, (a) and (b), the "Affiliate Contracts"). All Affiliate Contracts will be satisfied, discharged or terminated prior to or at the Closing.

4.21. Bank Accounts. <u>Schedule 4.21</u> of the Seller Disclosure Schedule sets forth (a) the name of each financial institution with which the Acquired Company has borrowing or investment agreements, deposit or checking accounts or safe deposit boxes and (b) the types of those arrangements and accounts, including the names in which the accounts or boxes are held, the account or box numbers and the name of each Person authorized to draw thereon or have access thereto.

4.22. Powers of Attorney. <u>Schedule 4.22</u> sets forth a complete list of all powers of attorney issued by the Acquired Company that will remain in effect as of the Closing Date.

4.23. Brokers' Fees. Except as set forth on <u>Schedule 4.23</u> of the Seller Disclosure Schedule, no broker, investment banker, financial advisor or other Person is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of the Acquired Company for which Buyer or the Acquired Company shall have any responsibility.

4.24. Certain Payments. None of the Acquired Company or Seller (solely with respect to the Business of the Acquired Company Group), or, to the Knowledge of Seller, any existing or former officer, director, employee, agent or other Person acting on behalf of the Acquired Company or Seller (solely with respect to the Business of the Acquired Company Group) or any such Person has, directly or indirectly, during the five (5) year period preceding the Closing Date, made, given or agreed to give any money, gift, contribution, bribe, rebate, payoff, influence payment, kickback or similar benefit to any Person, private or public, regardless of form, whether in money, property or services (a) to obtain favorable treatment in securing business for the Acquired Company Group, the Seller (solely with respect to the Business of the Acquired Company Group) or the Business of the Acquired Company Group, (b) to pay for favorable treatment for business secured by the Acquired Company or Seller (solely with respect to the Business of the Acquired Company Group) or (c) to obtain special concessions or for special concessions already obtained, for or in respect of the Acquired Company, Seller (solely with respect to the Business of the Acquired Company Group) or the Business of the Acquired Company Group, in each case of clauses (a)-(c) in violation of any applicable Law.

ARTICLE V

REPRESENTATIONS AND WARRANTIES REGARDING OGC AND OCC

Subject to the provisions of this <u>Article V</u> and the other terms and conditions of this Agreement and the exceptions and matters set forth on the Seller Disclosure Schedule, Seller hereby represents and warrants to Buyer as follows:

5.1. Organization; Qualification. Each of OGC and OCC is a limited liability company duly formed, validly existing and in good standing under the laws of the state in which OGC or OCC, as applicable, was formed. To the Knowledge of Seller, each of OGC and OCC has all requisite organizational power and authority to own, lease and operate its properties and to carry on its business as it is now being conducted, and is duly qualified, registered or licensed to do business as a foreign entity and is in good standing in each jurisdiction in which the property owned, leased or operated by it or the nature of its business makes such qualification necessary as required by Law, except where the failure to be so duly qualified, registered or licensed and in good standing would not reasonably be expected to have a Material Adverse Effect. To the Knowledge of Seller, Seller has made available to Buyer true and complete copies of the Organizational Documents of OGC and OCC, as in effect on the Closing Date.

5.2. Non-Contravention; Approvals.

(a) Except as set forth on <u>Schedule 5.2(a)</u> of the Seller Disclosure Schedule, the execution, delivery and performance of the Transaction Documents by Seller, and the consummation of the transactions contemplated thereby by Seller, does not and will not: (a) result in any breach of any provision of the Organizational Documents of OGC or OCC; (b) except for any requisite Governmental Consents (assuming compliance with the matters referred to in <u>Section 4.3</u>), violate any Law to which any of OGC or OCC is subject or by which their respective Assets are bound; or (c) constitute (with or without the giving of notice or the passage of time or both) an event which would result in the creation of any Lien (other than Permitted Liens) on any Asset of OGC or OCC; except for such defaults or rights of termination, cancellation, amendment, or acceleration, violations or Liens, as would not reasonably be expected to have a Material Adverse Effect.

(b) Except as set forth on <u>Schedule 5.2(b)</u> of the Seller Disclosure Schedule, no Governmental Consent needs to be obtained by OGC or OCC in connection with the consummation by Seller of the transactions contemplated by the Transaction Documents to which Seller is, or will be, a party, other than such declarations, filings, registrations, notices, authorizations, consents or approvals which (i) are customarily made or obtained after the Closing or (ii) if not obtained or made, would not reasonably be expected to have a Material Adverse Effect.

5.3. Capitalization. Schedule 5.3 of the Seller Disclosure Schedule sets forth all of the issued and outstanding Equity Securities in each of OGC and OCC and the holders thereof. other than the Other Ohio Interests. Other than the Other Ohio Interests, Seller has not authorized issuance of any Equity Securities in OGC or OCC other than as set forth on Schedule 5.3 of the Seller Disclosure Schedule. To the Knowledge of Seller, the Non-Op Interests have been duly authorized, validly issued and are fully paid and, subject to the Laws of the State of Delaware, nonassessable and were not issued in violation of any purchase option, call option, right of first refusal, preemptive right, or other similar right. There are no outstanding or authorized equity appreciation, phantom stock, profit participation, preemptive rights, registration rights, approval rights, proxies, rights of first refusal, or similar rights affecting the Non-Op Interests granted by Seller, Seller's Affiliate or the Acquired Company. The Non-Op Interests are free and clear of all Liens, except for any Corporate Encumbrances. Neither Seller, the Acquired Company nor any other Person owns any other interests of OGC or OCC or has any option, warrant, equity appreciation right, convertible security or other contractual right or security (whether or not currently exercisable) to acquire any additional interests of OGC or OCC except as set forth in the Organizational Documents of OGC and OCC, as applicable.

5.4. Absence of Changes. To the Knowledge of Seller, since the Balance Sheet Date and through the Closing Date, except as set forth in <u>Schedule 5.4</u> of the Seller Disclosure Schedule, (a) the Business of OGC and OCC has been conducted in the ordinary course of business in all material respects, and (b) there has not been any change, event, occurrence or development relating to OGC or OCC that has had or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

5.5. Legal Proceedings. Except as set forth on <u>Schedule 5.5</u> of the Seller Disclosure Schedule, to the Knowledge of Seller, there are no pending material Proceedings against OGC

or OCC or to which the Non-Op Interests are subject and, to the Knowledge of Seller, no such Proceeding has been threatened against OGC or OCC or the Non-Op Interests.

5.6. Taxes. OGC and OCC, respectively, have each filed all applicable material federal, state and local income Tax Returns required to be filed by each such company when due (taking into account any applicable extensions), and all such Tax Returns are true, correct and complete in all material respects.

ARTICLE VI

Representations and Warranties of Buyer

Buyer hereby represents and warrants to Seller as follows:

6.1. Organization. Buyer is a legal entity duly formed, validly existing and in good standing under the laws of Delaware and has all requisite organizational power and authority to enter into the Transaction Documents and to perform its obligations hereunder and thereunder.

6.2. Authority; Enforceability.

(a) Buyer has the requisite power and authority to execute and deliver the Transaction Documents to which it is, or will be, a party, to carry out its obligations hereunder and to consummate the transactions contemplated thereby. The execution, delivery and performance by Buyer of the Transaction Documents to which it is, or will be, a party, and the consummation by it of the transactions contemplated thereby, have been duly and validly authorized by Buyer, and no other entity proceedings on the part of Buyer are necessary to authorize the Transaction Documents to which it is, or will be, a party or to consummate the transactions contemplated by the Transaction Documents to which it is, or will be, a party.

(b) The Transaction Documents to which Buyer is, or will be, a party have been (or will be, when executed and delivered at the Closing) duly executed and delivered by Buyer, and, assuming the due authorization, execution and delivery by the other parties thereto, each Transaction Document to which Buyer is, or will be, a party constitutes (or will constitute, when executed and delivered at the Closing) the valid and binding agreement of Buyer, enforceable against Buyer in accordance with its terms, except as such enforceability may be limited by Creditors' Rights.

6.3. Non-Contravention. Except as set forth on <u>Schedule 6.3</u> of the Buyer Disclosure Schedule, the execution, delivery and performance of the Transaction Documents to which Buyer is, or will be, a party by Buyer and the consummation by Buyer of the transactions contemplated thereby does not and will not: (a) result in any breach of any provision of the Organizational Documents of Buyer; (b) require authorization or consent of, approval by, filing with or notice under, conflict with, constitute a default (or an event that with notice or passage of time or both would give rise to a default) under, or give rise to any right of termination, cancellation, amendment or acceleration (with or without the giving of notice, or the passage of time or both) under any of the terms, conditions or provisions of any Contract to which Buyer is a party; or (c) except for any requisite Governmental Consents, violate any Law to which Buyer is subject except, in the cases of <u>clauses (b)</u> and (c), for such defaults or rights of termination, cancellation, amendment, or acceleration or violations as would not reasonably be

expected to prevent or materially delay the consummation of the transactions contemplated by the Transaction Documents to which Buyer is, or will be, a party or to materially impair Buyer's ability to perform its obligations under the Transaction Documents to which it is, or will be, a party.

6.4. Governmental Approvals. Except as set forth on <u>Schedule 6.4</u> of the Buyer Disclosure Schedule, no declaration, filing or registration with, or notice to, or authorization, consent or approval of, any Governmental Authority is necessary for the consummation by Buyer of the transactions contemplated by the Transaction Documents to which it is a party, declarations, filings, registrations, notices, authorizations, consents or approvals which are customarily made or obtained after the Closing or if not obtained or made, would not reasonably be expected to prevent or materially delay the consummation of the transactions contemplated by the Transaction Buyer is, or will be, a party or to materially impair Buyer's ability to perform its obligations under the Transaction Documents to which it is, or will be, a party.

6.5. HSR Act Approval. Buyer represents that the so-called "size of transaction" jurisdictional test under the HSR Act is not satisfied. Capitalized or defined terms in this <u>Section</u> <u>6.5</u> have the meaning set forth in the HSR Act and regulations.

6.6. Legal Proceedings. There are no Proceedings pending or, to the Knowledge of Buyer, threatened against Buyer that challenge or seek to prevent, enjoin, alter or delay the transactions contemplated by the Transaction Documents or any of the other documents and instruments to be executed and delivered by Buyer in connection therewith, except such Proceedings as would not reasonably be expected to prevent or materially delay the consummation of the transactions contemplated by the Transaction Documents to which Buyer is, or will be, a party or to impair Buyer's ability to perform its obligations under the Transaction Documents to which it is, or will be, a party.

6.7. Matters Relating to Acquisition of the Acquired Interests.

Buyer has such knowledge and experience in financial and business matters so as (a) to be capable of evaluating the merits and risks of its investment in the Acquired Interests and is capable of bearing the economic risk of such investment. Buyer is an "accredited investor" as that term is defined in Rule 501 of Regulation D (without regard to Rule 501(a)(4)) promulgated under the Securities Act. Buyer is acquiring the Acquired Interests for investment for its own account and not with a view toward or for sale in connection with any distribution thereof, or with any present intention of distributing or selling the Acquired Interests. Buyer does not have any Contract or arrangement with any Person to sell, transfer or grant participations to such Person or to any third Person, with respect to the Acquired Interests in violation of the Securities Act. Buyer acknowledges and understands that (i) the acquisition of the Acquired Interests has not been registered under the Securities Act in reliance on an exemption therefrom and (ii) that the Acquired Interests will, upon the consummation of the transactions contemplated herein, be characterized as "restricted securities" under state and federal securities laws. Buyer agrees that the Acquired Interests may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of except pursuant to an effective registration statement under the Securities Act or pursuant to an available exemption from the registration requirements of the Securities Act, and in compliance with other applicable state and federal securities laws.

(b) Buyer has undertaken such investigation as it has deemed necessary to enable it to make an informed and intelligent decision with respect to the execution, delivery and performance of this Agreement and the acquisition of the Acquired Interests. Buyer has had an opportunity to ask questions and receive answers from Seller regarding the terms and conditions of the offering of the Acquired Interests and the Business, Assets, prospects, and financial condition of the Acquired Company (to the extent Seller possessed such information).

(c) Buyer agrees that at the Closing, it shall accept the Acquired Company based upon Buyer's own inspection, examination and determination with respect thereto as to all matters, and without reliance upon any express or implied representations or warranties of any nature, whether in writing, orally or otherwise, made by or on behalf of or imputed to Seller or the Acquired Company, except as expressly set forth in this Agreement.

6.8. Bankruptcy. There are no bankruptcy, insolvency, reorganization or receivership proceedings pending against, being contemplated by, or, to Buyer's Knowledge, threatened against Buyer. Buyer is not entering into the transactions contemplated by this Agreement with the actual intent to hinder, delay, or defraud either present or future creditors. At and immediately after the Closing, Buyer (a) will be solvent (in that both the fair value of its assets will not be less than the sum of its debts and the present fair saleable value of its assets will not be less than the amount required to pay its probable Liability on its debts as they become absolute and matured), (b) will have adequate capital and liquidity with which to engage in its business, and (c) will not have incurred and does not plan to incur debts beyond its ability to pay as they become absolute and matured.

6.9. Financing. Buyer has, and as of the Closing, Buyer will have, sufficient cash, available lines of credit or other sources of immediately available funds to enable it to fund and pay the Purchase Price and any other amounts to be paid by Buyer hereunder or under the Transaction Documents.

6.10. Brokers' Fees. Except as set forth on <u>Schedule 6.10</u> of the Buyer Disclosure Schedule, no broker, investment banker, financial advisor or other Person is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of Buyer for which, Seller or the Acquired Company shall have any responsibility.

6.11. Foreign Person. Buyer is not a foreign person as that term is used in Section 721 of the Defense Production Act of 1950, as amended, 50 U.S.C. § 4565, and the regulations promulgated thereunder.

6.12. No Reliance; Disclaimer.

(a) In making the decision to enter into this Agreement and consummate the transactions contemplated hereby, Buyer has relied solely upon the representations and warranties of Seller set forth in <u>Article III</u>, <u>Article IV</u> and <u>Article V</u> of this Agreement. Except for the representations and warranties of Seller set forth in <u>Article IV</u> and <u>Article III</u>, <u>Article V</u> of this Agreement.

Agreement, neither the Acquired Company nor Seller, nor any of their respective Affiliates or any of their respective stockholders, trustees, members, partners, equityholders, fiduciaries or Representatives, or any other Person has made or is making, and Buyer has not relied upon, any other representation or warranty of any kind or nature whatsoever, oral or written, express or implied, with respect to the Acquired Company (including, without limitation, regarding the probable success or profitability of the Acquired Company), Seller, their respective Affiliates, the Acquired Interests or OGC or OCC or this Agreement or the transactions contemplated herein.

EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES OF SELLER (b)SET FORTH IN ARTICLE III, ARTICLE IV AND ARTICLE V OF THIS AGREEMENT, BUYER DISCLAIMS ANY OTHER REPRESENTATIONS OR WARRANTIES, WHETHER MADE BY THE ACQUIRED COMPANY GROUP, SELLER OR ANY OF THEIR RESPECTIVE AFFILIATES, STOCKHOLDERS, PARTNERS, MEMBERS, OFFICERS, MANAGERS, EMPLOYEES, DIRECTORS, AGENTS OR REPRESENTATIVES (INCLUDING WITH RESPECT TO MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE, THE NATURE OR EXTENT OF ANY LIABILITIES, THE BUSINESS OR FINANCIAL PROSPECTS, OR THE EFFECTIVENESS OR SUCCESS OF ANY OPERATIONS OF THE ACQUIRED COMPANY GROUP, THE DISTRIBUTION OF, OR ANY PERSON'S RELIANCE ON, ANY INFORMATION, DISCLOSURE OR OTHER DOCUMENT OR OTHER MATERIAL MADE AVAILABLE TO ANY PARTY IN ANY DATA ROOM, ELECTRONIC DATA ROOM, MANAGEMENT PRESENTATION OR IN ANY OTHER FORM IN EXPECTATION OF, OR IN CONNECTION WITH, THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT).

6.13. Seller Insurance Policies. As of the Closing Date, Buyer does not have any potential pending claims, nor does it expect or intend as of the Closing Date to bring any claims, against the Seller Insurance Policies set forth on <u>Schedule 7.18(a)</u>.

ARTICLE VII

Covenants of the Parties

7.1. Intentionally Deleted.

7.2. Intentionally Deleted.

7.3. Consents and Approvals. Subject to the terms and conditions of this Agreement and except as otherwise expressly provided in this Agreement, from and after the Closing, each of the Parties will use commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, and assist and cooperate with the other in doing, all things necessary, proper or advisable under applicable Law to make effective the transactions contemplated by this Agreement and the other Transaction Documents as promptly as practicable, including obtaining and maintaining all approvals, consents, registrations, permits, authorizations and other confirmations required to be obtained from any Governmental Authority or other third party that are necessary, proper or advisable to make effective the transactions contemplated by this Agreement and the other Transaction Documents.

7.4. Continuing Indemnification.

Buyer shall, and shall cause its Subsidiaries (including after the Closing, the (a) Acquired Company Group) to, ensure that all rights to exculpation, indemnification and advancement of expenses existing immediately prior to the Closing Date in favor of the directors, managers, and officers of the Acquired Company Group Members ("D&O Indemnified Persons") as provided in the respective governing documents and indemnification agreements, including Organizational Documents, to which the Acquired Company Group Member is a party, shall survive the Closing and shall continue in full force and effect for a period of six (6) years from the Closing Date. During such time period, the indemnification agreements and the provisions with respect to indemnification and limitations on liability set forth in such Organizational Documents pertaining to pre-Closing period shall not be amended, repealed or otherwise modified (unless required by Law) in a manner that is adverse to any such Person; provided that all rights to indemnification and advancement of expenses in respect of any Proceeding arising out of or related to matters existing or occurring at or prior to the Closing and asserted or made within such sixyear period (each a "D&O Claim") shall continue until the final disposition of such Proceeding. For the avoidance of doubt, nothing in this Section 7.4 shall prevent Buyer, the Acquired Company Group, their Affiliates or successors/assigns from amending, repealing or otherwise modifying, at their sole direction, the indemnification agreements and the provisions with respect to indemnification and limitations on liability set forth in the Organizational Documents of the Acquired Company Group Members applicable to the directors, managers, and officers of the Acquired Company Group Members who are appointed, assigned or designated during the period after the Closing Date. The obligations of Buyer and the Acquired Company under this Section 7.4 shall survive the Closing and shall not be terminated or modified in such a manner as to affect adversely and materially any D&O Indemnified Person to whom this Section 7.4 applies without the consent (which consent shall not be unreasonably withheld or delayed) of such affected D&O Indemnified Person (it being expressly agreed that the D&O Indemnified Persons to whom this Section 7.4 applies shall be third party beneficiaries of this Section 7.4, each of whom may enforce the provisions of this Section 7.4).

(b) In the event that Buyer, the Acquired Company or any of their respective successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving Person in such consolidation or merger or (ii) transfers all or substantially all of its properties and assets of the Acquired Company to any Person, then, and in either such case, proper provision shall be made so that the successors and assigns of Buyer or the Acquired Company, as the case may be, shall assume all of the obligations set forth in this <u>Section 7.4</u>. The agreements and covenants contained herein shall not be deemed to be exclusive of any other rights to which any D&O Indemnified Person is entitled, whether pursuant to Law, Contract or otherwise. Nothing in this Agreement is intended to, shall be construed to or shall release, waive or impair any rights to directors' and officers' insurance claims under any policy that is or has been in existence with respect to the Company or its officers, directors and employees, it being understood and agreed that the rights provided for in this <u>Section 7.4</u> are not prior to, or in substitution for, any such claims under any such policies.

7.5. Retention of Books and Records.

Subject to the confidentiality obligations contained in the OGC LLC Agreement (a) and the OCC LLC Agreement as of the Closing Date, Seller and its Affiliates shall have the right to retain, at their sole cost and expense, copies of (i) all books and records and all Tax Returns and other information and documents (A) relating to Tax matters of the Acquired Company Group, in each case, relating to periods ending on or prior to the Closing Date (B) as required by any legal or regulatory authority, including any applicable Law or regulatory request and (C)(i) as may be necessary for Seller and its Affiliates to perform their respective obligations pursuant to this Agreement and the other Transaction Documents, in each case subject to compliance in all material respects with applicable Laws, (ii) all data room materials, and (iii) all books and records prepared in connection with the transactions contemplated herein, including any books and records that may be relevant in connection with the defense of disputes arising under this Agreement or financial information and all other accounting books and records prepared or used in connection with the preparation of financial statements of any of the Acquired Company Group. To the extent not already in the possession of the Acquired Company Group as of the Closing, within thirty (30) days after the Closing, Seller shall deliver to Buyer, at Buyer's cost, all books, records, Contracts, agreements, leases, permits and any other documentation that is necessary for the operation of the Acquired Company Group that are in the possession of Seller or any Affiliate of Seller.

From and after the Closing, Buyer will use its commercially reasonable efforts to (b) make the books, records and other documents pertaining to the Acquired Company Group's operations during the pre-Closing time period available for examination and copying by Seller or its Representatives, at Seller's expense, as may be reasonably necessary in connection with any pre-Closing insurance matters, filings with any Governmental Authority, the preparation and filing of any Tax Returns or as otherwise required to the extent reasonably necessary to implement the provisions of this Agreement, and/or as required or requested by Seller or its Affiliates in connection with the Excluded Liability, including the defense thereof, in each case upon reasonable notice to Buyer to the extent not retained by Seller in accordance with Section 7.5(a); provided that such access will not unreasonably interfere with the business or operations of Buyer. the Acquired Company Group's Business or their employees. Notwithstanding anything to the contrary herein, neither Buyer nor any of its Affiliates shall be required to disclose to Seller or any of its Representatives any information to the extent: (A) doing so would violate any Contract or Law to which Buyer, the Acquired Company Group or any of their Affiliates are a party or is subject or which they believe in good faith would reasonably be expected to result in a loss of the ability to successfully assert a claim of attorney-client privilege or (B) if Buyer or any of its Affiliates, on the one hand, and Seller or its Affiliate, on the other hand, are adverse parties in a litigation and such information is reasonably pertinent thereto.

7.6. Expenses. All costs and expenses (a) incurred by Seller and/or the Acquired Company in connection with the Transaction Documents and the transactions contemplated thereby, including the Transaction Expenses, shall be paid by Seller and (b) incurred by Buyer in connection with the Transaction Documents and the transactions contemplated thereby shall be paid by Buyer, including as contemplated in <u>Section 7.11</u>; provided, however, that if any action at law or equity is necessary to enforce or interpret the terms of the Transaction Documents, the prevailing Party shall be entitled to reasonable attorneys' fees and expenses in addition to any other relief to which such Party may be entitled.

7.7. Further Assurances. Subject to the terms and conditions of this Agreement, each Party shall use all commercially reasonable efforts to take, or cause to be taken, all action, and to do, or cause to be done, all things necessary, proper or advisable under applicable Law to consummate the transactions contemplated by the Transaction Documents; provided, however, that no such act or document shall increase a Party's liabilities, costs or obligations, or decrease its rights or benefits, under this Agreement or any other Transaction Document.

7.8. Public Statements. Except as otherwise provided in this Section 7.8, neither Buyer nor Seller shall issue, and each of them shall not permit any Affiliate to issue, any press release or other similar public announcement regarding the existence of this Agreement, the contents hereof or the transactions contemplated hereby without the prior written consent of the other Party (which consent shall not be unreasonably withheld, conditioned or delayed); provided, however, that the foregoing shall not restrict disclosures (a) to the extent necessary for a Party to perform this Agreement (including disclosures to Governmental Authorities), or (b) to the extent required, in the reasonable opinion of legal counsel, by applicable securities or other Laws or regulations or the applicable rules of any stock exchange having jurisdiction over the Parties or their respective Affiliates. Notwithstanding anything to the contrary contained herein, each Party and their respective Affiliates may make statements that are consistent with statements made in previous press releases, public disclosures or public statements made in compliance with this Section 7.8 without having to seek consent from the other Party. Notwithstanding any provision to the contrary contained in this Agreement, the provisions of this Section 7.8 will survive the termination of this Agreement pursuant to Article X and the Closing.

7.9. Transfer Taxes. Buyer shall be solely responsible for the timely payment of all state and local transfer, sales, use, stamp, registration or other similar Taxes resulting from the transactions contemplated by this Agreement ("Transfer Taxes") and shall indemnify, defend and hold harmless Seller Indemnified Parties against any Transfer Taxes. Buyer shall prepare and file when due all necessary documentation and Tax Returns with respect to such Transfer Taxes. Buyer and Seller shall cooperate in good faith to minimize, to the extent permissible under applicable Law, the amount of any such Transfer Taxes.

7.10. Tax Matters.

(a) <u>Filing of Tax Returns</u>. Following the Closing, Buyer shall prepare or cause to be prepared all Tax Returns required to be filed by the Acquired Company after the Closing Date. All such Tax Returns shall be prepared on a basis consistent with past practice except to the extent otherwise required by applicable Law. Buyer will provide Seller with copies of all such Tax Returns no later than five (5) days prior to the due date for filing thereof (including applicable extensions) for Seller's review. Buyer shall incorporate on any such Tax Return any reasonable comments provided by Seller. Buyer shall cause such Tax Returns to be timely filed and will provide a copy thereof to Buyer. Buyer shall, or shall cause the Acquired Company to, pay or cause to be paid all Taxes owed with respect to such Tax Returns to the appropriate Governmental Authority.

(b) <u>Cooperation</u>. Buyer and Seller shall cooperate fully, as and to the extent reasonably requested by the other Party, in connection with the filing of Tax Returns and any inquiry, claim,

assessment, audit, litigation or other proceeding (each a "Tax Proceeding") with respect to Taxes imposed on or with respect to the Assets, operations or activities of the Acquired Company. Such cooperation shall include the retention and (upon the other Party's request) the provision of records and information which are reasonably relevant to any such Tax Returns or Tax Proceedings and making representatives and agents available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder.

Post-Closing Actions. Neither Buyer nor any Affiliate of Buyer shall (or shall (c) cause or permit any member of the Acquired Company Group to) (i) make or change any election, amend, refile or otherwise modify any Tax Return, initiate a voluntary disclosure or otherwise voluntarily approach a taxing authority regarding any Taxes or Tax Returns, file any ruling or request with any Governmental Authority that relates to Taxes, enter into any closing agreement, settle any Tax claim or assessment, extend or waive the applicable statute of limitations or surrender any right to claim a refund of Taxes, in each case relating in whole or in part to the Acquired Company Group or any member thereof with respect to any tax period ending on or prior to the Closing Date, or (ii) take any action relating to Taxes that is outside the ordinary course of business that could create a Tax Liability for a taxable period (or portion thereof) ending on or prior to the Closing Date for which Seller (or its direct or indirect owners) could be liable, in the case of either of the foregoing clauses (i) and (ii), without the prior written consent of the Seller (which will not be unreasonably conditioned, withheld or delayed); provided however, nothing in this section or this Agreement to the contrary shall override or restrict any delegations of rights or authority related to Tax matters to Buyer or Buyer's Affiliate under the OGC LLC Agreement or the OCC LLC Agreement as in effect at the time of Closing. The Parties further agree to use their commercially reasonable efforts to have the boards of OGC and OCC elect a reasonable proration method pursuant to Sections 9.5 of the respective OGC LLC Agreement and OOC LLC Agreement.

(d) <u>Bulk Sales</u>. Each Party hereto hereby waives compliance by the other Party with the provisions of the bulk transfer Laws of any jurisdiction in connection with the transactions contemplated by this Agreement.

(e) <u>Tax Liability</u>. In the event of any breach by Seller of the representation and warranty in <u>Section 5.6</u>, to the extent Buyer makes a claim under the R&W Policy for a Loss as a result of such breach, Seller shall reimburse Buyer for an amount equal to the product of (i) the percentage set forth on <u>Schedule 1.1(a)</u> applicable to OCC or OGC, based on which such entity incurred the Loss, multiplied by (ii) the amount of the Loss arising from the breach of <u>Section 5.6</u> for which Buyer is seeking recovery under the R&W Policy; provided, however, in no event shall Seller be obligated to reimburse any amounts under this <u>Section 7.10(e)</u> in excess of \$1,250,000 and in no event shall Seller be obligated to reimburse any amounts under this <u>Section 7.10(e)</u> in the event that the JV Initial General Retention (as defined in the R&W Policy), as the same may be reduced pursuant to the R&W Policy, has been satisfied.

7.11. R&W Policy. The Parties acknowledge and agree that, as of or prior to the Closing Date, Buyer has procured the R&W Policy in the form attached hereto as Exhibit B. Buyer shall timely pay or cause to be paid, all costs and expenses related to the R&W Policy, including the total premium, underwriting costs, taxes, brokerage commission, and other fees and expenses of such policy. From and after the Closing and issuance of the R&W Policy,

Buyer shall not amend, modify, or otherwise change, terminate, or waive any provision of the R&W Policy in a manner adverse to Seller, any of its Affiliates or any of their respective Representatives, including any rights of the insurer to subrogate or seek recovery from Seller, any of Seller's Affiliates or any of Seller's Representatives.

7.12. Affiliate Services. Seller and its Affiliates shall have no obligation to provide any services with respect to the Acquired Company Group or their respective Assets from and after the Closing and, unless otherwise agreed to in writing by Seller and Buyer in the Transition Services Agreement, Seller shall, and shall cause the Acquired Company Group to, terminate all Affiliate Contracts prior to or effective as of the Closing Date (with no further Liability to the Acquired Company Group) and provide to Buyer evidence of the termination of all such transactions and Contracts.

7.13. Employee Matters.

Beginning on the date of this Agreement and continuing throughout the term of the (a) Transition Services Agreement, Seller shall make available to Buyer each of the Subject Employees to discuss potential employment with Buyer or an Affiliate of Buyer. A Subject Employee who receives an offer of employment from Buyer or an Affiliate of Buyer pursuant to this Section 7.13 is referred to herein as an "Offered Employee". For purposes of clarification and avoidance of doubt, Buyer has no obligation to hire any Subject Employee. The offers of employment pursuant to this Section 7.13 may be made by Buyer or an Affiliate of Buyer, in its sole and absolute discretion, commencing, as determined by Buyer, no earlier than the Closing Date and no later than the first day immediately following the last day of the Transition Services Agreement, except regarding any Offered Employee who is on an approved leave of absence, in which case the offer shall be for employment commencing as of the date on which such Offered Employee is eligible to return to work (provided that such return occurs within one hundred eighty (180) days after the intended employment commencement date as set forth in the employment offer or such later time as may be required by applicable Law). Seller shall not discourage any Subject Employee listed on Schedule 7.13 of the Seller Disclosure Schedule from accepting employment with Buyer or an Affiliate of Buyer. Seller shall (i) accept the resignation of each Offered Employee who accepts an offer of employment from Buyer or an Affiliate of Buyer immediately preceding such employee's Hire Date (as defined below) and (ii) waive any restrictive covenants or other obligations to which any Offered Employee may be subject pursuant to an agreement with Seller or the Acquired Company that would interfere with such employee's employment with Buyer or an Affiliate of Buyer. An Offered Employee who accepts an offer of employment from Buyer and timely commences employment with Buyer shall be a "Hired Employee" and the date on which such Hired Employee commences employment with Buyer shall be the "Hire Date."

(b) Subject to the terms of the Transition Services Agreement, Seller shall be responsible for payment of all obligations required or committed to the Hired Employees with respect to periods of employment with Seller and its Affiliates, including salaries, wages, payroll taxes, retirement, vacation pay, and any other obligations and expenses of any kind arising out of the employment by, or termination from the employment of, Seller or its Affiliates of such employees. Buyer shall be responsible for payment of all obligations required or committed to the Hired Employees with respect to periods of employment with Buyer and its Affiliates, including

salaries, wages, payroll taxes, retirement, vacation pay, and any other obligations and expenses of any kind arising out of the employment by, or termination from the employment of, Buyer or its Affiliates of such Hired Employees.

(c) The provisions of this <u>Section 7.13</u> are solely for the benefit of the Parties and nothing herein express or implied shall confer upon any employee, Subject Employee, Offered Employee, Hired Employee or any legal representative or beneficiary thereof any rights or remedies, including any right to employment, or continued employment for any specified period, of any nature or kind whatsoever.

7.14. Confidentiality.

(a) The Parties acknowledge that the Confidentiality Agreement and the Parties' obligations thereunder shall terminate at Closing.

(b) From and after the Closing, Seller shall, and shall cause its respective Affiliates and its and their respective Representatives to, keep confidential and not use for any purpose or disclose to any Person all confidential and proprietary or nonpublic information whether written, oral, visual or in electronic format, and together with analyses, compilations, notes, studies or other documents, whether prepared by or on behalf of Seller or its Representatives, which contain or otherwise reflect such information relating to Buyer, the Acquired Company Group Members and their respective Assets and Businesses ("Confidential Information").

(c) Notwithstanding the foregoing, term "Confidential Information" will not include any information if such information is known, or becomes known, to the public without wrongful disclosure by Seller or its attorneys, accountants, consultants, other professionals, Affiliates, or partners, or such information is required to be disclosed by the applicable Laws; provided, however, that if Seller's legal counsel determines that such disclosure is required, in the reasonable opinion of legal counsel, Seller shall as soon as reasonably practicable, to the extent permissible under applicable Law, notify Buyer in writing prior to disclosure so that Buyer can seek a protective order or other reasonable protections to avoid unnecessary disclosure at Buyer's sole cost and expense; and provided, further that Seller may disclose such information (i) to its respective attorneys, accountants, consultants and other professionals to the extent necessary to obtain their services; (ii) to any Affiliate or member or lender of Seller, provided that such Persons also agree to keep such information confidential and to protect such information as required by this Agreement; or (iii) to enforce this Agreement or any Transaction Document or perform any obligation hereunder or thereunder.

7.15. Use of Name and Seller's Marks.

(a) With respect to the phase-out of the use of the names, business names, trade names, brand names, registered and unregistered trademarks, logos, service marks or any other similar Intellectual Property (or any variation or derivative of such names, business names, trade names, brand names, registered and unregistered trademarks, logos, service marks or any other similar Intellectual Property confusingly similar thereto) of Seller or any of its Affiliates, including the use of the name or trademarks of "Summit" (collectively, the "Seller's Marks"), Buyer agrees as follows: (i) as soon as practicable following the Closing, but in no event later than thirty (30) days

after Closing, Buyer shall, and after the Closing shall cause the Acquired Company to, change the name of the Acquired Company to a name that does not include any of Seller's Marks or comprise any colorable imitations thereof; and (ii) as soon as practicable following the Closing, but in no event later than ninety (90) days after Closing, Buyer shall, and after the Closing shall cause the Acquired Company to, cease and permanently discontinue any and all uses of any of Seller's Marks and any colorable imitations thereof, and remove or cover all Seller's Marks from or on, or destroy, any publications, signage, corporate letterhead, invoices, stationery, business cards, marketing materials, website content or other materials in the possession or under the control of the Acquired Company bearing any of Seller's Marks.

(b) In no event shall Buyer or any of its Affiliates (including, after Closing, the Acquired Company) use any of Seller's Marks after Closing in any manner or for any purpose different from the use of such Seller's Marks by the Acquired Company preceding the Closing, and none of them shall affix or include any of Seller's Marks, or any colorable imitations thereof, on any publications, signage, corporate letterhead, invoices, stationery, business cards, marketing materials, website content or other materials that are created or produced after the Closing.

(c) Buyer expressly acknowledges and confirms that Seller is not transferring or assigning, and Buyer shall not receive, any right, title or interest in or to the Seller's Marks, except the limited right to use for the sole purpose of permitting Buyer to complete the phase-out in strict compliance with this <u>Section 7.15</u>.

(d) Notwithstanding anything to the contrary in this Agreement, neither Buyer nor any of its Affiliates (including, after the Closing, the Acquired Company) shall be deemed to have violated this Section 7.15, even after the periods provided above, solely by reason of: (i) their use of equipment and other similar articles used in the Business as of the Closing, notwithstanding that they may bear one or more of the Seller's Marks (provided that it is not reasonably practicable to remove or cover such Seller's Mark in accordance with Section 7.15a)); (ii) the appearance of the Seller Marks on any manuals, work sheets, operating procedures, other written or electronic data, materials or assets (including computer source code) that are used for internal purposes only in connection with the Business; (iii) the appearance of the Seller's Marks in or on any third party's publications, marketing materials, brochures or instruction sheets that were distributed prior to the Closing and that are not used by Buyer and its Affiliates (including, after the Closing, the Acquired Company) after such ninety (90) day period contemplated in Section 7.15(a); or (iv) the use by Buyer and its Affiliates (including, after the Closing, the Acquired Company) of any Seller's Marks in a non-disparaging and non-trademark manner in textual sentences that is factually accurate and non-prominent for purposes of conveying to customers or the general public that the Acquired Company's Business is no longer affiliated with Seller or to reference historical details concerning or make historical reference to the Acquired Company's Business; provided, however, Buyer, for itself and its Affiliates (including, after Closing, the Acquired Company), agrees that, after the Closing Date, Buyer and its Affiliates (including, after Closing, the Acquired Company) will not do business or represent themselves as having any affiliation or business relationship with Seller or any Affiliates of Seller, except pursuant to any separate agreement entered into, or assumed by, Buyer or its Affiliates on one hand, and Seller or its Affiliates, on the other hand.

7.16. Cash Distributions. Prior to the Closing, Seller, in its discretion, may cause the Acquired Company Group to distribute any or all Cash and Cash Equivalents of the

Acquired Company Group to Seller and its Affiliates. For the avoidance of doubt, any Cash and Cash Equivalents remaining with the Acquired Company Group as of Closing shall be taken into account for purposes of determining the Closing Date Cash and included in the Final Purchase Price in accordance with <u>Section 2.3</u>.

7.17. Bonds. The Parties acknowledge and agree that Seller has the right to terminate any of the bonds, letters of credit and guarantees provided by it or its Affiliates for the benefit of the Acquired Company Group that are listed on <u>Schedule 7.17</u> of the Seller Disclosure Schedule, and Buyer shall be responsible, at Buyer's sole cost and expense, to cause, after the Closing, the same to be released (and evidence thereof reasonably acceptable to Seller to be provided by Buyer to Seller at Closing) and, if applicable, replaced, after the Closing. From and after the Closing, to the extent such obligations have not been released, until all of the obligations listed on <u>Schedule 7.17</u> of the Seller Disclosure Schedules have been so released, Buyer shall (a) indemnify Seller and its Affiliates from and against any losses incurred by any of them arising out of or with respect to any of such obligations, and (b) not amend, modify or renew any Contract then subject to, or guaranteed or otherwise supported by, any such obligation without the consent of Seller.

7.18. Insurance Matters.

(a) Buyer acknowledges and agrees that, except as otherwise expressly provided for in the Transition Services Agreement or as set forth on <u>Schedule 7.18(a)</u>, from and after the Closing Date, (i) Seller or its Affiliates may terminate coverage with respect to the employees, the Business, properties and assets of the Acquired Company Group under any insurance policies maintained by any Acquired Company Group Member or with respect to which any Acquired Company Group Member or otherwise the beneficiary of coverage (the "Seller Insurance Policies"), (ii) none of the properties or assets of the Acquired Company Group will be covered under Seller Insurance Policies following the Closing Date, and (iii) Buyer shall become solely responsible for procuring, maintaining and paying for all insurance policies with respect to post-Closing issuance coverage for the properties and assets of the Acquired Company Group.

(b) Buyer and its Affiliates shall be entitled to receive and retain any and all amounts paid to insured Persons pursuant to any Seller Insurance Policy in respect of any insurance claim relating to the conduct of the business of the Acquired Company Group prior to the Closing Date.

(c) Seller and Seller's insurers waive their right of subrogation against Buyer and Buyer's Affiliates, and Buyer and Buyer's insurers waive their right of subrogation against Seller and Seller's Affiliates.

7.19. Intentionally Deleted.

7.20. Independent Investigation. BUYER ACKNOWLEDGES AND AGREES THAT, EXCEPT FOR FRAUD OR AS OTHERWISE EXPRESSLY PROVIDED HEREIN OR IN THE OTHER TRANSACTION DOCUMENTS, (A) IN ACQUIRING THE ACQUIRED INTERESTS, BUYER IS ACQUIRING THE ASSETS ON AN "AS IS, WHERE IS, WITH ALL FAULTS" BASIS AND (B) SELLER ON BEHALF OF ITSELF AND ITS AFFILIATES, EXPRESSLY DISCLAIMS, AND BUYER ACKNOWLEDGES AND AGREES THAT IT HAS NOT RELIED UPON, ANY REPRESENTATIONS OR WARRANTIES OF ANY KIND OR NATURE, EXPRESS OR IMPLIED, AS TO (I) TITLE TO ANY ASSETS OF THE ACQUIRED COMPANY GROUP, INCLUDING THE ASSETS, (II) THE CONTENTS, CHARACTER OR NATURE OF ANY DESCRIPTIVE MEMORANDUM RELATING TO THE BUSINESS, THE ACQUIRED INTERESTS OR ASSETS OF THE ACQUIRED COMPANY GROUP, (III) ANY ESTIMATES OF THE VALUE OF THE BUSINESS, THE ACQUIRED INTERESTS OR ASSETS OF THE ACQUIRED COMPANY GROUP, OR FUTURE REVENUES GENERATED THEREBY, (IV) THE MAINTENANCE, REPAIR, CONDITION, QUALITY, SUITABILITY, DESIGN, MARKETABILITY, PROSPECTS (FINANCIAL OR OTHERWISE) OR RISKS AND OTHER INCIDENTS OF THE BUSINESS, THE ACQUIRED INTERESTS OR ASSETS OF THE ACQUIRED COMPANY GROUP OR (V) ANY OTHER DUE DILIGENCE INFORMATION, AND FURTHER DISCLAIMS ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE OR CONFORMITY TO MODELS OR SAMPLES. IT BEING EXPRESSLY UNDERSTOOD AND AGREED BY THE PARTIES THAT EXCEPT FOR FRAUD OR THE REPRESENTATIONS, WARRANTIES AND COVENANTS SET FORTH IN THIS AGREEMENT AND THE OTHER TRANSACTION DOCUMENTS, BUYER SHALL BE DEEMED TO BE OBTAINING ALL OF THE SELLER'S EQUITY INTERESTS IN THE ACOUIRED COMPANY GROUP AND THEIR CORRESPONDING DIRECT OR INDIRECT INTERESTS IN THE ASSETS, IN THEIR PRESENT STATUS, CONDITION AND STATE OF REPAIR, "AS IS", "WHERE IS" AND "WITH ALL FAULTS".

7.21. Mutual Releases.

(a) <u>Seller Release</u>. Effective as of the Closing, except for any rights or obligations under this Agreement or any other Transaction Documents, Seller on behalf of itself and its Affiliates, and its and their successors and assigns hereby irrevocably, unconditionally, fully and forever waives, releases, discharges, and covenants not to sue Buyer and the Buyer Indemnified Parties with respect to any and all known or unknown Proceedings, Liabilities or Losses arising out of any matter, act, omission, cause or event relating to Buyer's or Buyer's Affiliates' function as an operator of OGC and OCC or as a member of OGC or OCC.

(b) <u>Buyer Release</u>. Effective as of the Closing, except for any rights or obligations under this Agreement or any other Transaction Documents, Buyer on behalf of itself and its Affiliates and its and their successors and assigns hereby irrevocably, unconditionally, fully and forever waives, releases, discharges, and covenants not to sue Seller and Seller Indemnified Parties with respect to any and all known or unknown Proceedings, Liabilities or Losses arising out of any matter, act, omission, cause or event relating to Seller's or Seller's Affiliates' function as a member of OGC or OCC.

(c) <u>Denial of Liability</u>. The Parties agree that this <u>Section 7.21</u> shall neither constitute nor be construed as an adjudication or finding on the merits of any of the matters released herein and shall not in any manner constitute or be deemed to be an admission of Liability or culpability on the part of any of the Parties, all such Liability being expressly denied.

No Claims. The releases contained in Section 7.21(a) and (b) shall be construed as (d) general releases with respect to the matters released in this Section 7.21 and Seller and Buyer, in each case individually and on behalf of their respective Affiliates, each irrevocably waives its respective rights with respect to the matters released in this Section 7.21 under any Law that provides that a general release does not extend to claims which a releasing party does not know of, or suspect to exist in its favor, at the time of execution of such release, which if known would have affected such releasing party's release agreement with the released party. THE RELEASES AND DISCHARGES OF ANY PERSON UNDER THIS RELEASE SHALL APPLY REGARDLESS OF CAUSE, INCLUDING IF SUCH PERSON WAS JOINTLY, SOLELY OR CONCURRENTLY NEGLIGENT, GROSSLY NEGLIGENT OR OTHERWISE. In entering into this mutual release in this Section 7.21, each Party assumes the risk of any mistake of law or fact with respect to the matters released in this Section 7.21. If a Party later discovers that any fact such Party relied upon in entering this this Section 7.21 is not true, or that such Party's understanding of the facts or the Law with respect to the matters released in this Section 7.21was incorrect, then such Party shall not be entitled to seek rescission of this Section 7.21 by reason thereof.

(e) <u>Covenants Not to Sue</u>. Except for any rights or obligations under this Agreement or any other Transaction Documents, effective with the releases contained in <u>Section 7.21</u>, it is the intent of the Parties that the mutual consideration received pursuant to this Agreement satisfies and finally resolves the controversies released in this <u>Section 7.21</u> and it is expressly intended and agreed that the Parties shall not assert against one another (or the Buyer Indemnified Parties or Seller Indemnified Parties released in <u>Section 7.21</u>) any future claims based on any conduct, act, or omission encompassed by the release set forth this <u>Section 7.21</u>.

ARTICLE VIII Intentionally Deleted

ARTICLE IX Closing

9.1. Time and Place of Closing. Upon the terms and subject to the conditions of this Agreement, the closing of the sale, assignment, conveyance, transfer and delivery of the Acquired Interests to Buyer and the other transactions contemplated by this Agreement (the "Closing") will take place by electronic transmission and shall occur simultaneously with the execution of this Agreement on the Execution Date. The date of the Closing is referred to in this Agreement as the "Closing Date." All actions to be taken and all documents and instruments to be executed and delivered at Closing shall be deemed to have been taken, executed, and delivered simultaneously and, except as permitted hereunder, no actions shall be deemed taken nor any document and instruments have been executed and delivered. Subject to the occurrence of the Closing, all transactions hereunder shall be deemed to have occurred as of 12:01 a.m. Central Time, on the Closing Date.

9.2. Deliveries and Actions at Closing.

(a) <u>Buyer Deliveries and Actions</u>. At the Closing, upon the terms and subject to the conditions of this Agreement, and subject to the simultaneous performance by Seller of its obligations pursuant to <u>Section 9.2(b)</u>, Buyer will execute and deliver, or cause to be executed and delivered, to Seller, each of the following documents, where the execution or delivery of documents is contemplated, and will take or cause to be taken the following actions, where the taking of actions is contemplated:

(i) <u>Purchase Price</u>. Payment of the Closing Purchase Price in accordance with <u>Section 2.2</u>;

(ii) <u>Assignment Agreement</u>. A counterpart of the Assignment Agreement, duly executed by Buyer;

(iii) <u>Transition Services Agreement</u>. A counterpart of a transition services agreement substantially in the form attached hereto as <u>Exhibit D</u> (the "Transition Services Agreement"), duly executed by Buyer;

(iv) <u>Letter Agreement Amendment</u>. A counterpart of the Letter Agreement Amendment, duly executed by MarkWest Liberty Midstream & Resources, L.L.C.; and

(v) <u>R&W Policy</u>. A copy of the final R&W Policy which will be bound and issued on the Closing Date.

(b) <u>Seller Deliveries and Actions</u>. At the Closing, upon the terms and subject to the conditions of this Agreement, and subject to the simultaneous performance by Buyer of its obligations pursuant to <u>Section 9.2(a)</u>, Seller will execute and deliver, or cause to be executed and delivered, to Buyer, each of the following documents, where the execution or delivery of documents is contemplated, and will take or cause to be taken the following actions, where the taking of actions is contemplated:

(i) <u>W-9</u>. A duly completed and executed Internal Revenue Service Form W-9, duly executed by Seller or Summit Parent;

(ii) <u>Assignment Agreement</u>. A counterpart of each of the Assignment Agreement, duly executed by Seller;

(iii) <u>Transition Services Agreement</u>. A counterpart of the Transition Services Agreement, duly executed by Seller;

(iv) <u>Letter Agreement Amendment</u>. A counterpart of the Letter Agreement Amendment, duly executed by Summit Parent;

(v) <u>Lien Release Letter</u>. The Lien Release Letter and all other documents and instruments necessary to release and discharge all Liens

and Debt affecting the Acquired Interests at or promptly after the Closing; and

(vi) <u>Resignations</u>. Resignations of each of the individuals set forth on <u>Schedule 9.2(b)(vi)</u> of the Seller Disclosure Schedule who serves as a committee member, officer, manager or in a similar capacity of the Acquired Company Group Member in his or her capacity as such.

ARTICLE X Intentionally Deleted

ARTICLE XI Indemnification

None of the Seller's representations or warranties in this 11.1. Survival. Agreement shall survive the Closing and all rights, claims and causes of action (whether in Contract or in tort or otherwise, or whether at Law or in equity) with respect thereto shall terminate at the Closing. Buyer's representations and warranties in Section 6.1 (Organization), Section 6.2 (Authority; Enforceability), Section 6.3 (Non-Contravention), Section 6.4 (Governmental Approvals) and Section 6.5 (HSR Act) in this Agreement shall survive the Closing for a period of twelve (12) months following the Closing; provided that none of the Buyer's other representations or warranties in this Agreement shall survive the Closing and all rights, claims and causes of action (whether in Contract or in tort or otherwise, or whether at Law or in equity) with respect thereto shall terminate at the Closing. In the case of any covenants or agreements of Seller and Buyer, as applicable, required to be performed after the Closing under this Agreement, all such covenants and agreements to be performed shall survive until fully performed or performance is no longer required. Furthermore, notwithstanding anything in this Agreement to the contrary, this Section 11.1 shall not limit in any manner (a) the ability of Buyer to recover under R&W Policy for any matters covered thereunder, or (b) Fraud.

11.2. Obligations of Seller. Subject to the terms of this <u>Article XI</u> and <u>Section</u> <u>11.1</u>, effective as of and after the Closing, Seller shall indemnify, defend and hold harmless Buyer, its Affiliates and its and their respective successors, assigns, directors, managers, officers, employees, shareholders, partners, members, equityholders, agents and controlling persons (collectively, "Buyer Indemnified Parties") from and against all Losses that any Buyer Indemnified Party may suffer, sustain or incur, or become subject to, to the extent resulting from, arising out of or in connection with (a) any breach of, or default in the performance of, any covenant or agreement of Seller in this Agreement, and (b) the Excluded Liability.

11.3. Obligations of Buyer. Subject to the terms of this <u>Article XI</u> and <u>Section</u> <u>11.1</u>, effective as of and after the Closing, Buyer shall indemnify, defend and hold harmless each Seller, its Affiliates and its and their respective successors, assigns, directors, managers, officers, employees, shareholders, partners, members, equityholders, agents and controlling persons (collectively, the "Seller Indemnified Parties") from and against all Losses that any Seller Indemnified Party may suffer, sustain or incur, or become subject to, to the extent resulting from or arising out of or in connection with (a) any breach of Buyer's representations and warranties in this Agreement or (b) any breach, or default in the performance of, any covenant or agreement of Buyer in this Agreement.

11.4. Limitations on Liability.

(a) Subject to this <u>Article XI</u>, except in the event of Fraud, after the Closing, the sole and exclusive remedy of Buyer and its Affiliates with respect to the breach of any representations and warranties of Seller in this Agreement shall be limited to the R&W Policy (including, for the avoidance of doubt, whether Buyer obtains such R&W Policy at or after the Closing, if at all, and maintains, following the Closing, the R&W Policy, whether the R&W Policy is revoked, cancelled or modified in any manner after the Closing, or whether any Proceeding by Buyer or any of its Affiliates or Representatives is denied under the R&W Policy). For the avoidance of doubt, the absence of coverage under the R&W Policy for any reason, including due to exclusions from coverage thereunder or the failure of the R&W Policy to be in full force and effect for any reason, shall not expand, alter, amend, change or otherwise affect Seller's or its Affiliates' or Representatives' or any of their respective successors' and assigns' liability under this Agreement.

(b) Notwithstanding anything to the contrary herein, except in the event of Fraud, in no event shall the aggregate liability of the Seller in connection with this Agreement, including the transactions contemplated herein and therein, exceed the Purchase Price.

(c) The Seller Indemnified Parties shall take, and shall cause Seller Indemnified Parties to take, all commercially reasonable steps to mitigate a Loss upon becoming aware thereof and of the event which gives rise thereto. Such Seller Indemnified Parties will each also use commercially reasonable efforts to pursue claims and collect any amounts to which they may be entitled under insurance policies or from third parties (pursuant to indemnification agreements or otherwise). The Parties agree that no Seller Indemnified Party shall have any recourse under this <u>Article XI</u> for any Losses that such Seller Indemnified Party would not have suffered had such Seller Indemnified Party exercised commercially reasonable efforts to mitigate such Losses within a reasonable amount of time following the discovery by such Seller Indemnified Party that is not a natural Person, the bringing to the attention of a responsible officer thereof of such fact, event or circumstance). Notwithstanding anything to the contrary herein, nothing in this <u>Article XI</u> shall operate to waive any Indemnified Party's common law duty to mitigate any Losses.

(d) The amount of any and all Losses indemnifiable under this Agreement shall be determined net of any amounts actually recovered by an Indemnified Party (or any of its Affiliates) with respect to such Losses from any third party, including any indemnity, contribution or other similar payment, or under or pursuant to any insurance policy pursuant to which or under which such Indemnified Party (or any of its Affiliates) has rights, it being understood that any amounts recovered under or pursuant to any insurance policy shall be calculated net of any reasonable costs associated with pursuing such insurance proceeds and reasonable out-of-pocket costs and any costs or expenses attributable to increases in insurance premiums.

(e) Notwithstanding anything to the contrary contained in this Agreement, under no circumstances will any Party or any of its Affiliates be entitled to recover more than one time for any Loss under this Agreement, and to the extent a Party or any of its Affiliates is compensated

for a matter through the adjustments provided for in <u>Section 2.3</u> or otherwise, such Party and its Affiliates will not have a separate right to indemnification or adjustment in another portion of <u>Section 2.3</u> for such matter. Without limiting the generality of the prior sentence, if a set of facts, conditions or events constitutes a breach of more than one covenant or agreement that is subject to the indemnification obligations under this <u>Article XI</u> as applicable, only one recovery of Losses shall be allowed, and in no event shall there be any indemnification or duplication of payments or recovery under different provisions of this Agreement arising out of the same facts, conditions or events.

(f) No indemnifying Person will be liable for any Losses that are subject to indemnification under <u>Section 11.2</u> or <u>Section 11.3</u> unless a written demand for indemnification under this Agreement is delivered by the indemnified Person to the indemnifying Person with respect thereto prior to 5:00 P.M. Central Time on the date pursuant to <u>Section 11.1</u> on which the applicable survival period expires, to assert a claim for indemnification describing such claim in reasonable detail, including the factual circumstances giving rise to and the provisions under this Agreement on which such claim is based. In the event notice of any claim for indemnification under this <u>Article XI</u> has been given within the applicable survival period, the representations, warranties or covenants that are the subject of such indemnification claim shall survive solely with respect to such claim until such time as such claim is finally resolved, it being understood that such representations, warranties and covenants shall not survive with respect to any other claim that has not been timely given within such survival period.

11.5. Indemnification Procedure.

Any Person making a claim for indemnification under this Article XI (a) (an "Indemnified Party") shall promptly notify the indemnifying party (an "Indemnifying Party") of the claim in writing after receiving written notice of any Proceeding or other claim against it by a third party (a "Third-Party Claim"), reasonably describing the nature of the Third-Party Claim, the amount thereof (if known and quantifiable), and the basis thereof and attaching copies of all correspondence related thereto; provided, that the failure to give reasonably prompt notice to an Indemnifying Party shall not relieve an Indemnifying Party of its obligations hereunder, except to the extent that an Indemnifying Party is prejudiced thereby. The Parties shall cooperate in good faith to resolve any disputed claim for indemnification. Any Indemnifying Party shall be entitled to participate in the defense of such Third-Party Claim giving rise to an Indemnified Party 's claim for indemnification at such Indemnifying Party's expense, and at its option (subject to the limitations set forth below) shall be entitled to assume the control of the defense thereof by appointing counsel reasonably acceptable to the Indemnified Party to be the lead counsel in connection with such defense; provided, that (i) the Indemnified Party shall be entitled to participate in the defense of such claim and to employ counsel of its choice for such purpose; provided, that the fees and expenses of such separate counsel shall be borne entirely by the Indemnified Party; (ii) Seller shall not be entitled to assume control of such defense if (A) the claim for indemnification relates to or arises in connection with any criminal Proceeding, (B) the claim primarily seeks an injunction or other equitable relief against a Buyer Indemnified Party. (C) the Indemnified Party has been advised in writing by legal counsel that a conflict of interest exists which, under applicable principles of legal ethics, would prohibit a single legal counsel from representing both the Indemnifying Party and Indemnified Party in such Third-Party Claim or there may be legal defenses to the Indemnified Party which are different from or additional to those available to the Indemnifying Party or (D) the maximum likely Losses under such Third-Party Claim are greater than the available indemnification amount from the Indemnifying Party (after taking into account the amount of all other claims for which the Indemnifying Party may be or may be claimed to be liable); and (iii) if an Indemnifying Party shall control the defense of any such Third-Party Claim, such Indemnifying Party shall obtain the prior written consent of the Indemnified Party (which shall not be unreasonably withheld, conditioned or delayed) before entering into any settlement of a claim; provided, however, subject to the limitations set forth in this Article XI, an Indemnifying Party may settle or consent to the entry of judgment in respect of such Third-Party Claim without the consent of the Indemnified Party, if such settlement or judgment is for (x) money damages only, (y) includes a full release of the Indemnified Party from any further Liability in respect of such claim, and (z) does not contain any admission of wrongdoing on the part of the Indemnified Party. Irrespective of which party controls the defense of any Third-Party Claim, the other parties to this Agreement will, and will cause any non-party Affiliate to, cooperate with the controlling party in such defense and make available to the controlling party all witnesses, pertinent records, materials and information in such non-controlling party's possession or under its control relating thereto as is reasonably required by the controlling party. The parties agree that all communications between any party and counsel responsible for or participating in the defense of any Indemnification Claim shall, to the extent possible, be made so as to preserve any applicable attorney-client or work-product privilege.

Any Person making a claim for indemnification under this Article XI on account of (b) a Loss which does not result from a Third-Party Claim (a "Direct Claim") shall assert such Direct Claim by giving the Indemnifying Party prompt written notice thereof. Such notice by the Indemnified Party shall describe the Direct Claim in reasonable detail, shall include copies of all material written evidence thereof (including without limitation appropriate timing justification for such Direct Claim), and shall indicate the estimated amount of the Loss that has been or may be sustained by the Indemnified Party. The Indemnifying Party shall have thirty (30) calendar days after its receipt of such notice to respond in writing to such Direct Claim. During such 30-day period, the Indemnified Party shall allow the Indemnifying Party and its professional advisors to investigate the matter or circumstance alleged to give rise to the Direct Claim at the Indemnifying Party 's cost, and whether and to what extent any amount is payable in respect of the Direct Claim and the Indemnified Party shall reasonably assist the Indemnifying Party's investigation by giving such information and assistance as the Indemnified Party or any of its professional advisors may reasonably request. If the Indemnifying Party does not so respond within such 30-day period, the Indemnifying Party shall be deemed to have rejected such claim, in which case the Indemnified Party shall be free to pursue such remedies as may be available to the Indemnified Party on the terms and subject to the provisions of this Agreement.

11.6. Damages Waiver. NOTWITHSTANDING ANYTHING TO THE CONTRARY IN THIS AGREEMENT, BUYER SHALL NOT BE LIABLE TO SELLER OR TO SELLER INDEMNIFIED PARTIES UNDER THIS <u>ARTICLE XI</u> FOR SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES SUFFERED BY SELLER INDEMNIFIED PARTIES RESULTING FROM OR ARISING OUT OF BUYER'S BREACH OF REPRESENTATIONS AND WARRANTIES IN <u>SECTION 6</u> OF THIS AGREEMENT, EXCEPT TO THE EXTENT ANY SELLER INDEMNIFIED PARTY SUFFERS SUCH DAMAGES TO A THIRD PARTY, WHICH DAMAGES (INCLUDING COSTS OF DEFENSE AND REASONABLE ATTORNEYS' FEES INCURRED IN CONNECTION WITH DEFENDING AGAINST SUCH DAMAGES), IN EACH CASE SHALL NOT BE EXCLUDED BY THIS PROVISION AS TO RECOVERY HEREUNDER.

Exclusive Remedy. Notwithstanding anything contained herein to the 11.7. contrary but subject to Section 12.16, except in the case of (i) Fraud, (ii) recovery against the R&W Policy, (iii) any claims seeking specific performance pursuant to Section 12.16 with respect to the matters covered by Section 2.3 (Purchase Price Adjustment) and (iv) any remedy for Losses that are provided pursuant to any other Transaction Document (which shall be governed exclusively by such Transaction Document), the indemnification provisions pursuant to the provisions of this Article XI shall be the sole and exclusive remedy for the Parties for any misrepresentation or breach of any representation, warranty, covenant, agreement or other provision contained in this Agreement, or in any certificate or other instrument or document delivered pursuant hereto, or otherwise in connection with any claims related to subject matter or the transactions contemplated herein, including any claims relating to the Acquired Company Group, the business of the Acquired Company Group or Seller's direct or indirect ownership of the Acquired Interests and the Non-Op Interests, regardless of the legal theory under which such liability or obligation may be sought to be imposed, whether sounding in contract or tort, or whether at law or in equity, or otherwise. Each of the Parties acknowledge and agree, on its own behalf and on behalf of the other Buyer Indemnified Parties or Seller Indemnified Parties, as applicable, that they may not avoid the limitation on liability set forth in this Section 11.7 by (x) seeking damages for breach of contract, tort or pursuant to any other theory of liability, all of which are hereby expressly waived or (y) asserting or threatening any claim against any Person that is not a party (or a successor to a Party) for breaches of the representations, warranties or covenants contained in this Agreement.

ARTICLE XII

Miscellaneous

Governing Law. THIS AGREEMENT, ANY DISPUTE OR MATTER 12.1. ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT, THE TRANSACTION DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY, INCLUDING ALL CLAIMS OR CAUSES OF ACTION (WHETHER AT LAW, IN EQUITY, IN CONTRACT, IN TORT, OR OTHERWISE) THAT MAY BE BASED UPON, ARISE OUT OF, OR RELATE TO THIS AGREEMENT, THE TRANSACTION DOCUMENTS, OR THE NEGOTIATION, EXECUTION OR PERFORMANCE HEREOF OR THEREOF, AND THE LEGAL RELATIONSHIP AMONG THE PARTIES, WILL BE GOVERNED BY, CONSTRUED, AND ENFORCED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE, INCLUDING, WITHOUT LIMITATION, DELAWARE LAWS RELATING TO APPLICABLE STATUTES OF LIMITATION AND BURDENS OF PROOF AND AVAILABLE REMEDIES, WITHOUT REGARD TO CHOICE OF LAW PRINCIPLES THAT WOULD REQUIRE THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION.

12.2. Consent to Jurisdiction. EACH PARTY AGREES THAT THE APPROPRIATE, EXCLUSIVE AND CONVENIENT FORUM FOR ANY DISPUTES BETWEEN ANY OF THE PARTIES RELATED TO, ARISING OUT OF OR IN

CONNECTION WITH THIS AGREEMENT. THE TRANSACTION DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY WILL BE IN THE DELAWARE COURT OF CHANCERY (OR, ONLY IF THE DELAWARE COURT OF CHANCERY DECLINES TO ACCEPT JURISDICTION OVER A PARTICULAR MATTER, IN ANY OF THE FEDERAL OR STATE COURTS LOCATED IN THE STATE OF DELAWARE) AND EACH OF THE PARTIES IRREVOCABLY SUBMITS TO THE JURISDICTION AND VENUE OF SUCH COURTS SOLELY IN RESPECT OF ANY SUCH DISPUTES AND AGREE THAT THOSE COURTS HAVE JURISDICTION OVER THE PARTIES AND THAT THEY CONSTITUTE A CONVENIENT FORUM. THE PARTIES FURTHER AGREE THAT THE PARTIES WILL NOT BRING SUIT WITH RESPECT TO ANY SUCH DISPUTES IN ANY COURT OR JURISDICTION OTHER THAN THE ABOVE SPECIFIED COURTS; PROVIDED, HOWEVER, THAT THE FOREGOING SHALL NOT LIMIT THE RIGHTS OF THE PARTIES TO OBTAIN EXECUTION OF JUDGMENT IN ANY OTHER JURISDICTION. EACH PARTY AGREES THAT A FINAL AND NO LONGER APPEALABLE JUDGMENT AGAINST A PARTY IN ANY SUCH DISPUTE SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW.

12.3. Waiver of Jury Trial. EACH PARTY HEREBY IRREVOCABLY WAIVES, AND COVENANTS THAT IT WILL NOT ASSERT (WHETHER AS PLAINTIFF, DEFENDANT OR OTHERWISE), ANY RIGHT TO TRIAL BY JURY IN ANY FORUM IN RESPECT OF ANY ISSUE, CLAIM, DEMAND, ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER SOUNDING IN TORT OR CONTRACT OR OTHERWISE), RELATED TO, ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT, THE TRANSACTION DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. ANY PARTY MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS <u>SECTION 12.3</u> WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF EACH PARTY TO THE WAIVER OF ITS RIGHT TO TRIAL BY JURY.

12.4. Amendment and Modification. This Agreement may be amended, modified or supplemented only by written agreement of the Parties.

12.5. Waiver of Compliance; Consents. Except as otherwise provided in this Agreement, any failure of any of the Parties to comply with any obligation, covenant, agreement or condition in this Agreement may be waived by the Party entitled to the benefits thereof only by a written instrument signed by the Party granting such waiver, but such waiver or failure to insist upon strict compliance with such obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure.

12.6. Notices. Any notice, demand or communication required or permitted under this Agreement shall be in writing and delivered personally, by reputable overnight delivery service or other courier or by certified mail, postage prepaid, return receipt requested, or by e-mail and shall be deemed to have been duly given (a) as of the date of delivery if delivered personally or by overnight delivery service or other courier, or by e-mail (if delivered prior to 5 p.m. Houston, Texas time or, if thereafter, then as of the next day), or (b) on the date receipt is acknowledged if delivered by certified mail, addressed as follows; provided that a notice of a change of address shall be effective only upon receipt thereof:

If to Seller:

Summit Midstream Holdings, LLC 910 Louisiana Street, Suite 4200 Houston, TX 77002 Attention: Legal Department Email: [Omitted.]

with a copy (which shall not constitute notice) to:

Locke Lord LLP 600 Travis Street Suite 2800 Houston, Texas 77002 Attention: H. William Swanstrom; Jennie Simmons Email: [Omitted.]

If to Buyer, to:

Mark West Operating Utica, L.L.C. 1515 Arapahoe Street Tower I, Suite 1600 Denver, CO 80202 Attn: Contract Administration Email: [Omitted.]

With a copy (which shall not constitute notice) to:

Mark West Operating Utica, L.L.C. 1515 Arapahoe Street Tower I, Suite 1600 Denver, CO 80202 Attn: Law Department Email: [Omitted.]

12.7. Assignment. This Agreement shall be binding upon and inure to the benefit of the Parties and their successors and permitted assigns. Neither Party may assign or transfer this Agreement or any of its rights, interests or obligations under this Agreement without the prior written consent of the other Party. Any attempted assignment or transfer in violation of this Agreement shall be null, void and ineffective.

12.8. Third Party Beneficiaries. Nothing in this Agreement shall entitle any Person other than Buyer and Seller to any claim, cause of action, remedy or right of any kind, except the rights expressly provided to the Persons in this Agreement, in each case, only to the extent such rights are exercised or pursued, if at all, by Seller or Buyer acting on behalf of such Person (which rights may be exercised in the sole discretion of the applicable Party hereunder). Notwithstanding the foregoing: (a) the Parties reserve the right to amend, modify, terminate, supplement, or waive any provision of this Agreement or this entire Agreement without the consent or approval of any other Person and (b) no Party hereunder shall have any direct Liability to any permitted third-party beneficiary, nor shall any permitted third-party beneficiary have any right to exercise any rights hereunder for such third-party beneficiary's benefit except to the extent such rights are brought, exercised and administered by a Party hereto.

12.9. Entire Agreement. This Agreement (including the Annexes, Schedules and Exhibits attached hereto) and the other Transaction Documents constitute the entire agreement and understanding of the Parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both oral and written, between the Parties with respect to such subject matter.

12.10. Severability. Whenever possible, each provision or portion of any provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable Law but if any provision or portion of any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any applicable Law or rule in any jurisdiction, such invalidity, illegality or unenforceability will not affect any other provision or portion of any provision in such jurisdiction, and this Agreement will be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision or portion of any provision or portion of any provision or portion.

12.11. Representation by Counsel. Each Party agrees that it has been represented by independent counsel of its choice during the negotiation and execution of this Agreement and the documents referred to herein, and that it has executed the same upon the advice of such independent counsel. Each Party and its counsel cooperated in the drafting and preparation of this Agreement and the documents referred to herein, and any and all drafts relating thereto shall be deemed the work product of both Parties and may not be construed against either Party by reason of its preparation. Therefore, the Parties waive the application of any Law providing that ambiguities in an agreement or other document will be construed against the Party drafting such agreement or document.

12.12. Seller Disclosure Schedule. Seller may, at its option, include in the Seller Disclosure Schedule items that are not material or required by this Agreement and the inclusion of any information (including dollar amounts) in any Schedule of the Seller Disclosure Schedule shall not be deemed to be an admission or acknowledgment by Seller that such information is required to be listed on such Schedule of the Seller Disclosure Schedule or is material to or outside the ordinary course of the business of Seller or the Acquired Company Group. The information contained in this Agreement, the Exhibits and the Seller Disclosure Schedule is disclosed solely for purposes of this Agreement, and no information contained in this Agreement, the Exhibits or the Seller Disclosure Schedule shall be deemed to be an admission by any Party to any third Person of any matter whatsoever (including any violation

of a legal requirement or breach of contract). In no event shall disclosure of any matter, fact, occurrence, information or circumstance in the Schedules be deemed or interpreted to broaden the scope of the representations and warranties, obligations, covenants, conditions, indemnities or agreements contained in this Agreement, or to create any representation, warranty, obligation, covenant, condition, indemnity or agreement that is not contained in this Agreement. In particular, if the Seller Disclosure Schedule contains supplementary information not specifically required under this Agreement to be included in the Schedules, such additional matters are set forth solely for informational purposes, are not represented or warranted in this Agreement and do not necessarily include other matters of a similar nature. In disclosing any matter, fact, occurrence, information or circumstance in the Schedules, no Party is waiving any attorney-client privilege associated with any such matter, fact, occurrence, information or circumstance, or any protection afforded by the "work product doctrine" with respect to any of the same. Any exception, qualification or other disclosure set forth on the Seller Disclosure Schedule with respect to a particular representation, warranty or covenant contained in this Agreement shall be deemed to be an exception, gualification or other disclosure with respect to all other representations, warranties and covenants contained in this Agreement to the extent any description of facts regarding the event, item or matter disclosed is adequate so as to make reasonably apparent that such exception, qualification or disclosure is applicable to such other representations, warranties or covenants whether or not such exception, qualification or disclosure is so numbered. Any capitalized terms used in the Seller Disclosure Schedule but not otherwise defined therein shall be defined as set forth in this Agreement.

12.13. Facsimiles; Counterparts. This Agreement may be executed in one or more counterparts, each of which, when executed, shall be deemed to be an original and all of which together shall constitute one and the same document. Either Party's delivery of any executed counterpart signature page by facsimile (or electronic .pdf format transmission) is as effective as executing and delivering this Agreement in the presence of the other Party, and such signature shall be deemed binding for all purposes hereof, without delivery of an original signature being thereafter required.

Privileged Communications. As to all communications among Locke Lord 12.14. LLP, on the one hand, and Seller, the Acquired Company, or their respective Affiliates and Representatives, on the other hand, that relate in any way to the transactions contemplated by this Agreement that constitute attorney-client privileged communications or are otherwise privileged under applicable Law (collectively, the "Privileged Communications"), the privilege and the expectation of client confidence belongs to Seller and may be controlled by Seller and shall not pass to or be claimed by Buyer, the Acquired Company or any of their respective Affiliates. The Privileged Communications are the property of Seller, and from and after the Closing Date, none of Buyer, the Acquired Company or any of their respective Affiliates, nor any Person purporting to act on behalf of Buyer, the Acquired Company or any of their respective Affiliates will access or use the Privileged Communications, including by way of review of any electronic data, communications or other information, or seek to obtain such Privileged Communications, whether by seeking a waiver of the privilege or through other means. As to any such Privileged Communications prior to the Closing Date, none of Buyer, the Acquired Company or any of their respective Affiliates, successors or assigns, may disclose, use or rely on any of the Privileged Communications after the Closing; provided, however, the foregoing shall not restrict the ability of Buyer, the Acquired Company or any of their respective Affiliates to challenge the fact that any communication is a Privileged Communication (other than as a result of Buyer becoming the owner of the Acquired Interests). The Privileged Communications may be used by Seller and its respective Affiliates in connection with any dispute that relates in any way to the transactions contemplated by this Agreement. Notwithstanding the foregoing, in the event that a dispute arises between Buyer, the Acquired Company and a third Person (other than a Party to this Agreement or any of such Party's respective Affiliates) after the Closing, the Acquired Company may assert the privilege to prevent disclosure of the Privileged Communications to such third Person; provided, however, that the Acquired Company may not, unless required by applicable Law, waive such privilege without the prior written consent of Seller. In the event that Buyer or the Acquired Company are legally required by order or otherwise to access or obtain a copy of all or a portion of the Privileged Communications, Buyer shall immediately (and, in any event, within two (2) Business Days) notify Seller in writing (including by making specific reference to this Section 12.14) so that Seller can seek a protective order and Buyer agrees to use all reasonable efforts to assist therewith.

12.15. Affiliate Liability. Each of the following is herein referred to as a "Non-Recourse Party": (a) any direct or indirect holder of Equity Interests or securities in a Party (whether limited or general partners, members, stockholders or otherwise), or any investment fund organized by or managed by any of the foregoing Persons, (b) any director, officer, employee, representative or agent of s Party or any Person who controls the Party, all as prior to the date of this Agreement or (c) any portfolio or operating company of any Person described in clauses (a), (b) or (c) (other than Buyer, Seller, the Acquired Company or any Person controlled by a Party or the Acquired Company). No Non-Recourse Party shall have any Liability or obligation of any nature whatsoever in connection with or under this Agreement, or the transactions contemplated herein or therein, and each Party, for itself and on behalf of the Seller Indemnified Parties or Buyer Indemnified Parties, as applicable, hereby waives and releases all claims of any such Liability and obligation. This Agreement may only be enforced against, and any dispute, controversy, matter or claim based on, related to, or arising out of this Agreement, or the negotiation, performance, or consummation of this Agreement, may only be brought against, the entities that are expressly named as Parties, and then only with respect to the specific obligations set forth herein with respect to such Party. Each Non-Recourse Party is expressly intended as a third-party beneficiary of this Section 12.15.

12.16. Specific Performance. Each Party hereby acknowledges and agrees that the rights of each Party to consummate the transactions contemplated hereby are special, unique and of extraordinary character and that, if any Party violates or fails or refuses to perform any covenant or agreement made by it herein, the non-breaching Party may be without an adequate remedy at Law. If any Party violates or fails or refuses to perform any covenant or agreement made by such Party herein, the non-breaching Party, subject to the terms hereof and in addition to any remedy at Law for damages or other relief permitted under this Agreement, may (at any time prior to the valid termination of this Agreement pursuant to <u>Article X</u>) institute and prosecute an action in any court of competent jurisdiction to enforce specific performance of such covenant or agreement or seek any other equitable relief, without the necessity of proving actual damages or posting of a bond.

12.17. Time is of the Essence. This Agreement contains a number of dates and times by which performance or exercise of rights is due, and the Parties intend that each and every such date and time be the firm and final date and time, as agreed. For this reason, each Party hereby waives and relinquishes any right it might otherwise have to challenge its failure to meet any performance or rights election date and time applicable to it on the basis that its late action constitutes substantial performance. Without limiting the foregoing, time is of the essence in this Agreement.

12.18. Seller Parent Guaranty. Subject to the terms and conditions of this Agreement, Seller Parent hereby guarantees the payment and performance of Seller's obligations under this Agreement solely with respect to Seller's indemnification obligations with respect to the Excluded Liability under Section 11.2(b) (the "Financial Security"). The Financial Security is valid and in full force and effect and constitutes the valid and binding obligation of Seller Parent, enforceable in accordance with its terms. The Financial Security is an irrevocable guarantee of payment and performance (and not of collection) of any Losses incurred by the Buyer Indemnified Parties in connection with Seller's indemnification obligations with respect to the Excluded Liability under Section 11.2(b), and shall continue in effect in accordance with the terms of this Agreement. The obligations of Seller Parent hereunder shall not be affected by or contingent upon (a) the liquidation or dissolution of, or the merger or consolidation of Seller or its Affiliates with or into any Person or any sale or transfer by Seller or its Affiliate of all or any part of its property or assets, (b) the bankruptcy, receivership, insolvency, reorganization or similar proceedings involving or affecting Seller or its Affiliates, or (c) any disability or any other defense of Seller or any other Person (with or without notice) which might otherwise constitute a legal or equitable discharge of a surety or a guarantor or otherwise. In connection with the foregoing, in the event of a default or failure to perform or make payment of the Financial Security by Seller when due under this Agreement (after any cure period set forth in this Agreement) or any other occurrence that might require Seller Parent to make payment(s) pursuant to the terms herein, Buyer shall be required to submit written notice to Seller Parent describing in reasonable detail (i) the default or failure on the part of Seller, (ii) the Financial Security to be paid and the amount of such Financial Security and (iii) Buyer's payment instructions including bank name, bank routing number and bank account number. Within fifteen (15) Business Days receipt of such written notice, Seller Parent shall be required to, or shall cause Seller to, complete the performance or make the payment due under this Agreement to Buyer. Notwithstanding anything to the contrary contained herein, all of Seller Parent's obligations under this Section 12.18 shall be subject to the terms, survival and limitations set forth in Article XI of this Agreement.

[Signature page follows]

IN WITNESS WHEREOF, each Party has caused this Agreement to be executed by its respective duly authorized officers as of the date first above written.

SELLER:

SUMMIT MIDSTREAM OPCO, LP

By: Summit Midstream Marketing, LLC, its general partner

By: <u>/s/ J. Heath Deneke</u> Name: J. Heath Deneke Title: President and Chief Executive Officer

[SIGNATURE PAGE TO PURCHASE AND SALE AGREEMENT]

SOLELY FOR PURPOSES OF <u>SECTION 12.18</u> HEREOF:

SELLER PARENT:

SUMMIT MIDSTREAM PARTNERS, LP

By: Summit Midstream GP, LLC, its general partner

By: <u>/s/ J. Heath Deneke</u> Name: J. Heath Deneke Title: President and Chief Executive Officer

[SIGNATURE PAGE TO PURCHASE AND SALE AGREEMENT]

BUYER:

MARKWEST UTICA OPERATING COMPANY, L.L.C.

By: <u>/s/ Greg Floerke</u> Name: Greg Floerke Title: President

[SIGNATURE PAGE TO PURCHASE AND SALE AGREEMENT]

<u>EXHIBIT A</u>

FORM OF SETTLEMENT STATEMENT

EXHIBIT B

R&W POLICY COMMITMENT

<u>EXHIBIT C</u>

FORM OF ASSIGNMENT OF INTERESTS

<u>EXHIBIT D</u>

FORM OF TRANSITION SERVICES AGREEMENT

<u>EXHIBIT E</u>

LETTER AGREEMENT AMENDMENT

<u>EXHIBIT F</u>

EXCLUDED LIABILITY

ANNEX 1

ILLUSTRATIVE WORKING CAPITAL CALCULATION

ANNEX 2

ACCOUNTING PRINCIPLES

FOURTH AMENDMENT TO LOAN AND SECURITY AGREEMENT

This FOURTH AMENDMENT TO LOAN AND SECURITY AGREEMENT (this "Fourth Amendment") is dated as of March 22, 2024 (the "Fourth Amendment Effective Date"), among SUMMIT MIDSTREAM PARTNERS, LP, a Delaware limited partnership (the "MLP Entity"), SUMMIT MIDSTREAM HOLDINGS, LLC, a Delaware limited liability company ("Borrower"), the Subsidiary Guarantors (as defined in the Loan Agreement referred to below), the Lenders (as defined in the Loan Agreement) party hereto and BANK OF AMERICA, N.A., as Agent (as defined in the Loan Agreement) for the Lenders (in such capacity, "Agent").

RECITALS:

WHEREAS, the MLP Entity, Borrower, Agent and the Lenders have entered into that certain Loan and Security Agreement dated as of November 2, 2021 (as amended prior to the date hereof including by the First Amendment to Loan and Security Agreement dated as of October 14, 2022, the Second Amendment to Loan and Security Agreement dated as of May 10, 2023, and the Third Amendment to Loan and Security Agreement dated as of May 10, 2023, and the Third Amendment to Loan and Security Agreement, as an Security Agreement, the "Loan Agreement,". Capitalized terms used herein but not otherwise defined herein have the meanings given to such terms in the Loan Agreement;

WHEREAS, Borrower has informed Agent and the Lenders that Summit Midstream Opco, LP, a Delaware limited partnership (the "Seller"), intends to sell all of the equity interests in Summit Midstream Utica, LLC, a Delaware limited liability company, which includes all of its assets, (such sale, the "Northeast Sale"), pursuant to that certain Purchase and Sale Agreement dated on or around March 22, 2024 (the "Northeast Sale Agreement"), by and among the Seller, MarkWest Utica Operating Company, L.L.C. and, solely for the limited purposes set forth therein, the MLP Entity, for a purchase price which requires the consent of the Required Lenders under Section 10.2.6 of the Existing Loan Agreement;

WHEREAS, the MLP Entity has advised Agent and the Lenders that it intends to make certain changes to its corporate structure to change from a master limited partnership structure to a "C-corporation" structure by restructuring the ownership of MLP Entity to include one or more new parent and intermediary entities (effected through a subsidiary merger or otherwise) or converting to a corporation (any such change, the "C-Corp Conversion");

WHEREAS, the MLP Entity and Borrower have requested that Lenders constituting Required Lenders and Regions Bank, as a Lender, agree to (a) make certain amendments to the Existing Loan Agreement to, among other things, permit the C-Corp Conversion and (b) consent to the Northeast Sale, and such Lenders have agreed to such amendments and consent on the terms and conditions set forth herein;

NOW, THEREFORE, in consideration of the premises and the mutual agreements, representations and warranties herein set forth, and for other good and valuable consideration, the receipt and sufficiency of which are acknowledged, the MLP Entity, Borrower, the Subsidiary Guarantors, Agent and the undersigned Lenders do hereby agree as follows:

SECTION 1. AMENDMENTS TO EXISTING LOAN AGREEMENT. In reliance on the representations, warranties, covenants and agreements contained in this Fourth Amendment, and subject to the satisfaction (or waiver) of the conditions precedent set forth in <u>Section 3</u> hereof, the Existing Loan Agreement is hereby amended effective as of the Fourth Amendment Effective Date in the manner provided in this <u>Section 1</u>.

1.1. <u>Amendments to Section 1.1</u>.

(a) Section 1.1 of the Existing Loan Agreement is hereby amended to add thereto, in alphabetical order, the following definitions which shall read in full as follows:

"C-Corp Conversion: has the meaning assigned to such term in the Fourth Amendment."

"<u>Fourth Amendment</u>: that certain Fourth Amendment to Loan and Security Agreement dated as of the Fourth Amendment Effective Date, among Borrower, the MLP Entity, Agent and the Lenders party thereto."

"Fourth Amendment Effective Date: March 22, 2024."

"New Parent: has the meaning assigned to such term in "Change in Control"."

"Northeast Sale: has the meaning assigned to such term in the Fourth Amendment."

(b) Section 1.1 of the Existing Loan Agreement is hereby further amended to amend and restate the following definitions in their respective entireties to read in full as follows:

"Change in Control: the occurrence of any of the following: (a) a "change of control" (or any other similar event) under any Material Debt, (b) the acquisition of ownership, directly or indirectly, beneficially or of record, by any Person or group (within the meaning of the Securities Exchange Act of 1934 and the rules of the SEC thereunder as in effect on the date hereof) of Equity Interests representing more than 50% of (A) if the MLP Entity is a limited partnership, (i) the aggregate ordinary voting power represented by the issued and outstanding Equity Interests of the General Partner or (ii) the economic interest represented by the issued and outstanding Equity Interests of the General Partner or (B) if the MLP Entity has converted from a limited partnership to a corporation or other entity, from and after the date of conversion, the aggregate ordinary voting power represented by the issued and outstanding Equity Interests of the MLP Entity entitled to vote for the board of directors (or similar governing body) of the MLP Entity, (c) if the MLP Entity is a limited partnership, the General Partner shall cease to be the sole general partner of the MLP Entity, with no substantial reduction in its powers to manage the MLP Entity as are granted to the General Partner under the MLP Entity's Partnership Agreement as in effect on the Closing Date or (d) the MLP Entity shall cease to own, directly or indirectly, 100% of the Equity Interests of Borrower, free and clear of all Liens other than Liens granted pursuant to the Loan Documents.

In addition, notwithstanding the foregoing, (A) a transaction in which the MLP Entity becomes a direct or indirect subsidiary of another person in connection with a C-Corp Conversion (such person the "<u>New Parent</u>" and any intermediate holding companies between New Parent and the MLP Entity, each an "<u>Intermediate Holding Company</u>") shall not constitute a Change in Control if immediately following the consummation of such transaction, (1) the persons who beneficially own, directly or indirectly, MLP Entity immediately prior to such transaction continue to beneficially own, directly or indirectly through one or more intermediaries, more than 50% of the aggregate ordinary voting power represented by the issued and outstanding Equity Interests of the General Partner or that are entitled to vote for the board of directors, as applicable, of the New Parent, and (2) no Person or group (within the meaning of the Securities Exchange Act of 1934 and the rules of the SEC thereunder as in effect on the date hereof) acquires ownership, directly or indirectly, beneficially or of record, of Equity Interests representing more than 50% of the aggregate ordinary voting power represented by the issued and outstanding Equity Interests of the General Partner (except for the New Parent and any Intermediate Holding Companies) or that are entitled to vote for the board of directors, as applicable, of the New Parent and (B) in connection with any C-Corp Conversion in which the New Parent exists. a Change of Control shall occur unless following such C-Corp Conversion (1) the New Parent and each Intermediary Holding Company shall execute and deliver to Agent a joinder to this Agreement and execute and deliver such other documents as are reasonably requested by Agent, and take such actions necessary or reasonably advisable as determined by Agent, to guarantee the Obligations, (2) the Borrower is a Wholly Owned Subsidiary of such New Parent and any Intermediary Holding Companies, (3) all Equity Interests of Borrower owned by the New Parent or any Intermediary Holding Company are pledged to Agent for the benefit of the Secured Parties (including, without limitation, delivery of original certificates evidencing the Equity Interests, if any, of Borrower, together with an appropriate undated stock powers for each certificate duly executed in blank by the registered owner thereof), and (4) the Obligors execute and deliver to Agent such additional documents and instruments as it may reasonably request in connection with the C-Corp Conversion."

"<u>Consolidated Net Debt</u>: at any date shall mean (a) Consolidated Debt of Borrower and the Restricted Subsidiaries on such date, minus (b) Unrestricted Cash on such date in an aggregate amount not to exceed \$50,000,000, minus (c) until the Senior Secured Mirror Notes Redemption Date, the aggregate amount of cash and Cash Equivalents held in the Proceeds Account on such date in an aggregate amount not to exceed the sum of (i) the initial proceeds from the issuance of the Senior Secured Mirror Notes and (ii) the projected accrued interest on the aggregate outstanding principal amount of Senior Secured Mirror Notes through the Senior Secured Mirror Notes Redemption Date; provided that if no Loans are outstanding, for purposes of calculating the Total Net Leverage Ratio, the Borrower will be permitted to subtract from Consolidated Debt prior to March 31, 2025 an amount equal to the then remaining portion of the Net Proceeds from the Northeast Sale not to exceed \$350,000,000.

"<u>General Partner</u>: (a) prior to the C-Corp Conversion, Summit Midstream GP, LLC, a Delaware limited liability company, the general partner of the MLP Entity, and (b) following the C-Corp Conversion and in the event the MLP Entity is not controlled by a general partner, the New Parent.

"<u>MLP Entity</u>: Summit Midstream Partners, LP, a Delaware limited partnership, and each of its successors and assigns (including through any conversion to a corporation).

"<u>Obligor</u>: each of Borrower, the MLP Entity, each Subsidiary Guarantor and, if applicable, New Parent and any Intermediate Holding Company."

(c) Section 1.1 of the Existing Loan Agreement is hereby further amended to amend the last paragraph of the definition of "Pro Forma Basis" to read as follows:

For the avoidance of doubt, when making a determination on a Pro Forma Basis, any Acquisition or Asset Disposition involving Equity Interests owned by Borrower or any Restricted Subsidiary shall be treated as if such acquisition or disposition had occurred on the first day of the applicable Reference Period. Pro forma calculations made pursuant to the definition of the term "Pro Forma Basis" shall be determined in good faith by a Senior Officer of Borrower and, for any fiscal period ending on or prior to the first anniversary of

an Acquisition or Asset Disposition (or any similar transaction or transactions that require a waiver or consent of the Required Lenders pursuant to Section 10.2.5 or 10.2.6), may include adjustments in an aggregate amount of up to 20% of Unadjusted EBITDA for the most recent Test Period for which financial statements are available to reflect operating expense reductions and other operating improvements or synergies or interest income increases, in each case, reasonably expected to be realized over 12 months from the date such Acquisition, Asset Disposition or other similar transaction was consummated, to the extent that Borrower delivers to Agent (A) a certificate of a Financial Officer of Borrower setting forth such operating expense reductions and other operating improvements or synergies or interest income increases and (B) information and calculations supporting in reasonable detail such estimated operating expense reductions and other operating improvements or synergies or interest income increases.

1.2. <u>Amendment to Section 10.1.9</u>. Section 10.1.9(f) of the Existing Loan Agreement is hereby amended and restated in its entirety to read in full as follows:

"(f) Cause all Equity Interests of Borrower to be pledged to Agent for the benefit of the Secured Parties (including, without limitation, delivery of original certificates evidencing the Equity Interests, if any, of Borrower, together with an appropriate undated stock powers for each certificate duly executed in blank by the registered owner thereof)."

1.3. <u>Amendment to Section 10.2.4(e)</u>. Section 10.2.4(e) of the Existing Loan Agreement is hereby amended and restated in its entirety to read in full as follows:

"(e) Borrower may declare and make distributions on or with respect to its Equity Interests to the MLP Entity or one or more intermediate holding companies between the Borrower and the MLP Entity, the proceeds of which shall be used by the MLP Entity, a New Parent or such intermediate holding company to pay (or to make investments to allow any of its direct or indirect subsidiaries (other than any Excluded MLP Operating Subsidiary or any operating subsidiary of such intermediate holding company) to pay) (i) fees and expenses (including franchise or similar taxes) required to maintain the MLP Entity's, a New Parent's or such intermediate holding company's (or any of their respective direct or indirect subsidiaries') corporate existence, (ii) accounting, legal and administrative expenses and similar corporate overhead expenses of the MLP Entity, such New Parent or such intermediate holding company (or any of their respective direct or indirect subsidiaries), (iii) other ordinary course fees and expenses of the MLP Entity, such New Parent or such intermediate holding company relating to its status as a public company. (iv) other ordinary course fees and expenses of the MLP Entity, such New Parent or such intermediate holding company (or any of their respective direct or indirect subsidiaries) customarily incurred by "passive" holding companies, and (v) compensation and other benefits payable to, and indemnities provided on behalf of, officers, employees and consultants of the MLP Entity, such New Parent or such intermediate holding company (or any of their respective direct or indirect subsidiaries); provided that none of such proceeds shall be used by the MLP Entity, such New Parent, such intermediate holding company or any of their respective direct or indirect subsidiaries to pay (x) any fees and expenses of or reasonably attributable to any Excluded MLP Operating Subsidiary or any operating subsidiary of such intermediate holding company or (y) compensation and other benefits payable to, and indemnities provided on behalf of, officers, employees and consultants of, any Excluded MLP Operating Subsidiary, any operating subsidiary of such intermediate holding company or any other Person to the extent reasonably attributable to such Person's ownership or operation of any Excluded MLP Operating Subsidiary or any operating subsidiary of such intermediate holding company;"

1.4. <u>Amendment to Section 10.2.4(f)</u>. Section 10.2.4(f) of the Existing Loan Agreement is hereby amended and restated in its entirety to read in full as follows:

"(f) with respect to any taxable period (or portion thereof) for which (x) the Borrower or any of its Subsidiaries are members of a consolidated, combined, affiliated, unitary or similar income Tax group for U.S. federal or applicable foreign, state and/or local income tax purposes (each, a "Consolidated Tax Group") of which a direct or indirect parent of the Borrower taxable as a corporation is the common parent, or for which the Borrower is a partnership or disregarded entity for U.S. federal, state or local income tax purposes that is wholly-owned (directly or indirectly) by a Person that is taxable as a corporation for such income tax purposes, dividends or distributions by Borrower or an applicable Subsidiary, as may be relevant, to any direct or indirect parent of the Borrower in an amount required to pay federal, state, and/or local income taxes of such Consolidated Tax Group that are attributable to the taxable income of the Borrower and its Subsidiaries; provided that, for each such taxable period, the amount of payments made in respect of such taxable period shall not exceed the actual amount of any U.S. federal, state and/or local income taxes that the Borrower and its Subsidiaries would have paid for such taxable period had the Borrower and its Subsidiaries been a stand-alone corporate taxpayer or a stand-alone corporate group; or (y) the Borrower is a pass through entity (including a partnership or disregarded entity) for U.S. federal income tax purposes and is not wholly owned (directly or indirectly) by a Person that is taxable as a corporation for U.S. federal income tax purposes, the Borrower may make guarterly distributions to the MLP Entity in an amount not in excess of any tax distributions permitted to be made by the MLP Entity pursuant to the MLP Entity's Partnership Agreement and calculated as if the MLP Entity did not hold any assets other than Equity Interests of Borrower; provided, that, in the case of distributions under either clause (x) or clause (y) above, (i) Borrower may not make any such distribution after the occurrence, and during the continuance, of an Event of Default pursuant to Section 11.1(a), (e), (f), (g) or (h) or if such Event of Default would result therefrom, (ii) Borrower may only make such distributions with respect to the taxable income of an Unrestricted Subsidiary to the extent of the amount of cash distributed by such Unrestricted Subsidiary to the Borrower or a Restricted Subsidiary during the applicable taxable period and (iii) unless the Secured Parties have exercised or the Required Lenders have voted to exercise any rights or remedies pursuant hereto or under the Security Documents, Borrower may make only one such distribution after the occurrence, and during the continuance, of any other Event of Default;"

1.5. <u>Amendment to Section 10.3.2</u>. Section 10.3.2 of the Existing Loan Agreement is hereby amended and restated to read in full as follows:

10.3.2 Interest Coverage Ratio. Borrower shall not permit the Interest Coverage Ratio as of the last day of any Fiscal Quarter to be less than (i) for any Fiscal Quarter ending on or before December 31, 2024, (x) if no Loans are outstanding and Unrestricted Cash exceeds \$200,000,000 on such date, 1.50:1.00 and (y) if any Loans are outstanding or Unrestricted Cash is less than such threshold on such date, 1.75:1.00 and (ii) 1.90:1.00 for any Fiscal Quarter ending thereafter.

SECTION 2. LIMITED CONSENT. Notwithstanding anything in the Loan Documents, including Sections 5.2(b) and 10.2.6 of the Loan Agreement, to the contrary, in reliance on the

representations, warranties, covenants and agreements contained in this Fourth Amendment, and subject to the terms and conditions set forth in this Section 2 and subject to the conditions precedent set forth in Section 3, the Required Lenders hereby consent to the consummation of the Northeast Sale, so long as (a) the Northeast Sale is consummated on or before April 12, 2024 in accordance with the terms of the Northeast Sale Agreement delivered to Agent pursuant to Section 3.2 below, without giving effect to any waiver, modification or consent thereunder that could reasonably be expected to be material and adverse to the interests of Agent and the Lenders unless approved by Agent, and (b) Borrower uses Net Proceeds received from the Northeast Sale to prepay all outstanding Loans within one Business Day following the consummation of the Northeast Sale (which payment shall be deemed to satisfy all of Borrower's obligations under Section 5.2(b) of the Loan Agreement in respect of the Northeast Sale). The Required Lenders hereby instruct Agent to deliver, at the sole expense of the Borrower, all Lien and guarantee releases and related documentation reasonably requested by the Borrower in connection with the consummation of the Northeast Sale. No provision hereof shall constitute a waiver of, or a consent to the departure from, any of the other terms and conditions of the Loan Agreement or any other Loan Document, except for the limited consent as expressly set forth herein. This limited consent is a one-time limited consent to the Northeast Sale and shall not entitle Obligors to a consent or waiver in any other or similar circumstances.

SECTION 3. CONDITIONS PRECEDENT. The effectiveness of this Fourth Amendment is subject to the satisfaction (or waiver in accordance with Section 14.1 of the Existing Loan Agreement) of the following conditions precedent:

3.1. <u>Executed Counterparts</u>. Agent (or its counsel) shall have received duly executed counterparts of this Fourth Amendment from the MLP Entity, Borrower, each Subsidiary Guarantor and the Lenders constituting the Required Lenders, which shall include Regions Bank, as a Lender.

3.2. <u>Northeast Sale Agreement</u>. Agent shall have received an executed copy or substantially final version of the Northeast Sale Agreement, including the exhibits and schedules thereto, and each material agreement related thereto, in each case in form and substance reasonably satisfactory to Agent.

3.3. <u>Borrowing Base Report</u>. Borrower shall have delivered, or have caused to be delivered, to Agent an updated Borrowing Base Report giving pro forma effect to the Northeast Sale.

3.4. <u>Representations and Warranties</u>. The representations and warranties of each Obligor in <u>Section 4</u> of this Fourth Amendment shall be true and correct as of the date hereof.

3.5. <u>Fees and Expenses</u>. Borrower shall have paid all invoiced fees of counsel to Agent.

Notwithstanding anything to the contrary set forth in Section 14.1 of the Existing Loan Agreement or otherwise, Agent is hereby authorized and directed to declare this Fourth Amendment to be effective on the date that it receives the foregoing, to the reasonable satisfaction of Agent, or the waiver of such conditions as permitted hereby. Such declaration shall be final, conclusive and binding upon the Lenders and all other parties to the Existing Loan Agreement, as amended hereby, for all purposes.

SECTION 4. GENERAL REPRESENTATIONS AND WARRANTIES. Each Obligor represents and warrants to Agent and each of the Lenders that:

4.1. <u>Reaffirmation of Representations and Warranties</u>. The representations and warranties of each Obligor in the Loan Documents are true and correct in all material respects (without

duplication of any materiality qualifier contained therein) on the date hereof and will be true and correct in all material respects (without duplication of any materiality qualifier contained therein), in each case, immediately after giving effect to the amendments set forth in Section 1 hereof except for representations and warranties that expressly apply only on an earlier date which shall be true and correct in all material respects as of such earlier date (without duplication of any materiality qualifier contained therein).

4.2. <u>No Default</u>. Both immediately before and immediately after giving effect to this Fourth Amendment, no Default or Event of Default exists.

4.3. <u>Power and Authority</u>. Each Obligor is duly authorized to execute, deliver and perform this Fourth Amendment. The execution, delivery and performance by each Obligor of this Fourth Amendment have been duly authorized by all necessary action, and do not (a) require any consent or approval of any holders of Equity Interests of such Obligor, except those already obtained; (b) contravene the Organic Documents of such Obligor; (c) violate any Applicable Law; (d) violate, be in conflict with, result in a breach of or constitute (alone or with notice or lapse of time or both) a default under, give rise to a right of or result in any cancellation or acceleration of any right or obligation (including any payment) or to a loss of a material benefit under any indenture, lease, agreement or other instrument to which any Obligor or any Restricted Subsidiary is a party or by which any of them or any of their respective property is or may be bound, where any such conflict, violation, breach or default could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect; or (e) result in or require imposition of a Lien (other than a Permitted Lien) on any Property of Borrower or any Restricted Subsidiary.

4.4. <u>Enforceability</u>. This Fourth Amendment has been duly executed and delivered by each Obligor and constitutes a legal, valid and binding obligation of each Obligor, enforceable against each Obligor in accordance with its terms, except as enforceability may be limited by (a) bankruptcy, insolvency moratorium, reorganization, fraudulent conveyance or other laws affecting creditors' rights generally, (b) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law) and (c) implied covenants of good faith and fair dealing.

SECTION 5. MISCELLANEOUS.

5.1. Confirmation and Effect. The provisions of the Existing Loan Agreement (as amended by this Fourth Amendment) shall remain in full force and effect in accordance with its terms following the effectiveness of this Fourth Amendment, and this Fourth Amendment shall not operate as a waiver of any right, power or remedy of any Lender or Agent under any of the Loan Documents, nor constitute a waiver of any provision of any of the Loan Documents. Each reference in the Existing Loan Agreement to "this Agreement", "hereunder", "hereof", "herein", or words of like import shall mean and be a reference to the Existing Loan Agreement as amended hereby, and each reference to the Existing Loan Agreement in any other document, instrument or agreement executed and/or delivered in connection with the Existing Loan Agreement shall mean and be a reference to the Existing Loan Agreement as amended hereby. All Obligations under the Existing Loan Agreement and the other Loan Documents shall continue to be outstanding and shall be governed in all respects by the Existing Loan Agreement, as amended hereby, and the other Loan Documents, it being understood that neither this Fourth Amendment nor the amendments to the Existing Loan Agreement effectuated by this Fourth Amendment constitute a novation, satisfaction or re-borrowing of any Obligations under Existing Loan Agreement or any other Loan Document.

5.2. <u>Ratification and Affirmation of Obligors</u>. Each Obligor hereby (a) acknowledges and consents to all of the terms and conditions of this Fourth Amendment, (b) ratifies and affirms all of its obligations, including, without limitation, all of its payment and performance obligations, contingent or otherwise, under the Existing Loan Agreement (as amended hereby) and the other Loan

Documents to which it is a party, (c) ratifies and reaffirms any and all of the Liens or security interests granted by it on any of its Properties pursuant to any Loan Documents and confirms that such Liens and security interests continue to secure the Obligations and are in full force and effect as of the date hereof after giving effect to this Fourth Amendment and (d) ratifies and reaffirms its obligations under the Guaranty and agrees that such Guaranty is in full force and effect as of the date hereof after giving effect to this Fourth Amendment.

5.3. <u>Loan Document</u>. This Fourth Amendment shall constitute a "Loan Document", under and as defined in the Existing Loan Agreement, for all purposes under the other Loan Documents.

5.4. <u>Successors and Assigns; Amendments; Entire Agreement</u>. This Fourth Amendment (a) shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns (provided, however, no party may assign its rights hereunder except in accordance with the Loan Agreement); (b) may be modified or amended only in accordance with the Loan Agreement; and (c) TOGETHER WITH THE OTHER LOAN DOCUMENTS, EMBODIES THE ENTIRE AGREEMENT AND UNDERSTANDING AMONG THE PARTIES WITH RESPECT TO THE SUBJECT MATTER HEREOF AND SUPERSEDES ALL PRIOR AGREEMENTS, CONSENTS AND UNDERSTANDINGS RELATING TO SUCH SUBJECT MATTER.

Electronic Execution; Electronic Records; Counterparts. This Fourth 5.5. Amendment may be in the form of an Electronic Record and may be executed using Electronic Signatures (including, without limitation, facsimile and .pdf) and shall be considered an original, and shall have the same legal effect, validity and enforceability as a paper record. This Fourth Amendment may be executed in as many counterparts as necessary or convenient, including both paper and electronic counterparts, but all such counterparts are one and the same Fourth Amendment. For the avoidance of doubt, the authorization under this paragraph may include, without limitation, use or acceptance by Agent of a manually signed paper Communication which has been converted into electronic form (such as scanned into PDF format), or an electronically signed Communication converted into another format, for transmission, delivery and/or retention. Notwithstanding anything contained herein to the contrary, Agent is under no obligation to accept an Electronic Signature in any form or in any format unless expressly agreed to by Agent pursuant to procedures approved by it; provided, further, without limiting the foregoing, (a) to the extent Agent has agreed to accept such Electronic Signature, Agent shall be entitled to rely on any such Electronic Signature without further verification and (b) any Electronic Signature shall be promptly followed by a manually executed, original counterpart.

5.6. <u>Payment of Fees and Expenses</u>. The Obligors hereby agree, jointly and severally, to pay on demand all reasonable and documented legal (limited to reasonable and documented fees of one counsel for Agent and one counsel for Agent in each relevant jurisdiction) and other reasonable and documented out-of-pocket fees and expenses incurred by Agent in connection with the preparation, negotiation and execution of this Fourth Amendment and all related documents, in all cases to the extent required pursuant to Section 3.4 of the Loan Agreement.

5.7. <u>GOVERNING LAW; Submission to Jurisdiction; Waiver of Venue and Jury</u> <u>Trial</u>. THIS FOURTH AMENDMENT SHALL BE GOVERNED BY THE LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO ANY CONFLICT OF LAW PRINCIPLES THAT WOULD APPLY THE LAWS OF ANOTHER JURISDICTION, EXCEPT FEDERAL LAWS RELATING TO NATIONAL BANKS. The terms of the Existing Loan Agreement with respect to submission to jurisdiction, waiver of venue and waiver of jury trial are incorporated herein by reference, mutatis mutandis, and the parties hereto agree to such terms.

SECTION 6. RELEASE. AS PART OF THE CONSIDERATION FOR THE LENDERS' AND AGENT'S EXECUTION OF THIS FOURTH AMENDMENT, EACH OF THE MLP ENTITY, BORROWER AND EACH OTHER OBLIGOR. ON BEHALF OF ITSELF AND ITS SUCCESSORS. ASSIGNS. EOUITYHOLDERS. SUBSIDIARIES, OFFICERS, PARTNERS, DIRECTORS. EMPLOYEES, AGENTS AND ATTORNEYS HEREBY FOREVER, FULLY, UNCONDITIONALLY AND IRREVOCABLY WAIVES AND RELEASES THE LENDERS AND AGENT, AND EACH OF THEIR SUCCESSORS, ASSIGNS, EQUITYHOLDERS, SUBSIDIARIES, AFFILIATES, OFFICERS, DIRECTORS, EMPLOYEES, AGENTS AND ATTORNEYS (COLLECTIVELY, THE "RELEASEES") FROM ANY AND ALL CLAIMS, LIABILITIES, OBLIGATIONS, DEBTS, CAUSES OF ACTION (WHETHER AT LAW OR IN EQUITY OR OTHERWISE), DEFENSES, COUNTERCLAIMS, SETOFFS, OF ANY KIND, WHETHER KNOWN OR UNKNOWN, WHETHER LIQUIDATED OR UNLIQUIDATED, MATURED OR UNMATURED, FIXED OR CONTINGENT, DIRECTLY OR INDIRECTLY ARISING OUT OF, CONNECTED WITH, RESULTING FROM OR RELATED TO ANY ACT OR OMISSION UNDER THE LOAN AGREEMENT OR ANY OTHER LOAN DOCUMENT BY ANY LENDER, AGENT, OR ANY OTHER RELEASEE PRIOR TO THE DATE HEREOF (COLLECTIVELY, THE "CLAIMS"); PROVIDED THAT NO RELEASE SHALL OCCUR UNDER THIS SECTION 5 WITH RESPECT TO CLAIMS RESULTING FROM THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF ANY RELEASEE AS DETERMINED BY A FINAL NON-APPEALABLE JUDGMENT OF A COURT OF COMPETENT JURISDICTION. BORROWER, THE MLP ENTITY AND EACH OTHER OBLIGOR FURTHER AGREES THAT IT SHALL NOT COMMENCE, INSTITUTE, OR PROSECUTE ANY LAWSUIT, ACTION OR OTHER PROCEEDING, WHETHER JUDICIAL, ADMINISTRATIVE OR OTHERWISE, TO COLLECT OR ENFORCE ANY CLAIM, OTHER THAN A CLAIM RESULTING FROM THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF ANY RELEASEE AS DETERMINED BY A FINAL NON-APPEALABLE JUDGMENT OF A COURT OF COMPETENT JURISDICTION. THE FOREGOING RELEASE, COVENANT AND WAIVERS OF THIS SECTION 5 SHALL SURVIVE AND REMAIN IN FULL FORCE AND EFFECT REGARDLESS OF THE CONSUMMATION OF THE TRANSACTIONS CONTEMPLATED HEREBY, THE REPAYMENT OR PREPAYMENT OF ANY OF THE LOANS, OR THE TERMINATION OF THE EXISTING LOAN AGREEMENT, THE LOAN AGREEMENT, THIS FOURTH AMENDMENT, ANY OTHER LOAN DOCUMENT OR ANY PROVISION HEREOF OR THEREOF.

[Remainder of page intentionally left blank; signatures begin on following page]

IN WITNESS WHEREOF, this Fourth Amendment has been executed and delivered as of the date set forth above.

BORROWER: SUMMIT MIDSTREAM HOLDINGS, LLC

By: <u>/s/ William J. Mault</u> Name: William J. Mault Title: Executive Vice President and Chief Financial Officer

MLP ENTITY: SUMMIT MIDSTREAM PARTNERS, LP By: SUMMIT MIDSTREAM GP, LLC, its general partner

By: <u>/s/ William J. Mault</u> Name: William J. Mault Title: Executive Vice President and Chief Financial Officer

SUBSIDIARY GUARANTORS: DFW MIDSTREAM SERVICES LLC EPPING TRANSMISSION COMPANY, LLC SUMMIT DJ-O, LLC SUMMIT DJ-O OPERATING, LLC GRAND RIVER GATHERING, LLC SUMMIT DJ-S, LLC GRASSLANDS ENERGY MARKETING LLC SUMMIT MIDSTREAM FINANCE CORP. MOUNTAINEER MIDSTREAM COMPANY, LLC POLAR MIDSTREAM, LLC MEADOWLARK MIDSTREAM COMPANY, LLC SUMMIT MIDSTREAM UTICA, LLC SUMMIT MIDSTREAM MARKETING, LLC SUMMIT MIDSTREAM PERMIAN II, LLC RED ROCK GATHERING COMPANY, LLC SUMMIT MIDSTREAM NIOBRARA, LLC

By: <u>/s/ William J. Mault</u> Name: William J. Mault Title: Executive Vice President and Chief Financial Officer

SUMMIT MIDSTREAM OPCO, LP By: SUMMIT MIDSTREAM MARKETING, LLC, its general partner

By: <u>/s/ William J. Mault</u> Name: William J. Mault Title: Executive Vice President and Chief Financial Officer

Signature page to Fourth Amendment to Loan and Security Agreement

BANK OF AMERICA, N.A., as Agent

By: <u>/s/ Tanner J. Pump</u> Name: Tanner J. Pump Title: Senior Vice President

Signature page to Fourth Amendment to Loan and Security Agreement

BANK OF AMERICA, N.A., as a Lender and Issuing Bank

By: <u>/s/ Tanner J. Pump</u> Name: Tanner J. Pump Title: Senior Vice President

Signature page to Fourth Amendment to Loan and Security Agreement

ROYAL BANK OF CANADA, as a Lender

By: <u>/s/ Emilee Scott</u> Name: Emilee Scott Title: Authorized Signatory

Signature page to Fourth Amendment to Loan and Security Agreement

ING CAPITAL LLC, as a Lender

By: <u>/s/ Jean Grasso</u> Name: Jean Grasso Title: Managing Director

By: <u>/s/ Jeff Chu</u> Name: Jeff Chu Title: Director

Signature page to Fourth Amendment to Loan and Security Agreement

REGIONS BANK, as a Lender

By: <u>/s/ Aaron Wade</u> Name: Aaron Wade Title: Managing Director

Signature page to Fourth Amendment to Loan and Security Agreement

WINGSPIRE CAPITAL LLC, as a Lender

By: <u>/s/ Christopher Coutu</u> Name: Christopher Coutu Title: Senior Managing Director

Signature page to Fourth Amendment to Loan and Security Agreement

THE TORONTO-DOMINION BANK, NEW YORK BRANCH, as a Lender

By: <u>/s/ Jonathan Schwartz</u> Name: Jonathan Schwartz Title: Authorized Signatory

Signature page to Fourth Amendment to Loan and Security Agreement

FIRST-CITIZENS BANK & TRUST COMPANY (as successor by merger to CIT BANK, N.A.), as a Lender

By: <u>/s/ John Feeley</u> Name: John Feeley Title: Managing Director

Signature page to Fourth Amendment to Loan and Security Agreement

TRUIST BANK, as a Lender

By: <u>/s/ Greg Krablin</u> Name: Greg Krablin Title: Director

Signature page to Fourth Amendment to Loan and Security Agreement

MITSUBISHI HC CAPITAL AMERICA, INC., as a Lender

By: <u>/s/ James M. Giaimo</u> Name: James M. Giaimo Title: Chief Credit Officer Commercial Finance

Signature page to Fourth Amendment to Loan and Security Agreement



Summit Midstream Partners, LP 910 Louisiana Street, Suite 4200 Houston, TX 77002

Summit Midstream Partners, LP Announces Sale of Utica Position for \$625 Million

Concludes Strategic Alternatives Process; Plans to Seek Approval from Unitholders to Convert to a C-Corp

HOUSTON, (March 22, 2024) – Summit Midstream Partners, LP (NYSE: SMLP) ("Summit", "SMLP" or the "Partnership") announced today the sale of Summit Midstream Utica, LLC, which includes its approximately 36% interest in Ohio Gathering Company, LLC ("OGC"), approximately 38% interest in Ohio Condensate Company, LLC ("OCC", collectively with OGC, "Ohio Gathering") and wholly owned Utica assets (collectively, "Utica Position") to a subsidiary of MPLX LP ("MPLX") for \$625 million in cash (the "Utica Divestiture").

This transaction is the culmination of the comprehensive strategic review process undertaken by the Summit Board of Directors (the "Board"), in consultation with external advisors, that was publicly announced on October 3, 2023. As part of this process, the Board considered a wide range of opportunities to maximize value for unitholders, including an outright sale of Summit and other divestiture and partnership-level transactions. The Utica Divestiture and conclusion of the strategic review were unanimously approved by the Board. The Board and management team have completed their active process, but will remain open to all potential value-enhancing transactions.

In connection with the Company's strategic review, the Board also evaluated various corporate structures to determine how to drive the greatest long-term value for unitholders. The Board believes converting to a C-Corp positions the Company to maximize value by enhancing trading liquidity, greatly expanding the universe of potential investors and optimizing the long-term tax consequences to unitholders. The Board and management plan to seek approval from Summit unitholders to convert the Partnership to a C-Corp at a Special Meeting later this year. Summit expects to file a proxy statement to provide unitholders with additional information about the rationale and benefits of the C-Corp conversion in advance of the Special Meeting.

Transaction Highlights

- · Delivers significant value and dramatically improves credit profile and financial flexibility
- Reduces Summit's current net leverage by 1.5x to sub-4.0x, furthering progress toward achieving 3.5x net leverage target
- Increases liquidity with undrawn \$400 million credit facility and more than \$325 million of unrestricted cash
- Shifts Summit's portfolio to approximately 55% crude oil-oriented basins
- Accelerates timing of potential preferred equity and common equity distributions
- Positions Summit to continue to fund and execute on organic growth projects including further commercialization of Double E and potential synergistic bolt-on acquisitions, primarily in its Rockies segment
- Enables Summit to further reduce cost of capital in elevated interest rate environment
- Revised pro forma 2024 Adjusted EBITDA guidance of \$185 million to \$220 million¹

Management Commentary

Heath Deneke, President, Chief Executive Officer, and Chairman, commented, "We are pleased to announce this compelling transaction with MPLX, which is the result of a thorough strategic review, and generates substantial value creation opportunities for our unitholders. MPLX has been a great joint venture partner and operator of the OGC and OCC assets since Summit entered the Utica Shale in 2014 and we thank both MPLX and our valued Summit employees for their hard work and dedication to the partnership and producer customers in the region.

¹ Represents pro forma Adjusted EBITDA assuming the transaction closed on December 31, 2023.

As we evaluate the strategic and commercial opportunities for our existing assets, we believe there are several value optimizing strategies to pursue to further build scale, particularly in our Permian and Rockies segments. We continue to believe in the sizeable growth potential of the Delaware Basin and expect to further commercialize Double E's available pipeline capacity. We are confident there are additional commercial opportunities, similar to the recently announced Janus Processing Plant connection, that will continue to drive incremental free cash flow and the value of the pipeline. Further, we believe the DJ Basin and Williston Basin remain fragmented with a sizeable opportunities value-accretive and synergistic bolt-on acquisitions and organic growth opportunities. With a more focused portfolio and improved credit profile following completion of the transaction, we will be even better positioned to pursue those opportunities, expanding our footprint and service offerings in those regions. We also believe the transaction accelerates our ability to potentially resume preferred and common equity distributions in the near future as we rebuild scale and achieve our longer-term net leverage targets."

Utica Position Overview

Summit Utica: The Summit Utica system is a natural gas gathering system located in Belmont and Monroe counties in southeastern Ohio and serves producers targeting the drygas reserves of the Utica and Point Pleasant shale formations. The Summit Utica system gathers and delivers natural gas, primarily under long-term, fee-based gathering agreements, which include acreage dedications.

Ohio Gathering: Ohio Gathering comprises a natural gas gathering system and condensate stabilization facility located in the Utica Shale in southeastern Ohio. The gathering system spans the condensate, liquids-rich and dry-gas windows of the Utica Shale for multiple producers that are targeting production from the Utica and Point Pleasant shale formations across Belmont, Monroe, Guernsey, Harrison and Noble counties in southeastern Ohio. Substantially all gathering services on the Ohio Gathering system are provided pursuant to long-term, fee-based gathering agreements.

Advisors

RBC Capital Markets, LLC served as financial advisor for the Utica Transaction and Locke Lord L.L.P. served as legal advisor to Summit.

Use of Non-GAAP Financial Measures

SMLP reports financial results in accordance with U.S. generally accepted accounting principles ("GAAP"). SMLP also presents Adjusted EBITDA, a non-GAAP financial measure.

Adjusted EBITDA

SMLP defines Adjusted EBITDA as net income or loss, plus interest expense, income tax expense, depreciation and amortization, SMLP's proportional Adjusted EBITDA for equity method investees, adjustments related to MVC shortfall payments, adjustments related to capital reimbursement activity, unit-based and noncash compensation, impairments, items of income or loss that SMLP characterizes as unrepresentative of its ongoing operations and other noncash expenses or losses, income tax benefit, income (loss) from equity method investees and other noncash income or gains. Because Adjusted EBITDA may be defined differently by other entities in SMLP's industry, SMLP's definition of this non-GAAP financial measure may not be comparable to similarly titled measures of other entities, thereby diminishing its utility.

Management uses Adjusted EBITDA in making financial, operating and planning decisions and in evaluating SMLP's financial performance. Furthermore, management believes that Adjusted EBITDA may provide external users of SMLP's financial statements, such as investors, commercial banks, research analysts and others, with additional meaningful comparisons between current results and results of prior periods as they are expected to be reflective of SMLP's core ongoing business.

Adjusted EBITDA is used as a supplemental financial measure to assess:

- the ability of SMLP's assets to generate cash sufficient to make future potential cash distributions and support SMLP's indebtedness;
- the financial performance of SMLP's assets without regard to financing methods, capital structure or historical cost basis;
- · SMLP's operating performance and return on capital as compared to those of other entities in the midstream energy sector, without regard to financing or capital structure;
- the attractiveness of capital projects and acquisitions and the overall rates of return on alternative investment opportunities; and

 the financial performance of SMLP's assets without regard to (i) income or loss from equity method investees, (ii) the impact of the timing of minimum volume commitments shortfall payments under SMLP's gathering agreements or (iii) the timing of impairments or other income or expense items that SMLP characterizes as unrepresentative of SMLP's ongoing operations.

Adjusted EBITDA has limitations as an analytical tool and investors should not consider it in isolation or as a substitute for analysis of SMLP's results as reported under GAAP. For example:

- certain items excluded from Adjusted EBITDA are significant components in understanding and assessing an entity's financial performance, such as an entity's cost of capital and tax structure;
- · Adjusted EBITDA does not reflect SMLP's cash expenditures or future requirements for capital expenditures or contractual commitments;
- · Adjusted EBITDA does not reflect changes in, or cash requirements for, SMLP's working capital needs; and
- although depreciation and amortization are noncash charges, the assets being depreciated and amortized will often have to be replaced in the future, and adjusted EBITDA
 does not reflect any cash requirements for such replacements.

SMLP compensates for the limitations of Adjusted EBITDA as an analytical tool by reviewing the comparable GAAP financial measures, understanding the differences between the financial measures and incorporating these data points into SMLP's decision-making process.

SMLP does not provide the GAAP financial measure of net income or loss on a forward-looking basis because SMLP is unable to predict, without unreasonable effort, certain components thereof including, but not limited to, (i) income or loss from equity method investees and (ii) asset impairments. These items are inherently uncertain and depend on various factors, many of which are beyond SMLP's control. As such, any associated estimate and its impact on SMLP's GAAP performance could vary materially based on a variety of acceptable management assumptions.

About Summit Midstream Partners, LP

SMLP is a value-driven limited partnership focused on developing, owning and operating midstream energy infrastructure assets that are strategically located in the core producing areas of unconventional resource basins, primarily shale formations, in the continental United States. SMLP provides natural gas, crude oil and produced water gathering, processing and transportation services pursuant to primarily long-term, fee-based agreements with customers and counterparties in five unconventional resource basins: (i) the Appalachian Basin, which includes the Marcellus shale formation in West Virginia; (ii) the Williston Basin, which includes the Bakken and Three Forks shale formations in North Dakota; (iii) the Denver-Julesburg Basin, which includes the Niobrara and Codell shale formation in Colorado and Wyoming; (iv) the Fort Worth Basin, which includes the Barnett Shale formation in Texas; and (v) the Piceance Basin, which includes the Mesaverde formation as well as the Mancos and Niobrara shale formations in Colorado. SMLP has an equity method investment in Double E Pipeline, LLC, which provides interstate natural gas transportation service from multiple receipt points in the Delaware Basin to various delivery points in and around the Waha Hub in Texas. SMLP is headquartered in Houston, Texas.

Forward Looking Statements

This press release includes certain statements concerning expectations for the future that are forward-looking within the meaning of the federal securities laws. Forward-looking statements include, without limitation, any statement that may project, indicate or imply future results, events, performance or achievements and may contain the words "expect," "intend," "plan," "anticipate," "estimate," "believe," "will be," "will continue," "will likely result," and similar expressions, or future conditional verbs such as "may," "will," "should," "would," and "could", including statements about the divestiture, and the potential C-Corp conversion, uses of proceeds from the divestiture, the benefits of the divestiture and/or the C-Corp conversion, and any related opportunities, and the plans and objectives of management for future operations. In addition, any statement concerning future financial performance (including future revenues, earnings or growth rates), ongoing business strategies and possible actions taken by SMLP or its subsidiaries are also forward-looking statements. Forward-looking statements also contain known and unknown risks and uncertainties (many of which are difficult to predict and beyond management's control) that may cause SMLP's actual results in future periods to differ materially from anticipated or projected results. An extensive list of specific material risks and uncertainties affecting SMLP is contained in its Annual Report on Form 10-K for the year ended December 31, 2023, which filed with the Securities and Exchange Commission (the "SEC") on March 15, 2024 (the "Annual Report"), as amended and updated from time to time. Any forward-looking statements in this

press release are made as of the date of this press release and SMLP undertakes no obligation to update or revise any forward-looking statements to reflect new information or events.

Additional Information and Where to Find It

This communication relates to the proposed corporate reorganization of the Partnership. This communication may be deemed to be solicitation material in respect of the proposed C-Corp conversion. The proposed conversion is expected to be submitted to the Partnership's unitholders for their consideration. In connection with the proposed conversion, the newly formed corporation is expected to file with the SEC a Form S-4 containing a proxy statement/prospectus (the "Proxy Statement/Prospectus") to be distributed to the Partnership's unitholders in connection with the Partnership's solicitation of proxies for the vote of the Partnership's unitholders in connection with the proposed conversion and other matters as described in such Proxy Statement/Prospectus. The Proxy Statement/Prospectus will also serve as the prospectus relating to the offer of the securities to be issued to the Partnership's unitholders in connection with the completion of the proposed conversion. The Partnership and the newly formed corporation may file other relevant documents with the SEC regarding the proposed conversion. The definitive Proxy Statement/Prospectus will be mailed to the Partnership's unitholders when available. BEFORE MAKING ANY VOTING OR INVESTMENT DECISION WITH RESPECT TO THE PROPOSED CONVERSION, INVESTORS AND UNITHOLDERS AND OTHER INTERESTED PERSONS ARE URGED TO READ THE DEFINITIVE PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED CONVERSION (INCLUDING ANY AMENDMENTS OR SUPPLEMENTS THERETO) AND OTHER RELEVANT MATERIALS CAREFULLY AND IN THEIR ENTIRETY WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT THE PROPOSED CONVERSION.

The Proxy Statement/Prospectus, any amendments or supplements thereto and other relevant materials, and any other documents filed by the Partnership or the newly formed corporation with the SEC, may be obtained once such documents are filed with the SEC free of charge at the SEC's website at *www.sec.gov* or by directing a written request to the Partnership at 910 Louisiana Street, Suite 4200, Houston, Texas 77002.

No Offer or Solicitation

This communication does not constitute an offer to sell or the solicitation of an offer to buy any securities, or a solicitation of any vote or approval, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offering of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

Participants in the Solicitation

The Partnership, Summit Midstream GP, LLC, the general partner of the Partnership (the "General Partner"), and certain of the General Partner's executive officers, directors, other members of management and employees may, under the rules of the SEC, be deemed to be "participants" in the solicitation of proxies in connection with the proposed conversion. Information regarding the General Partner's directors and executive officers is available in the Annual Report. To the extent that holdings of the Partnership's securities have changed from the amounts reported in the Annual Report, such changes have been or will be reflected on Statements of Changes in Beneficial Ownership on Form 4 filed with the SEC. These documents may be obtained free of charge from the sources indicated above. Information regarding the participants in the proxy solicitation and a description of their direct and indirect interests, by security holdings or otherwise, will be contained in the Form S-4, the Proxy Statement/Prospectus and other relevant materials relating to the proposed conversion to be filed with the SEC when they become available. Unitholders and other investors should read the Proxy Statement/Prospectus carefully when it becomes available before making any voting or investment decisions.

Contact: 832-413-4770, ir@summitmidstream.com

SOURCE: Summit Midstream Partners, LP

SUMMIT MIDSTREAM PARTNERS, LP AND SUBSIDIARIES PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

March 2024 disposition of Summit Midstream Utica, LLC

On March 22, 2024, Summit Midstream Partners, LP ("Summit" and, together with its subsidiaries, "SMLP" or the "Partnership"), and Summit Midstream OpCo, LP, an indirectly owned subsidiary of Summit ("OpCo"), completed the sale of Summit Midstream Utica, LLC ("Utica") to a subsidiary of MPLX LP ("MPLX") for cash consideration of \$625.0 million, subject to customary postclosing adjustments (the "Transaction"), pursuant to a Purchase and Sale Agreement, dated March 22, 2024, by and among OpCo, as Seller, MPLX, as Buyer, and solely for purposes of Section 12.18 thereto, Summit, as Seller Parent (each as defined therein). Utica is the owner of (i) approximately 36% of the issued and outstanding equity interests in Ohio Gathering Company, L.L.C. ("OGC"), (ii) approximately 38% of the issued and outstanding equity interests in Ohio Condensate Company, L.L.C. (together with OGC, "Ohio Gathering") and (iii) midstream assets located in the Utica Shale. Ohio Gathering is the owner of a natural gas gathering system and condensate stabilization facility located in Belmont and Monroe counties in the Utica Shale in southeastern Ohio. MPLX is the operator of Ohio Gathering and, prior to the closing of the Transaction, was OpCo's joint venture partner.

Unaudited Pro Forma Condensed Consolidated Financial Statements

The following unaudited pro forma condensed consolidated financial statements of the Partnership are presented to illustrate the effect to the Partnership's historical financial position and operating results of the Transaction.

The Transaction constitutes a significant disposition for purposes of Item 2.01 of Form 8-K. As a result, the Partnership prepared the accompanying unaudited pro forma condensed consolidated financial statements in accordance with Article 11 of Regulation S-X. This divestiture does not qualify as a discontinued operation as it does not represent a strategic shift that will have a major effect on SMLP's operations or financial results.

The accompanying unaudited pro forma condensed consolidated balance sheet as of December 31, 2023 has been prepared to give effect to the Transaction as if it had occurred on December 31, 2023. The unaudited pro forma condensed consolidated statements of operations for the year ended December 31, 2023 have been prepared to give effect to the Transaction as if it had occurred on January 1, 2023.

The unaudited pro forma condensed consolidated balance sheet and statements of operations included herein are for information purposes only and are not necessarily indicative of the results that might have occurred had the Transaction taken place on the respective dates assumed. Actual results may differ significantly from those reflected in the unaudited pro forma condensed consolidated financial statements for various reasons, including but not limited to, the differences between the assumptions used to prepare the unaudited pro forma condensed consolidated financial statements of operations included herein include the use of estimates and actual results. The pro forma adjustments in the unaudited pro forma condensed consolidated balance sheet and the statements of operations included herein include the use of estimates and assumptions as described in the accompanying notes. The pro forma adjustments are based on information available to the Partnership at the time these unaudited pro forma condensed consolidated financial statements were prepared. The Partnership believes its current estimates provide a reasonable basis of presenting the significant effects of the Transaction. However, the estimates and assumptions are subject to change as additional information becomes available. The unaudited pro forma condensed consolidated financial statements include only those adjustments related to the Transaction.

The unaudited pro forma condensed consolidated financial statements are based on the historical consolidated financial statements of SMLP and should be read in conjunction with the consolidated financial statements and accompanying footnotes included in the Partnership's Annual Report on Form 10-K for the year ended December 31, 2023, filed with the Securities and Exchange Commission on March 15, 2024.

SUMMIT MIDSTREAM PARTNERS, LP AND SUBSIDIARIES PRO FORMA CONDENSED CONSOLIDATED BALANCE SHEET (UNAUDITED) As of December 31, 2023

| (In thousands, except unit amounts) | | Historical | | Pro forma Adjustments | | | Pro forma Combined |
|---|----|------------|----|--------------------------|---------|----|-----------------------|
| ASSETS | | | | | | | |
| Cash and cash equivalents | \$ | 14,044 | \$ | 311,432 | (a) (b) | \$ | 325,476 |
| Restricted cash | | 2,601 | | — | | | 2,601 |
| Accounts receivable | | 76,275 | | (9,017) | (b) | | 67,258 |
| Other current assets | | 5,502 | | (336) | (b) | | 5,166 |
| Total current assets | | 98,422 | | 302,079 | | | 400,501 |
| Property, plant and equipment, net | | 1,698,585 | | (184,563) | (b) | | 1,514,022 |
| Intangible assets, net | | 175,592 | | (15,691) | (b) | | 159,901 |
| Investment in equity method investees | | 486,434 | | (210,213) | (b) | | 276,221 |
| Other noncurrent assets | | 35,165 | | (2,485) | (b) | | 32,680 |
| TOTAL ASSETS | \$ | 2,494,198 | \$ | (110,873) | | \$ | 2,383,325 |
| LIABILITIES AND CAPITAL | | | | | | | |
| Trade accounts payable | \$ | 22,714 | \$ | 5,316 | (b) (c) | \$ | 28,030 |
| Accrued expenses | | 32,377 | | (125) | (b) | | 32,252 |
| Deferred revenue | | 10,196 | | (43) | (b) | | 10,153 |
| Ad valorem taxes payable | | 8,543 | | (7) | (b) | | 8,536 |
| Accrued compensation and employee benefits | | 6,815 | | (193) | (b) | | 6,622 |
| Accrued interest | | 19,298 | | _ | | | 19,298 |
| Accrued environmental remediation | | 1,483 | | _ | | | 1,483 |
| Accrued settlement payable | | 6,667 | | — | | | 6,667 |
| Current portion of long-term debt | | 15,524 | | _ | | | 15,524 |
| Other current liabilities | | 10,395 | | (1,280) | (b) | | 9,115 |
| Total current liabilities | | 134,012 | | 3,668 | | | 137,680 |
| Long-term debt, net | | 1,455,166 | | (313,000) | (a) | | 1,142,166 |
| Noncurrent deferred revenue | | 30,085 | | (264) | (b) | | 29,821 |
| Noncurrent accrued environmental remediation | | 1,454 | | — | | | 1,454 |
| Other noncurrent liabilities | | 30,266 | | (1,270) | (b) | | 28,996 |
| Total liabilities | | 1,650,983 | | (310,866) | | | 1,340,117 |
| Commitments and contingencies | | | | | | | |
| Mezzanine Capital | | | | | | | |
| Subsidiary Series A Preferred Units (93,039 issued and outstanding at December 31, 2023) | | 124,652 | | — | | | 124,652 |
| Partners' Capital | | | | | | | |
| Series A Preferred Units (65,508 units issued and outstanding at December 31, 2023) | | 96,893 | | _ | | | 96,893 |
| Common limited partner capital (10,376,189 units issued and outstanding at December 31, 2023) | | 621,670 | | 199,993 | (b) (d) | | 821,663 |
| Total partners' capital | | 718,563 | | 199,993 | | | 918,556 |
| TOTAL LIABILITIES AND CAPITAL | \$ | 2,494,198 | \$ | (110,873) | | \$ | 2,383,325 |
| | _ | | | | | _ | |

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

SUMMIT MIDSTREAM PARTNERS, LP AND SUBSIDIARIES PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF OPERATIONS (UNAUDITED) For the Year Ended December 31, 2023

| (In thousands, except per-unit amounts) | 1 | Historical | - | Pro forma djustments | | Pro forma Combined |
|---|----|------------|----|-------------------------|-----|-----------------------|
| Revenues: | | | | . | | |
| Gathering services and related fees | \$ | 248,223 | \$ | (38,920) | (e) | \$ 209,303 |
| Natural gas, NGLs and condensate sales | | 179,254 | | _ | | 179,254 |
| Other revenues | | 31,426 | | _ | | 31,426 |
| Total revenues | | 458,903 | | (38,920) | | 419,983 |
| Costs and expenses: | | · · · · · | | <u> </u> | | |
| Cost of natural gas and NGLs | | 112,462 | | _ | | 112,462 |
| Operation and maintenance | | 100,741 | | (4,499) | (e) | 96,242 |
| General and administrative | | 42,135 | | (470) | (e) | 41,665 |
| Depreciation and amortization | | 122,764 | | (8,589) | (e) | 114,175 |
| Transaction costs | | 1,251 | | 7,889 | (h) | 9,140 |
| Acquisition integration costs | | 2,654 | | _ | | 2,654 |
| Gain on asset sales, net | | (260) | | — | | (260) |
| Long-lived asset impairments | | 540 | | | | 540 |
| Total costs and expenses | | 382,287 | | (5,669) | | 376,618 |
| Other income, net | | 865 | | 12,312 | (g) | 13,177 |
| Gain on interest rate swaps | | 1,830 | | — | | 1,830 |
| Gain (loss) on sale of business / equity method investment | | (47) | | 207,881 | (i) | 207,834 |
| Interest expense | | (140,784) | | 26,227 | (f) | (114,557) |
| Loss on early extinguishment of debt | | (10,934) | | | | (10,934) |
| Income (loss) before income taxes and equity method investment income | | (72,454) | | 213,169 | | 140,715 |
| Income tax expense | | (322) | | _ | | (322) |
| Income from equity method investees | | 33,829 | | (22,922) | (e) | 10,907 |
| Net income (loss) | \$ | (38,947) | \$ | 190,247 | | \$ 151,300 |
| Less: Net income attributable to Subsidiary Series A Preferred Units | | (12,581) | | _ | | (12,581) |
| Net income (loss) attributable to Summit Midstream Partners, LP | \$ | (51,528) | \$ | 190,247 | | \$ 138,719 |
| Less: net income attributable to Series A Preferred Units | | (11,566) | | _ | | (11,566) |
| Net income (loss) attributable to common limited partners | \$ | (63,094) | \$ | 190,247 | | \$ 127,153 |
| Net income (loss) per limited partner unit: | | | | | | |
| Common unit – basic | \$ | (6.11) | | | | \$ 12.30 |
| Common unit – diluted | \$ | (6.11) | | | | \$ 12.06 |
| Weighted-average limited partner units outstanding: | | | | | | |
| Common units – basic | | 10,334 | | | | 10,334 |
| Common units – diluted | | 10,334 | | | | 10,544 |

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

SUMMIT MIDSTREAM PARTNERS, LP AND SUBSIDIARIES NOTES TO PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

1. BASIS OF PRESENTATION

The December 31, 2023 unaudited pro forma condensed consolidated balance sheet gives effect to the pro forma adjustments necessary to reflect the Transaction as if it had occurred on December 31, 2023. The unaudited pro forma condensed consolidated statement of operations gives effect to the pro forma adjustments necessary to reflect the Transaction as if it had occurred on January 1, 2023. The unaudited pro forma adjustments related to the sale of Utica are based on available information and assumptions that management believes are (i) directly attributable to the Transaction; (ii) factually supportable and (iii) with respect to the unaudited consolidated statements of operations, expected to have a continuing impact on consolidated operating results.

2. PRO FORMA ADJUSTMENTS

The unaudited pro forma condensed consolidated statements reflect the following adjustments:

Balance Sheet

"Historical" - represents the historical audited consolidated balance sheet of the Partnership as of December 31, 2023.

(a) To adjust Partnership balance sheet accounts for cash proceeds received associated with the divestiture, inclusive of (i) \$625.0 million of sale proceeds, (ii) a \$313.0 million pay down of the Partnership's asset-based revolving credit facility and (iii) related third party costs to effectuate the Transaction.

(b) To remove the Partnership's assets and liabilities associated with the Transaction. The following is a summary of the estimated gain to be recognized by the Partnership related to the Transaction:

| (in thousands) | |
|--|---------------|
| Sale price | \$ 625,000 |
| Cash on hand | (568) |
| Accounts receivable | (9,017) |
| Other current assets | (336) |
| Property, plant and equipment, net | (184,563) |
| Intangible assets | (15,691) |
| Investment in Ohio Gathering | (210,213) |
| Other noncurrent assets | (2,485) |
| Trade accounts payable, accrued expenses and other current liabilities | 4,177 |
| Deferred revenues | 307 |
| Other noncurrent liabilities | 1,270 |
| Estimated gain on sale of Utica | \$ 207,881 |
| | |

(c) To adjust for the unrecognized transaction costs associated with the Transaction, primarily for investment bank advisory and legal fees.

(d) To adjust for the estimated gain to be recognized by the Partnership related to the Transaction.

Income Statement

"Historical" - represents the historical audited statement of operations of the Partnership for the year ended December 31, 2023.

(e) Adjustments are to eliminate revenues and costs of Utica from the Partnership's consolidated financial results.

(f) To adjust interest expense for the impact of an estimated \$313.0 million pay down of the Partnership's asset-based revolving credit facility with proceeds received from the Utica divestiture, partially offset by increased interest expense resulting from an increase in commitment fees owed on the asset-based revolving credit facility.

(g) To adjust for the estimated interest income associated with the proceeds received from the Transaction.

(h) To adjust for unrecognized transaction costs associated with the Transaction, primarily for investment bank advisory and legal fees.

(i) To adjust for the estimated gain to be recognized as a result of the Transaction.